| NEW APPL   | ICATION   |  |
|--|---|--|
| ORIGINAL   | RECEIVE   | 0000120291   |
| <b>BEFORE THE ARIZONA COR</b>  |   |  |
| COMMISSIONERS  | 2310 NOV 22 P   | ų: 35  |
| KRISTIN K. MAYES, Chairman<br>GARY PIERCE<br>PAUL NEWMAN<br>SANDRA D. KENNEDY  | AN CORP COMMS<br>DOCKET CONTR   |  |
| BOB STUMP  | E-0   | 1345A-10-0474  |
| IN THE MATTER OF THE APPLICATION<br>OF ARIZONA PUBLIC SERVICE<br>COMPANY FOR AUTHORIZATION FOR<br>THE PURCHASE OF GENERATING<br>ASSETS FROM SOUTHERN CALIFORNIA<br>EDISON AND FOR AN ACCOUNTING<br>ORDER.  | DOCKET NO. E  |  |
| "APS Coal-Plant Plan Benefits All A creating intractable differences to reach a common goal:<br>- The Arizona Republic, Novembricher Internet I | a cleaner environm  |  |
| I. INTRODUCTION.   |   |  |
| Four Corners Power Plant ("Four Corner   |   |  |
| pressures and uncertainties on several fronts,   | , plant operations  | are threatened by the  |
|  |   |  |
| prospect of a forced shut down within the next   | -   |  |
| prospect of a forced shut down within the next<br>for APS customers. Four Corners provides A   | -   |  |
| prospect of a forced shut down within the next   | rizona with 1,053   | megawatts ("MW") of  |
| prospect of a forced shut down within the next<br>for APS customers. Four Corners provides A   | rizona with 1,053<br>n cost-effectively a   | megawatts ("MW") of<br>nd reliably power APS   |
| prospect of a forced shut down within the next<br>for APS customers. Four Corners provides A<br>critical baseload generation, 791 MW of which  | rizona with 1,053<br>h cost-effectively a<br>orners were to stop                        | megawatts ("MW") of<br>nd reliably power APS<br>generating energy, the                           |
| prospect of a forced shut down within the next<br>for APS customers. Four Corners provides A<br>critical baseload generation, 791 MW of which<br>customers' homes and businesses. If Four Co   | rizona with 1,053<br>h cost-effectively a<br>prners were to stop<br>y natural gas gener | megawatts ("MW") of<br>nd reliably power APS<br>generating energy, the<br>ation. But natural gas |

DOCKETED BY

M

transmission lines would also likely be needed to deliver replacement power to customers, further increasing capital costs.

In addition to the cost and reliability concerns, Four Corners has much more at stake. Four Corners is the economic lifeblood of the Navajo Indian Reservation ("Navajo Nation"), contributing millions of dollars in tax revenue to the Navajo Nation and surrounding community – dollars that make up more than one third of the Nation's general fund.<sup>2</sup> Four Corners and the Navajo mine provide jobs to over 1,000 full time local workers and to hundreds of others who work in support of the plant and mine, the great majority of whom are Navajo – people whose livelihoods depend upon the plant and will need to look elsewhere for work in a tough economy if it closes.<sup>3</sup> The Navajo Nation already suffers from around a 50% unemployment rate – five times the national average.<sup>4</sup> In the words of the Nation's president, retiring all of Four Corners' five units would be "cataclysmic" to the economic well-being of the tribe and surrounding community.<sup>5</sup>

There are also important environmental considerations. The Environmental Protection Agency ("EPA") has recently proposed a ruling that would require the Four Corners participants to install nitrogen oxide ("NOx") emission controls on each of the plant's five units, and additional mercury controls on Units 1-3. If finalized, this ruling could cause APS to invest more than \$660 million in capital by 2016.<sup>6</sup>

Against this backdrop, APS has been working diligently to find a solution that best balances the competing tensions at play. That effort came to a head earlier this year after Southern California Edison Company ("SCE"), for reasons unique to California utilities, stated that it would no longer make "life extending" capital investments in the plant and

- $25 \int_{4}^{3} See \ id.$ 
  - <sup>4</sup> See id. at 4. <sup>5</sup> See id.

<sup>6</sup> See id. at 5.

<sup>&</sup>lt;sup>2</sup> See Testimony of Mark A. Schiavoni ("Schiavoni Testimony") at 3.

would divest or otherwise terminate its 48% ownership share (739 MW) of Units 4 and 5 (the largest of the plant's five units) by 2016.<sup>7</sup> If no one picks up SCE's share, the coowners of Units 4 and 5 may elect to close those units, rather than assume the risk of a multimillion dollar expenditure for which there may be no subsequent recovery.

After analyzing several alternatives, APS determined upon a logical solution: retire Units 1, 2, and 3 (560 MW of less efficient generation that is wholly-owned by APS) and acquire SCE's share of Units 4 and  $5.^{8}$  This approach makes good sense:

• It saves APS customers money, providing them a nearly \$500 million net present value benefit. We estimate that the cost of purchasing SCE's share of and installing the EPA-proposed environmental upgrades on Units 4 and 5 is half of what it would cost APS to replace its Four Corners output with natural gas generation and build the transmission needed to bring that power to customers.<sup>9</sup>

• It has a lower customer bill impact than that of every likely alternative.<sup>10</sup>

• It significantly reduces Four Corners' regional carbon dioxide ("CO2") and other pollutant emissions by retiring three less efficient coal units and installing environmental upgrades on more efficient ones.<sup>11</sup>

• It saves hundreds of jobs and millions of dollars of revenue that are critical to the Navajo Nation and the local economy.<sup>12</sup>

20 21

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

It preserves the diversity of APS's current generation portfolio while tempering the Company's exposure to volatile natural gas prices.<sup>13</sup>

22 23

26

<sup>7</sup> See id. at 5-6. <sup>8</sup> See id. at 6. Testimorus

<sup>8</sup> See id. at 6; Testimony of Patrick Dinkel ("Dinkel Testimony") at 2.

24 <sup>9</sup> See Dinkel Testimony at 7.

- <sup>10</sup> See Testimony of Jeffrey B. Guldner ("Guldner Testimony") at 3-4.
- 25 <sup>11</sup> See Schiavoni Testimony at 8.
  - $^{12}$  See id. at 9.
  - <sup>13</sup> See Dinkel Testimony at 12.

It maintains APS's mix of reliable baseload energy. By providing a marginal 179 MW baseload capacity increase, it hedges the Company's energy mix against the possibility that output from other coal units also at risk could be retired and helps further defer the need for future baseload resources.<sup>14</sup>

These many benefits are compelling. To move forward on this path, APS needs the Commission's support of the proposed approach in certain important ways. First, APS cannot acquire SCE's share of Units 4 and 5 without authorization to do so under the terms of the "self-build moratorium" imposed in Decision No. 67744 (April 7, 2005). APS believes that the circumstances of this transaction fully support this request and respectfully asks that the Commission grant it.

In addition, while undeniably cost-effective compared to the alternatives and a good value for customers, this transaction requires significant investment by APS. To address the timing mismatch between costs and benefits that will occur between when the transaction closes and when associated costs are recovered in rates, this application also requests an accounting order that will: (1) allow the Company to defer for future recovery depreciation and amortization costs, operations and maintenance costs, property taxes, final coal mine reclamation, and carrying charges associated with APS acquiring SCE's share of Units 4 and 5; and (2) provide assurance that APS will be allowed to fully recover its investment in and carrying costs of Units 1-3, and any additional costs (most notably, decommissioning and mine reclamation) incurred in connection with the closure of those units.

Finally, APS respectfully asks that the Commission rule on this matter expeditiously. If the Commission rejects the Company's requests, Four Corners risks

<sup>14</sup> See id.

closing no later than 2016.<sup>15</sup> APS must start working to implement a contingency plan, accelerating the acquisition and construction of new generation and transmission infrastructure and/or installing emission control devices on Units 1-3.<sup>16</sup> Without a timely order from this Commission, time may run out to construct or buy new replacement generation.<sup>17</sup>

This application is supported by the testimonies of Mark Schiavoni, Patrick Dinkel, and Jeff Guldner.

II. BACKGROUND.

<sup>15</sup> See Schiavoni Testimony at 6.
<sup>16</sup> See Dinkel Testimony at 13-14.

<sup>18</sup> See Schiavoni Testimony at 2.

#### A. Four Corners: The Basics

Four Corners is located on the Navajo Nation in Fruitland, New Mexico, about 25 miles west of Farmington.<sup>18</sup> The plant consists of five generating units. The first three units, wholly-owned by APS, went online in 1963-1964.<sup>19</sup> Units 4 and 5, which are co-owned by APS, SCE, and four other utilities, entered commercial operation in 1969-1970.<sup>20</sup> Collectively, the five units generate 2,100 MW of baseload energy, enough electricity to power half a million homes.<sup>21</sup> For more than 40 years, Four Corners has been a reliable, high-performing generating resource, critical for economically meeting the region's energy needs.<sup>22</sup> The plant serves customers in Arizona, California, New Mexico, and Texas.<sup>23</sup>

As a baseload resource, Four Corners generates energy night and day, and it has
 done very well by APS customers in that regard. Over the past decade, the plant has had

- $\begin{bmatrix} 24 \\ 20 \end{bmatrix} \begin{bmatrix} 19 \\ 20 \end{bmatrix}$  See id.
- 25 <sup>21</sup> See id.

<sup>17</sup> See id. at 14.

- $2^2$  See id.
- $26 | ^{23} See id.$

- 5 -

an average annual capacity factor of 86%.<sup>24</sup> Units 1-3, the smallest of the five units, have a combined output of 560 MW. Units 4 and 5 are much larger, each providing 770 MW of electricity.<sup>25</sup> SCE owns 48% of Units 4 and 5 (a total entitlement of 739 MW) and APS owns 15% (231 MW). The other owners of Units 4 and 5 are Public Service Company of New Mexico (13%), Salt River Project ("SRP") (10%), El Paso Electric Company (7%) and Tucson Electric Power Company ("TEP") (7%).<sup>26</sup> APS operates the plant on behalf of all participants.<sup>27</sup>

Four Corners was the first mine-mouth generation station to take advantage of the large deposits of coal in the Four Corners region.<sup>28</sup> The Navajo mine, located adjacent to the plant and owned and operated by BHP Billiton, provides all of the plant's fuel.<sup>29</sup> Together, Four Corners and the Navajo mine provide jobs to roughly 1,000 people, more than 75% of whom are Native American.<sup>30</sup> The combined annual payroll is over \$100 million, a key contribution to the local economy.<sup>31</sup> The Navajo Nation receives approximately \$65 million in tax and royalty payments annually as a result of plant operations, making up over one-third of the Nation's total annual general fund.<sup>32</sup> Federal, 16 state, and local governments also benefit from nearly \$40 million in tax dollars that Four Corners and the Navajo mine pay each year.<sup>33</sup> Plant operations support local vendors as 18 well, contributing an estimated \$20 million annually for the services and goods those 19 vendors provide.34

20

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

17

21

<sup>25</sup> See id. at 3. 22 <sup>26</sup> See id. <sup>27</sup> See id. 23 <sup>28</sup> See id. at 2. <sup>29</sup> See id. 24 <sup>30</sup> See id. at 3. See id. 25 <sup>32</sup> See id. <sup>33</sup> See id. 26

<sup>24</sup> See id.

<sup>34</sup> See id.

- 6 -

The plant, mine, and their employees also support the community in other ways. APS and its employees donate over \$600,000 each year to the local United Way and provide more than 10,000 hours of volunteer community service in the Four Corners region.<sup>35</sup> The Navajo mine grants a number of college scholarships to the surrounding tribal community.<sup>36</sup> It is not an overstatement to say that the plant anchors the region's economy.

#### **B.** The Plant's Several Challenges.

Four Corners faces several complex environmental issues that threaten the plant's economic viability. On one front, the EPA has promulgated, or is expected to soon promulgate, a series of regulations that could require the plant to install a retinue of environmental controls in the near future.<sup>37</sup> These include, among others, Clean Air Act Regional Haze rules (requiring certain plants, including Four Corners, to install the "Best Available Retrofit Technologies" ("BART") to reduce haze in national parks and wilderness areas), Coal Combustion By-Products regulation, and strict emissions limitations for mercury and other air pollutants.<sup>38</sup> Environmental groups and the EPA are investigating or have alleged a series of "New Source Review" (a Clean Air Act program) violations at coal units around the country, including Four Corners.<sup>39</sup> And, of course, there is the yet uncertain cost of potential federal carbon legislation, which, if enacted, will impact many energy resources but will strike at coal generation the hardest.<sup>40</sup>

None of these pressures is unique to Four Corners. Power companies across the country struggle with how strategically to position their energy supply portfolios to address similar uncertainties. But these issues have now come to a head at Four Corners.

<sup>35</sup> See id.
 <sup>36</sup> See id.
 <sup>37</sup> See id.

- <sup>37</sup> See id. at 4. <sup>38</sup> See id.
- <sup>39</sup> See id. <sup>40</sup> See id.

See

- 7 -

A recent EPA proposed ruling would require the plant's participants to install postcombustion controls (a technology called Selective Catalytic Reduction ("SCR")) to address NOx emissions on each of the plant's five units and additional particulate emissions controls on Units 1-3, at a total estimated cost to APS customers of \$660 million.<sup>41</sup> If the proposed rules become final, APS will have to install the equipment and incur this capital cost by approximately 2016.<sup>42</sup>

Further exacerbating these stressors, Four Corners is located on the Navajo reservation pursuant to a lease that expires in 2016.<sup>43</sup> Before installing any environmental controls that will extend the life of the plant beyond then, the plant's participants must negotiate and gain Navajo Council approval of reasonable lease renewal and right-of-way extension agreements with the Nation – an effort that is politically complicated and currently ongoing. Because environmental improvements must begin soon, for all practical purposes, these agreements must be quickly approved.<sup>44</sup> An extension of the existing fuel agreement between APS and BHP must also be negotiated.<sup>45</sup> In short, many pieces must fall in place if Four Corners is to remain a viable long-term energy resource.

Compounding this pressure, SCE has advised APS that rules established by the California Public Utilities Commission ("CPUC") to implement a state greenhouse gas law prevent California utilities from making "life extending" capital expenditures at baseload power plants that do not meet certain greenhouse gas emissions standards, including Four Corners.<sup>46</sup> SCE applied for an exemption from those rules with respect to its ownership interest at Four Corners, but the CPUC rejected a wholesale exemption and

- <sup>41</sup> See id. at 5. This number is based upon APS's current 15% share of Units 4-5. The EPA has also indicated that 23 Navajo Generating Station, which meets 315 MW of APS's baseload energy needs, could also require significant environmental controls in the near future. See id.
- 24 See id. See id. 25

1

2

3

4

5

6

7

8

9

10

.11

12

13

14

15

16

17

18

19

20

21

22

- See id.
- <sup>45</sup> See id. 26 <sup>46</sup> See id.

implicitly cautioned SCE against making any "life extending" capital investments in that plant.<sup>47</sup> The CPUC decision prompted SCE to inform its co-owners that it will not pay its share of any environmental compliance or other costs for Units 4 and 5 that could be deemed "life extending" (such as SCRs), and that it will pull out from the plant entirely by 2016.<sup>48</sup> If no one picks up SCE's ownership interest, the other owners of Units 4 and 5 may well opt to shut those units down, rather than take on the risk of spending millions of dollars for SCE's share of environmental capital additions, for which there may be no later recovery. This circumstance has given APS a unique resource opportunity available on a short time clock.<sup>49</sup>

10

1

2

3

4

5

6

7

8

9

11

# C. A Solution that Benefits Customers, Communities, and the Environment.

Rather than sorting through this conundrum, an initial impulse may be to shut down the whole plant. But that would be a mistake. Coal generation is a fundamental and cost-effective component of APS's balanced resource portfolio, and Four Corners is a key part of APS's coal fleet.<sup>50</sup> A better solution is one that strikes a balance between minimizing costs to customers, mitigating environmental impacts, maintaining a diverse and reliable energy portfolio, and protecting the local economy.

APS has carefully analyzed this issue and run economic analyses of the various
 alternatives.<sup>51</sup> Informed by this effort, the Company proposes an approach here that best
 benefits APS customers, local communities, and the environment: acquire SCE's share of
 Units 4 and 5 and retire Units 1-3.<sup>52</sup> Here is why this solution makes sense.

- 22
- 23
- 24

25

<sup>47</sup> See id.
 <sup>48</sup> See id. at 5-6.
 <sup>49</sup> See id. at 5.

<sup>50</sup> See Dinkel Testimony at 3.

51 See id. at 3-4.

26  $5^2$  See id. at 5-7.

#### 1. Our customers save money.

#### a. Long term cost savings.

Simply put, the proposed transaction is the best value for APS customers compared to every other reasonable alternative, providing a nearly \$500 million net present value benefit.<sup>53</sup> A shutdown of Units 4 and 5 results in APS losing 231 MW of a reliable and economic baseload resource now serving APS customers.<sup>54</sup> Units 1-3 provide APS customers another 560 MW of baseload energy.<sup>55</sup> Although Units 1-3 currently comply with all environmental regulations, they will require significant environmentally-driven capital investment over the next five years if they are to remain in service. The first expected tranche, \$235 million for mercury emission controls, could come as early as 2014; the second, a potential \$351 million to comply with BART visibility requirements, is due as early as 2016.<sup>56</sup> Units 1-3 are cost-effective for APS customers now, but the math changes markedly when a total of \$586 million must be spent in five short years to keep them online – not to mention the uncertain impact of potential future carbon costs.<sup>57</sup>

If all five units are retired, APS will lose 791 MW of an important baseload resource that currently provides 19% of the Company's generation needs.<sup>58</sup> Navajo Generating Station, in which APS, SRP, and TEP each own a share, faces many of the same issues.<sup>59</sup> If it closes, APS would lose yet another 315 MW of baseload capacity, posing a real risk that APS could lose 1,106 MW – that is 26% of its total energy resources in just a few years.<sup>60</sup> "Baseload" generation is so defined because it is designed to run 24 hours a day, seven days a week, to meet the Company's lowest around-the-clock

- $25 \int_{58}^{57} See \, id.$

<sup>53</sup> See id. at 7.

<sup>54</sup> See id at 2. <sup>55</sup> See id.

<sup>56</sup> See id. at 3.

- <sup>58</sup> See id. <sup>59</sup> See id.
- 26 60 See id.

- 10 -

demand. Because this generation is continually called on to run, such a resource must be both reliable and cost-effective, or else customers will pay more for their energy.<sup>61</sup> Potential replacement alternatives for any lost Four Corners generation include coal and nuclear (large, conventional "baseload" resources), geothermal and biomass/biogas (small, renewable baseload resources), and natural gas (an "intermediate" resource that is reliable but that has higher cost volatility relative to others, and is most cost-effective when serving peak load).<sup>62</sup> Solar and wind generation, while increasingly important components of APS's energy mix, are intermittent resources that cannot adequately substitute for one that is required night and day, 365 days each year.<sup>63</sup>

10 Few of these alternative resources are realistically available to fill the void left if 11 Four Corners Units 4 and 5 were to shut down in 2016. Arizona does not have sufficient 12 geothermal resources to provide such capacity, and the geothermal that is available in 13 Southern California sells at a premium.<sup>64</sup> While APS will continue to seek to add biogas 14 and biomass generation irrespective of the outcome of this Application, Arizona has 15 limited amounts of these resources available.<sup>65</sup> Nuclear energy takes upwards of ten years 16 to develop and requires a comparatively large upfront capital relative to the other 17 alternatives.<sup>66</sup> Putting aside the large capital outlay required to build new nuclear, any 18 such resource would not be available by a 2016 need date (nor anywhere close).<sup>67</sup> While 19 energy efficiency will fill a portion of these requirements, APS is already committed to 20 aggressively pursuing its cost effective energy efficiency programs.<sup>68</sup> In any case, energy 21 efficiency cannot be a complete solution – a point well-demonstrated in Graph 4 on page

22

1

2

3

4

5

6

7

8

9

- 23
- 24
- <sup>65</sup> See id. 25 <sup>66</sup> See id.
- <sup>67</sup> See id. 26

<sup>61</sup> See id.

<sup>62</sup> See id. 63 See id. at 3-4.

68 See id.

<sup>64</sup> See id. at 4.

- 11 -

17, which compares what APS's energy mix will look like if this Application is approved to what it will be if it is not.<sup>69</sup>

1

2

3

4

5

6

7

8

9

10

11

12

If Units 4 and 5 close in 2016, APS is left with two realistic options: (1) continue the operation of Units 1-3 (which still leaves APS 231 MW short in 2016, possibly rising to 546 MW just a few years later if Navajo Generating Station retires); or (2) replace the power lost from Four Corners with combined-cycle gas generation.<sup>70</sup> But neither of these options is as beneficial to APS customers as retiring Units 1-3 early and replacing their output with the purchase of SCE's interest in Units 4 and 5, for several reasons.

First, from a cost perspective, customers will pay less under the proposed transaction than under either of these alternatives. The following graphs indicate why:

**GRAPH 1: CAPITAL COST COMPARISON** 



1 Graph 1 compares, on a dollar per kilowatt basis, the initial capital dollars required 2 for various generation resources.<sup>71</sup> For the Four Corners-related alternatives, the noted 3 value includes the cost of installing all anticipated environmental controls, a \$294 million 4 acquisition price for SCE's additional 739 MW, and the recovery of the decommissioning 5 transaction.72 reclamation liabilities associated the and final mine with 6 7 **GRAPH 2: LIFE CYCLE LEVELIZED COSTS** 8 150 9 10 125 11 12 100 13 \$/MWh 75 14 15 50 16 17 25

23

18

19

20

21

22

0

FC 4-5

- 24
- 25

<sup>71</sup> See id. at 5.

26 <sup>72</sup> The estimated cost of the assumed liabilities noted is \$32M. Pages 22-23 of this Application give a high-level summary of the contractual terms of the proposed transaction.

FC 1-3

Graph 2 compares, on a dollar per megawatt hour basis, the total cost of the

generation resource, fully integrated into the electrical system, levelized over the full life

CC

cycle of the plant.<sup>73</sup> For the Four Corners-related alternatives, the noted values include the cost of the environmental upgrades and an assumed carbon price of \$20/ton beginning in 2013 and escalating thereafter.<sup>74</sup>

As these graphs illustrate, none of the alternative resource scenarios brings the same cost benefit to APS customers as that proposed here. Consider the potential for keeping Units 1-3 in service, for example. In that case, APS customers will pay 44% more in capital costs to install the emission controls likely needed on Units 1-3 to keep those units operating than they will under the proposed transaction (an analysis that includes the cost of making the necessary environmental upgrades on Units 4 and 5).<sup>75</sup> In terms of net present value, keeping Units 1-3 online produces a customer revenue requirement that is \$1 billion greater than that resulting from the alternative proposed in this Application.<sup>76</sup>

Moreover, investing to keep Units 1-3 in-service simply preserves a resource that is already serving APS customers and does nothing to replace the other 231 MW of costeffective generation that APS would forego with the retirement of Units 4 and 5 in 2016, or protect against the potential loss of another 315 MW at Navajo Generating Station not long thereafter.<sup>77</sup> APS customers would incur that much more in replacement power costs if the Company pursued this option.<sup>78</sup> Retiring Units 4 and 5 in favor of Units 1-3 also makes little sense from an operational perspective, given that Units 1-3 are smaller and less efficient, and lack the same economies of scale benefits of Units 4 and 5.79

21

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

- 22
- 23
- 24
- <sup>77</sup> See id. 25 <sup>78</sup> See id.

<sup>74</sup> See id. <sup>75</sup> See id. at 7.

<sup>76</sup> See id.

<sup>73</sup> See Dinkel Testimony at 6.

<sup>79</sup> By way of example, the cost of installing SCRs on Units 1-3 is approximately \$627 per kW, while the cost of 26 installing the same equipment on Units 4 and 5 is roughly \$325 per kW - a marked difference. See id.

Natural gas remains the most viable resource alternative. The drawbacks of that option, however, are significant. First, it is much more expensive. Apart from the capital costs associated with additional combined cycle generation, a new gas resource would require APS both to build new transmission infrastructure and to maintain the current schedule of now-planned transmission lines.<sup>80</sup> As Graph 1 on page 12 shows, the cost of building new combined-cycle and transmission infrastructure is double the cost of purchasing SCE's share of Units 4 and 5 and installing the required environmental controls on those units, on a dollar per kilowatt basis. Moreover, as Graph 2 depicts, APS customers will pay almost 20% more per megawatt hour over the life cycle of a new gas plant than they will if APS acquires SCE's interest in Units 4 and 5.<sup>81</sup> In net present value terms, the natural gas alternative increases the customer revenue requirement by nearly \$500 million above that resulting from the proposed transaction.<sup>82</sup>

In addition, unlike Four Corners' fuel costs – which are dependable by virtue of a negotiated long-term fuel agreement with the supplier – gas prices are highly volatile, as evidenced by the following:<sup>83</sup>

 16

 17

 18

 19

 20

 21

 22

 23

 24

  $\frac{^{80} See id. at 8.}{15 ee id. at 6.}}{10.8^{82} See id. at 10.}$  

 83 See id. at 9.

- 15 -



#### **GRAPH 3: HISTORICAL U.S. FOSSIL FUEL PRICES**

Gas is a reliable resource that has an important place in a utility's resource portfolio, but if APS's resource mix becomes too dependent on natural gas, our customers will be highly exposed to potential fuel cost increases and volatility.<sup>84</sup> APS's resource choices, like those of all power generators, each have a variety of benefits, risks and uncertainties. This is why having a diverse energy mix, which reduces reliance on any single power source, mitigates risk and makes good business sense.<sup>85</sup> But, as Graph 4 shows, if APS replaces all 791 MW of its existing Four Corners coal capacity with natural gas generation, the Company's resource diversity decreases and customer reliance on natural gas generation increases by 90%.<sup>86</sup>

24

15

16

17

18

19

20

21

22

23

1

- 25 <sup>84</sup> See id. at 10.
  - <sup>85</sup> See id.
- 26 <sup>86</sup> See id.



As this graph shows, APS's future energy growth through 2017 will be met with 16 either renewable resources or energy efficiency measures. The proposed transaction 17 serves only to maintain the current balance of the Company's baseload resource mix.

There is also a practical risk to replacing the Four Corners output with natural gas. Additional gas generation and the associated transmission must be sited, permitted and constructed in a very short time frame if it is to be serving APS customers by 2016.<sup>87</sup> As with any construction project, there is always the risk that projects will be delayed and the resources will not be available to customers when needed.<sup>88</sup> Moreover, to execute this contingency, APS's currently planned and certificated Morgan to Sun Valley transmission line (commonly known as "TS-5 to TS-9") would need to be energized by 2016 - a feat

<sup>87</sup> See id. at 11. 26

<sup>88</sup> See id.

18

19

20

21

22

23

24

25

8

9

10

11

12

13

14

15

16

17

18

19

1

which may prove difficult given the unresolved right-of-way issues for that project.<sup>89</sup> The tight time clock not only makes the Four Corners alternative more appealing, but demonstrates the practical need for having this Application processed quickly.

#### b. Near-term rate impact.

Clearly, of the options that exist, the approach requested in this Application makes the most sense from a cost-perspective. It makes sense from a rate-impact perspective, too. Every scenario relating to Four Corners will cause customer bills to rise. With any large generation investment, the up-front capital costs and operating expenses of the asset will increase rates as the investment is added to rate base.<sup>90</sup> With the proposed transaction, however, the higher costs of the new generation will be mitigated in part by the reduced operating costs for Units 1-3.<sup>91</sup> Moreover, because Units 4 and 5 are approximately 10% more efficient than Units 1-3, they produce the same amount of energy at 10% lower fuel cost, producing a commensurate fuel cost savings.<sup>92</sup> Transaction costs will also be offset by the additional fuel savings that result from the displacement of 179 MW of generation that would otherwise be produced by natural gasfueled generating units or purchased from the wholesale market.<sup>93</sup> These latter fuel savings can be significant depending on the cost of natural gas at the time, and would accrue to customers almost concurrently as they run through the Company's Power Supply Adjustor ("PSA").

20

21

- 22
  - » <sup>89</sup> See id.

Because of these factors, the anticipated rate impact of APS's proposed approach is

the lowest of the alternatives.<sup>94</sup> As it stands today, APS will have to spend \$660 million

- 25  $9^{1}$  See id. at 3. 92 See id. at 3-4.
  - $^{93}$  See *id.* at 3-4.
- 26 <sup>94</sup> See id.

 <sup>&</sup>lt;sup>52e td.</sup>
 <sup>90</sup> For the proposed transaction, these costs include both the \$294 million cash price, as well as additional decommissioning and mine reclamation obligations that APS will assume with the purchase. The latter costs are currently estimated to total \$32 million. See Guldner Testimony at 8.

to install environmental controls at Four Corners if the EPA rules are passed as proposed. As a result, customer bills would already increase under the status quo. If the proposed transaction moves forward, customer bills would increase by about 4% by 2017, approximately one percentage point of which relates to APS's existing ownership obligations.<sup>95</sup> If the plant owners chose to shutdown all five units in 2016 and APS replaced the lost energy with natural gas, APS customer bills would increase by approximately 8%. If the plant owners retired Units 4 and 5 in 2016, but APS continued to operate Units 1-3, customer bills would increase by more than 6.5% in the same period. The proposed transaction thus saves customers as much as four percentage points on their electric bills when compared against the alternatives.

#### 2. Local communities benefit.

There is also a meaningful "people" element to this equation. If Four Corners stops operations entirely:

- Over 1,000 plant and mine workers, more than 75% of whom are Native American, will lose high-quality jobs in a still-struggling economy.<sup>96</sup>
- With the loss of those jobs comes the loss of over \$100 million in payroll, an important economic stimulus to the tribe and Farmington community.<sup>97</sup>
- The Navajo Nation will lose \$65 million in annual tax, fee, and royalty payments monies that make up 35% of its general fund.<sup>98</sup>
  - Federal, state, and local economies will lose yet another \$40 million or so in annual tax revenue.<sup>99</sup>
- <sup>95</sup> This and all cited bill impact analyses assume that costs are reflected in rates at the beginning of the year after that in which they are incurred. The actual bill impact will vary depending upon how and when the costs are reflected in rates and will not be determined until a subsequent rate case, in which the Commission will decide how the cost of the transaction should flow to customers. See id.
- 25 <sup>96</sup> See Schiavoni Testimony at 3.
  - <sup>97</sup> See id.
     <sup>98</sup> See id.
- 26 <sup>99</sup> See id.

| 1  | On the other hand, if the proposed transaction moves forward:                                  |
|----|--|
| 2  | • Jobs will be saved, and no Four Corners employee will suffer a layoff as a                   |
| 3  | result. APS expects that all position reductions resulting from the retirement                 |
| 4  | of Units 1-3 will occur naturally, through retirement or otherwise. <sup>100</sup>             |
| 5  | • The Navajo Nation and surrounding community will continue to benefit                         |
| 6  | from over \$100 million in yearly payroll- a critical asset to the local                       |
| 7  | economy. <sup>101</sup>  |
| 8  | • The Navajo Nation will continue to receive more than \$60 million annually                   |
| 9  | in tax, fee and royalty contributions, due to the continued operation of Units                 |
| 10 | 4 and 5. $^{102}$  |
| 11 | From a community perspective, the proposed transaction preserves the lion's share              |
| 12 | of the economic value that presently exists, which no other alternative can provide.           |
| 13 | 3. The environment will be cleaner.  |
| 14 | The proposed transaction also results in the emission of fewer environmental                   |
| 15 | pollutants, providing a cleaner energy resource for customers. If this application is          |
| 16 | approved and the Company accelerates the retirement of Units 1-3, the plant's capacity         |
| 17 | will be reduced from 2,100 MW to 1,540 MW and additional emission controls will be             |
| 18 | installed on Units 4 and $5.^{103}$ As a result, the plant will burn approximately 2.6 million |
| 19 | fewer tons of coal each year compared to what it would were all five units to remain           |
| 20 | online, significantly lowering the emission of pollutants into the atmosphere. <sup>104</sup>  |
| 21 | Consider the following:  |
| 22 |  |
| 23 |  |
| 24 | <sup>100</sup> See id.<br><sup>101</sup> See id.   |
| 25 | $^{102}$ See id. at 9.<br>$^{103}$ See id. at 7-8.   |
| 26 | <sup>104</sup> See id. at 8.   |

- 20 -



**GRAPH 5: FOUR CORNERS WILL HAVE SIGNIFICANT** 

As the foregoing makes clear, the proposed transaction will significantly reduce 17 Four Corners site emissions compared to current levels. Closing Units 1-3 alone reduces 18 site emissions for mercury by 61%, particulate matter by 43%, sulfur dioxide by 24%, and 19 carbon dioxide by 30%.<sup>105</sup> Retiring those units also reduces site NOx emissions by 36% -20 - a number that rises to 86% if post-combustion controls are installed on Units 4 and 5, as 21 the EPA has proposed.<sup>106</sup> The proposed transaction will also decrease water consumption 22 at the site by 20%. Although APS will have slightly more coal in its portfolio as a result 23

24

1

2

- 25
- 26
- <sup>105</sup> See id. <sup>106</sup> See id.

7

8

9

10

11

12

13

14

15

16

17

18

19

20

1

of this transaction, the 179 MW increment will be cleaner and will have very little impact on the Company's emissions profile compared to what it is today.<sup>107</sup>

In sum, the proposed transaction reduces pollutant emissions compared to what they now are and allows APS to maintain its cost-effective baseload resource to complement the Company's diversified energy portfolio. The environment benefits as a result.

#### D. The Purchase and Sale Agreement with Southern California Edison.

After months of negotiations, SCE has agreed to sell its 48% ownership interest in Units 4 and 5 – currently providing 739 MW of cost-effective baseload energy – for a cash price of \$294 million on the anticipated 10/1/12 contractual closing date.<sup>108</sup> APS will also assume certain of SCE's current obligations, such as recording plant decommissioning and mine reclamation liabilities.<sup>109</sup> Other transaction terms are specifically outlined in the Purchase and Sale Agreement executed between SCE and APS on November 8, 2010, which is attached to the Testimony of Mark Schiavoni at Attachment MAS-2.<sup>110</sup>

From SCE's perspective, these terms represent primarily the "keep whole" price for its customers, who will pay higher replacement power costs for four years due to SCE's earlier-than-contemplated withdrawal from Four Corners compared to what they would pay absent the agreement.<sup>111</sup> From APS's view, as described in detail above, this

21 <sup>107</sup> See id. at 8-9.

<sup>109</sup> See id. at 6-7. 23

<sup>&</sup>lt;sup>108</sup> See id. at 6. This price increases or decreases by \$7.5 million per month for each month the closing date is 22 accelerated or delayed respectively. See id.

<sup>&</sup>lt;sup>110</sup> Under the terms of the Four Corners Co-Tenancy Agreement, the plant's other owners have a Right of First Refusal on the opportunity to pick up their proportionate share of SCE's ownership interest, and have 120 days after 24 APS and SCE execute the Asset Purchase Agreement to exercise this right. See id. at 7. If any does, APS's acquired interest would decrease commensurately. See id. APS and SCE have also signed two ancillary agreements 25 addressing land lease and transmission issues, the latter of which will be submitted to the Federal Energy Regulatory Committee for timely approval. See id. 26 <sup>111</sup> See id.

transaction is an economic value for APS customers. The price is also a reasonable one to ensure that valuable assets already serving APS customers remain in service.<sup>112</sup>

#### **III. REQUEST.**

APS cannot move forward with the preferred approach without this Commission's action. Specifically, as required under the terms of the "self-build moratorium" imposed in Decision No. 67744, APS asks that the Commission authorize it to acquire SCE's share of Units 4 and 5. APS also seeks an accounting order that will (1) authorize the Company to defer for future recovery certain costs relating to the transaction; and (2) provide assurance that APS may continue to recover the capital carrying costs, depreciation, decommissioning, mine reclamation, and other obligations that may arise with respect to Units 1-3.

#### A. Under the terms of the "self-build moratorium" approved in Decision No. 67744, the Commission should authorize the acquisition of SCE's interest in Units 4 and 5.

Decision No. 67744 imposed certain restrictions on the Company's ability to construct or acquire an ownership interest in additional generating capacity. Except in certain circumstances not applicable here,<sup>113</sup> that Decision broadly requires that the Commission expressly approve "the acquisition of a unit or interest in a generating unit from any merchant or utility generator" with an in-service date prior to January 1, 2015 before APS may acquire it.<sup>114</sup> The Decision also sets forth criterion for APS to address when seeking such approval, intended to inform the Commission's determination as to whether or not it should be granted.<sup>115</sup> Generally stated, these include:

<sup>&</sup>lt;sup>112</sup> See id. at 7; Dinkel Testimony at 14.

 <sup>&</sup>lt;sup>113</sup> Several exceptions to the need to seek prior Commission authorization are identified in Decision No. 67744 (April 7, 2005). These include renewable resources, distributed generation less than 50 MW, temporary generation, etc. *See* Decision No. 67744 at Attachment A, Paragraph 74.

<sup>&</sup>lt;sup>114</sup> Decision No. 67744 (Finding of Fact 33).

<sup>26 &</sup>lt;sup>115</sup> See id. at Appendix A, Paragraph 75.

2 3 4

5

6

7

8

9

1

1.

That there is a long term need for the resource;<sup>116</sup>

- 2. That the Company has considered and sought the availability, if any, of similar resources from the competitive wholesale power market;<sup>117</sup>
- 3. That the acquisition is consistent with APS's resource plan and competitive acquisition rules;<sup>118</sup> and
  - 4. That the life cycle costs of the resource compared to alternatives are reasonable.<sup>119</sup>
    - 1. Long Term Need.

10 APS's Loads and Resources table shows that, even with this transaction, APS will 11 require 545 MW of additional resources to meet its 2017 load requirements (this is the 12 need remaining after the addition of over 1400 MW of renewable resources and energy 13 efficiency programs).<sup>120</sup> But if the proposed transaction fails, APS's need for new 14 resources could increase to over 1,500 MWs in 2017.<sup>121</sup> Output from Navajo Generating 15 Station may also be lost to similar vulnerabilities, giving need for yet another 315 MW of 16 replacement power. Were both Four Corners and Navajo Generating Station to shut down 17 entirely, APS's existing baseload resources would be limited to Cholla Power Plant 18 (providing a total of 647 MW) and Palo Verde Nuclear Generating Station (providing 19 1,146 MW) – a total of 1,793 MW to serve a 2020 minimum system demand of 2,530 20 MW.<sup>122</sup> Such a scenario would dramatically increase APS's reliance on natural gas and 21 our customers' exposure to gas price volatility.<sup>123</sup>

- 22
- <sup>116</sup> See id. at Paragraph 75(a).
  <sup>117</sup> See id. at Paragraph 75(b) and (c).
  <sup>118</sup> See id. at Paragraph 75(d).
  <sup>119</sup> See id. at Paragraph 75(e).
  <sup>120</sup> See Dinkel Testimony at 12.
  <sup>121</sup> See id.
  <sup>122</sup> See id.
  <sup>123</sup> See id. at 12.

Given that potential, the long-term need for maintaining sufficient, reliable baseload resources is clear. The proposed transaction essentially preserves a wellbalanced energy portfolio, with a net increase of 179 MW – a small difference that is unavoidable under the circumstances.<sup>124</sup> That additional 179 MW provides protection against volatile natural gas prices as well as the potential loss of the Navajo Generating Station capacity.<sup>125</sup> APS also expects to further defer the need for new baseload generation if the transaction is approved.<sup>126</sup>

8 9

1

2

3

4

5

6

7

### 2. Availability of Comparable Resources in the Competitive Wholesale Market.

Decision No. 67744 requires APS to address its efforts to obtain resources from the 10 competitive market. APS has looked at what exists in the competitive wholesale market, 11 but none of its offerings reasonably compare to the transaction with SCE.<sup>127</sup> As discussed 12 above, gas-fired generation, the most practical alternative to Four Corners in these 13 circumstances, would further expose APS customers to uncertain gas prices and require 14 that new transmission be built for any new gas-fired power to reach the Company's 15 primary load center in the Metropolitan Phoenix area.<sup>128</sup> Moreover, as discussed in detail 16 above, the gas generation option will likely be more expensive to APS customers in the 17 end, even after factoring in the acquisition price and cost of SCRs and other 18 environmental upgrades.<sup>129</sup> Although APS might also procure new coal through a 19 competitive acquisition, no such resource could likely be built in time to meet the 20 Company's need.<sup>130</sup> 21

- 22
- 23
- 24
- 24 <sup>126</sup> See id. <sup>127</sup> See id. <sup>128</sup> See id.
  - 5 <sup>128</sup> See id. at 13. <sup>129</sup> See id.

<sup>124</sup> See id. <sup>125</sup> See id.

26 <sup>130</sup> See id.

Moreover, although in recent years APS has procured the majority of its long-term power through RFPs, it need not have done so for the Commission to approve this transaction. Decision No. 67744 made clear that APS could negotiate bilateral agreements for power with non-affiliated parties, which is what the Company has done here.<sup>131</sup> The Commission reiterated this position in the Secondary Procurement Protocol (endorsed by the Commission in Decision No. 67744 at Paragraph 79) and again later in the Commission's resource procurement Best Practices.<sup>132</sup> Finally, the Commission's recently-enacted Resource Planning Rules state that a bilateral acquisition from a non-affiliate is appropriate even in the absence of a formal competitive solicitation if it represents "... a genuine, unanticipated opportunity to acquire a power supply resource at a clear and significant discount compared to the cost to acquire new generating facilities, and will provide unique value to the load-serving entity's customers." A.A.C. R14-2-705 (A) (4) and (B) (5).

As discussed above, each of these considerations is present here, particularly the
 last. Policy-making actions of a neighboring state gave rise to the route proposed in this
 Application, and the opportunity to purchase SCE's interest is now.

17

1

2

3

4

5

6

7

8

9

10

11

12

13

18

## 3. Consistency with Resource Plan and Competitive Acquisition Rules.

As previously noted, APS's Loads and Resources table indicates that even after acquiring SCE's share of Four Corners Units 4 and 5 and retiring Units 1–3, APS will still need over 500 MW of resources in the 2017 timeframe.<sup>133</sup> The Resource Plan currently on file with the Commission also stresses the value of maintaining a diverse energy supply portfolio – one that balances coal, gas, and nuclear generation to complement the ever-

- 24
- 25 <sup>131</sup> See Decision No. 67744 at Attachment A, Paragraph 77.
  - <sup>132</sup> See Decision No. 70032 (December 4, 2007) at 3.
- 26 <sup>133</sup> See Dinkel Testimony at 13.

#### 4. Life Cycle Costs.

rules, as addressed above.

As previously discussed, APS has analyzed the life cycle cost of acquiring SCE's share of Four Corners Units 4 and 5 against a hypothetical natural gas alternative, among others. This analysis shows that the SCE acquisition (including environmental upgrades) is far less expensive than building gas units and constructing new transmission for these units and results in a \$500 million net present value to APS customers.<sup>135</sup> The long-term cost of gas generation is, of course, sensitive to future natural gas prices. APS has conducted sensitivity analyses, however, that demonstrate that the economic advantage of acquiring SCE's interest in Four Corners persists over a wide range of natural gas prices. In order to break even with the life cycle cost of the proposed transaction, natural gas prices would have to be 20% lower than the current long-term forecast.<sup>136</sup> Or, the price assigned to carbon would have to rise above \$50 per ton, compared to a \$20/ton base (a number that itself is hypothetical, given the current actual carbon cost of \$0 per ton).<sup>137</sup> Alternatively, replacement combined-cycle capital costs would have to be half of current cost estimates to build that resource.

growing role of renewable resources and energy efficiency in meeting its customers'

energy needs.<sup>134</sup> Acquiring the SCE interest in Units 4 and 5, combined with the early

proposed transaction is also consistent with the Commission's competitive acquisition

The

retirement of Units 1-3, is fully consistent with the Company's resource plans.

<sup>137</sup> See id.

 <sup>&</sup>lt;sup>134</sup> See id. The same is true of the Resource Plan that APS will submit pursuant to the Commission's recently enacted
 Integrated Resource Planning rules.
 <sup>135</sup> See id. at 10

<sup>&</sup>lt;sup>135</sup> See id. at 10. <sup>136</sup> See id. at 9.

## B. The Commission should grant APS an accounting order authorizing cost deferral and facilitating the early retirement of Units 1-3.

Finally, APS respectfully requests that the Commission grant an accounting order that will (1) authorize APS to defer and capitalize for future recovery through rates all non-fuel costs of owning, operating and maintaining the acquired interest in Four Corners Units 4 and 5 (these costs include depreciation and amortization costs associated with the acquisition, operations and maintenance, property taxes, final coal reclamation, carrying charges, and other miscellaneous costs); and (2) provide assurance that APS will be allowed to fully recover its investment in Units 1-3, including carrying costs and all associated expenses (most notably, decommissioning and mine reclamation). An accounting order is a ratemaking mechanism that provides APS the ability to defer costs that would otherwise be expensed using Generally Accepted Accounting Principles ("GAAP"). In this case, such an order will address the timing mismatch between costs and benefits that occurs from when the proposed transaction closes (at which time customers will receive an immediate cost-benefit) to when APS is allowed to recover the asset in rates.

APS customers will see substantial long-term cost savings if the proposed transaction is approved, as discussed at length above. Those, savings, however, come at a significant short term cost that will be absorbed entirely by APS, absent a deferral. There is, for example, the \$294 million purchase price and the increased operating expenses associated with the additional ownership.<sup>138</sup> APS will also assume certain of SCE's assets and liabilities (such as those associated with final plant decommissioning and coal reclamation), which APS will record at fair value at the time of the acquisition.

24 25

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

26

<sup>138</sup> See Schiavoni Testimony at 6; Guldner Testimony at 8.

Collectively, these new costs amount to an estimated \$70 million per year.<sup>139</sup> If APS is not able to defer these costs, it will forever lose the opportunity to recover them.<sup>140</sup>

This inequitable result is amplified by the Company's PSA. Under that adjustment mechanism, customers will immediately benefit from the fuel savings that will result from the proposed transaction.<sup>141</sup> But because of the 90/10 sharing component of the PSA, APS will have only ten percent of that savings available to help offset the transaction's costs.<sup>142</sup> Put another way, APS customers instantly will pay less because of this transaction, but will not be required to pay for it until the end of the Company's next rate case. The Company will incur significant new costs to own and operate in Units 4 and 5, but will not be able to offset those cost increases in any meaningful way with the anticipated fuel savings. A deferral order will partially remedy this inequity until the increased costs are reflected in rates.<sup>143</sup>

In addition, if APS retires Units 1-3 after the SCE transaction is consummated in 2012, but is not clearly authorized by the Commission to fully recover the remaining plant value on those units and their associated costs, APS could be forced to take a write-off on those assets.<sup>144</sup> This would weaken the Company's equity ratio (a key concern in recent APS rate proceedings) and further pressure anticipated 2012 financial results.<sup>145</sup>

To address these consequences, APS requests an accounting order that will preserve its ability to recover costs related to the proposed transaction in a future rate case- accounting relief that the Commission has historically allowed. Specifically, APS seeks an order allowing it to defer and capitalize for future recovery all prudent non-fuel

23 <sup>139</sup> See Guldner Testimony at 8.
24 <sup>140</sup> See id.
141 See id. at 3-4 for a description of the referenced fuel savings.
25 <sup>142</sup> See id. at 8.
143 See id. at 9.
144 See id.
26 <sup>145</sup> See id.

operating, maintenance, depreciation, and capital carrying costs associated with SCE's share of Units 4 and 5. For purposes of the cost deferral, APS will calculate capital costs using the embedded cost of debt as of December 31, 2010 and the 11% cost of equity used in APS's last general rate case, at the ratio of 46.21% debt and 53.79% equity also set in that case.<sup>146</sup> The property taxes deferred will be based on actual property taxes that APS pays on the SCE share post-acquisition, less any property tax savings on Units 1-3.<sup>147</sup> APS will calculate depreciation at the Commission-authorized depreciation rates for APS's present interest in Unit's 4 and 5.<sup>148</sup> Operations and maintenance ("O&M") costs will be deferred and capitalized only to the extent that APS's share of overall Four Corners O&M increases as a result of the acquisition of SCE's interest.<sup>149</sup> The deferral would not increase customer bills today, but would simply provide a mechanism for APS to ensure future recovery of the costs related to the acquisition and mitigate the impact of the transaction on the Company's financial condition.<sup>150</sup>

The Commission has historically approved accounting orders in appropriate circumstances, such as these. For example, the Commission granted APS an accounting order that allowed the Company to defer the costs of owning, operating, and maintaining Sundance Generating Station (a simple cycle natural gas plant) when APS acquired that facility in 2005.<sup>151</sup> Authorizing the requested deferral, the Commission expressly recognized that the "PSA's cost/savings sharing component . . . coupled with the timing of the acquisition in relation to the pending rate case, could create a potential for inequity that justifies the creation of a deferral in this case."<sup>152</sup>

- 24 <sup>147</sup> See id.
  - 4 <sup>148</sup> See id. <sup>149</sup> See id. **a**t 9-10.
- 25 <sup>150</sup> See id. at 10.

<sup>146</sup> See id.

- <sup>151</sup> See Decision No. 67504 (January 20, 2005).
- 152 *Id.* at 25.

The Commission similarly granted APS two accounting deferral orders so that the Company would be able manage the financial impact of rising costs associated with owning new generation at Palo Verde Unit 2 and Palo Verde Unit 3.<sup>153</sup> In both of these instances, the Commission found the financial burden on APS absent the ability to defer the costs of owning and operating Palo Verde Units 2 and 3 and the essential inequity of the mismatch between cost recovery by APS and benefits realized by APS customers as sufficient reasons to grant the Company's request.<sup>154</sup> That rationale applies equally here.

The ability to defer for future recovery costs associated with the newly purchased share of Units 4 and 5 and to continue recovering the depreciation, decommissioning, mine reclamation and other expense obligations related to Units 1-3 will address the mismatch between when the transaction closes and when the underlying costs are recovered through rates, mitigating the inequity of APS forever losing the ability to recover expenses paid while the asset is in-service, benefiting customers. It will also put APS in a much better position to finance this transaction.<sup>155</sup> The deferral ensures that the deferred costs will eventually be recovered and mitigates the impact on the Company's earnings profile. The regulatory certainty of recovery that a deferral order will provide will likely allow APS to finance the transaction more easily and at lower costs, to the long-term benefit of customers.<sup>156</sup>

GAAP requires that the regulatory authorization of the deferral include language

that provides APS and its auditors with reasonable assurance that the Company will

- <sup>153</sup> See Decision Nos. 55325 (December 5, 1986) and 55939 (April 6, 1988).

<sup>155</sup> See Guldner Testimony at 11.

26 <sup>156</sup> See id. at 11.

See Decision Nos. 55325 (December 3, 1980) and 55959 (April 0, 1988).
 <sup>154</sup> More recently, the Commission granted APS an accounting order as part of the 2009 Settlement Agreement, in which it permitted APS to defer higher employee benefit costs. See Decision No. 71448 (December 30, 2009) at Appendix A, Paragraph 9. Previous Commission decisions similarly authorized APS to defer costs associated with net metering and vegetation management. See Decision Nos. 69663 at 89 (June 27, 2007) and 67744 at Attachment A, Paragraphs 110 and 111, respectively.

recover prudently incurred costs in future rates.<sup>157</sup> The accounting order must also include language that provides for the future recovery of costs associated with the closure of Units 1-3 in order to avoid the potential for immediate write-off of all or a portion of those costs upon closure (or even the final announcement by APS of closure).<sup>158</sup> Such language is proposed in Attachment B. To effectuate any cost-deferral, APS respectfully requests that the Commission adopt this or substantially similar language in its own final order.

An important note: the accounting order requested will not alleviate the financial impact of the capital investment that APS and other plant participants will likely make over the course of the next five years to bring Units 4 and 5 into environmental compliance. As in prior days of high generation capital requirements (such as when APS was bringing Palo Verde online), it will be important to find ways to phase-in the significant near-term incremental capital investment related to APS's coal strategy so as to avoid the financial pressure that will result were these costs shouldered all at once.<sup>159</sup> APS does not ask that any such rate making mechanisms be addressed in this docket.<sup>160</sup> The conversation applies equally to all coal-fired generation, not just Four Corners, and is a topic that APS intends to include in its next rate filing.<sup>161</sup>

′ | IV.

#### CONCLUSION.

For reasons outside of APS's control, the status quo is no longer an option. Of the options that exist, there is little question that acquiring SCE's interest in Units 4 and 5 and retiring early Units 1-3 is the best economic and environmental move. It has the lowest relative capital cost, greatest cost certainty, lowest customer rate impact, and allows APS to maintain a reliable and cost-effective source of baseload generation – all while

24 <sup>157</sup> See id.
 25 <sup>159</sup> See id. at 12.
 26 <sup>160</sup> See id.

| 1  | preserving the Navajo economy and improving the plant's environmental impact.                        |  |  |
|----|--|--|--|
| 2  | Although, like every alternative, there will be significant capital cost requirements in the         |  |  |
| 3  | short term, the approach outlined in this application provides nearly a \$500 million net            |  |  |
| 4  | present value benefit to APS customers. In other words, in a changed environment where               |  |  |
| 5  | every outcome has a price, APS's proposed transaction will cost our customers the least.             |  |  |
| 6  | RESPECTFULLY SUBMITTED this 22 <sup>nd</sup> day of November, 2010.                                  |  |  |
| 7  | RESPECTFULLY SUBMITTED this 22 day of November, 2010.  |  |  |
| 8  | PINNACLE WEST CAPITAL CORP.  |  |  |
| 9  | Law Department   |  |  |
| 10 | Me alian It. Chall   |  |  |
| 10 | Meghan/H. Grabel   |  |  |
|    | Thomas L. Mumaw<br>Attorneys for Arizona Public Service Company                                      |  |  |
| 12 |  |  |  |
| 13 | Original and 13 copies of the foregoing were filed this 22 <sup>nd</sup> day of November, 2010 with: |  |  |
| 14 |  |  |  |
| 15 | Docket Control<br>Arizona Corporation Commission   |  |  |
| 16 | 1200 West Washington   |  |  |
| 17 | Phoenix, Arizona 85007   |  |  |
| 18 | tillion Aprima   |  |  |
| 19 | KIN COMPL  |  |  |
| 20 |  |  |  |
| 21 |  |  |  |
| 22 |  |  |  |
| 23 |  |  |  |
| 23 |  |  |  |
|    |  |  |  |
| 25 |  |  |  |
| 26 |  |  |  |

# Attachment A

#### THE ISSUE: POLLUTION AT FOUR CORNERS

Attachment A Page 1 of 1



BUSINESS WIRE

Manager Larry Francis stands in front of APS' Four Corners Power Plant. APS is closing down the most-polluting part of the plant.

# APS COAL-PLANT PLAN BENEFITS ALL

hat does Arizona Public Service have in common with Alexander the Great? They both cut the Gordian knot.

The Gordian knot of Greek legend had defeated years of attempts to untie it. Alexander looked at the problem another way, pulled out his sword and sliced it in half.

APS was facing prohibitively expensive requirements to clean up one of the dirtiest generating stations in the nation, the coalfired Four Corners Power Plant.

The usual long, costly battle among a utility, regulators and environmentalists was looming.

Instead, APS sliced through the complex issues and came up with a creative solution: Close the dirtiest part of the plant and expand its ownership stake in the newer, cleaner part. APS is buying Southern California Edison's share of Four Corners, which is outside Farmington, N.M. It will also add more pollution controls.

With one ingenious stroke, APS has found a way to benefit customers, the environment, the Navajo Nation and its own long-term corporate interests.

"It's not only the right thing, but it's good business," says Mark Schiavoni, senior vice president of fossil generation at APS.

The plan still requires state and federal approval, as well as lease extensions from the Navajo Nation and the coal mine.

Under this proposal, Four Cor-

ners should finally lose the title of No. 1 U.S. emitter of nitrogen oxides, which not only cause respiratory problems but also add to haze at the Grand Canyon and other nearby national parks.

The partial closure is cheaper for APS than trying to clean up the old units, which will save customers money. The utility will actually end up with more energy from Four Corners, because its share of the remaining plant will supply more megawatts than the shuttered units. That will provide the critical round-the-clock "base load" that complements renewable energy.

Southern California Edison benefits, too, because California's regulated utilities are not allowed to invest in power plants that don't meet environmental standards on global warming.

For Navajos, the picture is more complex. The power plants and the coal that fuels them are major employers on the reservation. APS says no jobs will be lost in the near future, but employment is certain to shrink.

With concerns over pollution, the Navajo Nation must find economic-development opportunities that go beyond coal. That could include developing wind and solar resources. APS has consulted extensively with the tribe and could provide critical help.

A creative solution has cut through seemingly intractable differences to reach a common goal: a cleaner environment.

# Attachment B
# **Proposed Accounting Order Language**

IT IS THEREFORE ORDERED that Arizona Public Service Company's request for an Accounting Order permitting it to defer and capitalize, for later recovery through rates, all non-fuel costs of owning, operating, and maintaining the acquired SCE interest in Four Corners Units 4&5 and associated facilities be, and hereby is, granted. Costs to be deferred include depreciation, amortization of the acquisition adjustment, decommissioning costs, operations and maintenance costs, property taxes, final coal reclamation costs, carrying charges, and miscellaneous other costs. The carrying charges shall be computed using the embedded cost of debt as of December 31, 2010 and the 11% cost of equity used in Arizona Public Service Company's last general rate case, at the ratio of 46.21% debt and 53.79% equity also set in that case.

IT IS FURTHER ORDERED that Arizona Public Service Company shall reduce the deferral by non-fuel operations and maintenance and property tax savings associated with the closure of Four Corners Units 1–3.

IT IS FURTHER ORDERED that Arizona Public Service Company shall be allowed to recover all unrecovered costs associated with Four Corners Units 1-3 and any additional costs incurred in connection with the closure of Four Corners Units 1-3.

IT IS FURTHER ORDERED that Arizona Public Service Company shall be allowed to defer a return on all of the deferred costs computed using the embedded cost of debt as of December 31, 2010 and the 11% cost of equity used in Arizona Public Service Company's last general rate case, at the ratio of 46.21% debt and 53.79% equity also set in that case until the inclusion of any such unrecovered costs in rates in Arizona Public Service Company's next rate proceeding.

IT IS FURTHER ORDERED that the accumulated deferred balance, including the related deferred return, associated with all amounts deferred pursuant to this Decision will be included in cost of service and rate base for ratemaking purposes. Nothing in this Decision shall be construed to limit the Commission's authority to review such balance and to make disallowances thereof due to imprudence, errors or inappropriate application of the requirements of this Decision.

# Direct Testimony of Mark A. Schiavoni

| 1               |  |
|-----------------|--|
| 2               |  |
| 3               |  |
| 4               |  |
| 5               |  |
| 6               |  |
| 7               | DIRECT TESTIMONY OF MARK A. SCHIAVONI              |
| 8               |  |
| 9               | <b>On Behalf of Arizona Public Service Company</b> |
| 10 <sup>-</sup> | Docket No. E-01345A-10                             |
| 11<br>12        |  |
| 12              |  |
| 13              |  |
| 15              |  |
| 16              |  |
| 17              |  |
| 18              |  |
| 19              |  |
| 20              |  |
| 21              |  |
| 22              |  |
| 23              |  |
| 24              |  |
| 25              |  |
| 26              | November 22 2010                                   |
| 27<br>28        | November 22, 2010                                  |
| 28              |  |
|                 |  |
|                 |  |

# **Table of Contents** I. II. III. IV. V. March 1, 2010 Letter from President Dr. Joe Shirley of the Navajo Nation to Dr. Anita Lee of the EPA.....Attachment MAS-1 Purchase and Sale Agreement.....Attachment MAS-2

i

2

5

6

7

8

9

10

11

1

#### DIRECT TESTIMONY OF MARK A. SCHIAVONI ON BEHALF OF ARIZONA PUBLIC SERVICE COMPANY (Docket No. <u>E-01345A-10-</u>)

## 3 I. INTRODUCTION

## 4 Q. PLEASE STATE YOUR NAME AND POSITION WITH APS?

A. My name is Mark A. Schiavoni. I am the Senior Vice President of Fossil Generation at Arizona Public Service Company ("APS" or "Company"). In that capacity, I am responsible for overseeing all facets of the Company's non-nuclear generation operations and maintenance, including nearly 7,000 megawatts of fossil and renewable energy. This includes outage and work management, plant engineering and technical support, and operational oversight of co-owned facilities.

12

## Q. WHAT IS YOUR PROFESSIONAL BACKGROUND?

A. I joined APS from Exelon Corporation, where I spent nine years, most recently as 13 Senior Vice President of Exelon Generation and President of Exelon Power. I 14 began work at Exelon in 2000 as a nuclear plant manager, after which I assumed 15 various senior executive positions overseeing Exelon's fossil, hydroelectric and 16 renewable facilities, as well as the construction and integration of new generation 17 assets. Before joining Exelon, I worked as a nuclear plant maintenance director 18 for both Florida Power and Centerior Energy Corporation. Prior to that, I spent 19 22 years in the U.S. Navy's nuclear program. 20

## 21 Q. HAVE YOU PREVIOUSLY TESTIFIED BEFORE THIS COMMISSION?

- A. I recently spoke at the Commission's November 18, 2010 Special Open Meeting
  on the topic of how an Environmental Protection Agency ("EPA") proposed rule
  impacts Four Corners Power Plant ("Four Corners").
- 25 26

Q.

# WHAT IS THE PURPOSE OF YOUR DIRECT TESTIMONY IN THIS PROCEEDING?

- A. My testimony supports APS's application for authorization and other support needed to purchase Southern California Edison's ("SCE") current ownership
- 28

interest in Four Corners Units 4 and 5 and retire Units 1-3 of that plant. Specifically, I will describe the significance of Four Corners to Arizona and the three other states the plant serves, as well as its economic value to the Navajo Indian Reservation ("Navajo Nation") and surrounding community. I will also describe the challenges the plant faces, the terms of the purchase agreement reached between APS and SCE, and the environmental and community benefits that will result if APS's proposed transaction moves forward.

FOUR CORNERS: THE BASICS

PLEASE GENERALLY DESCRIBE FOUR CORNERS POWER PLANT.

Four Corners is located on the Navajo Nation in Fruitland, New Mexico, about 25 miles west of Farmington. Four Corners was the first mine-mouth generation station to take advantage of the large deposits of coal in the Four Corners region. The Navajo mine, which is adjacent to the plant, is owned and operated by BHP Billiton and exclusively provides the plant's fuel.

The plant consists of five generating units. The first three units, wholly-owned by APS, went online in 1963-1964. Units 4 and 5 – co-owned by APS, SCE, and four other utilities – entered commercial operation in 1969-1970. Collectively, the five units generate 2,100 MW of base load energy – enough electricity to power half a million homes. For more than 40 years, Four Corners has been a high-performing power plant, critical for economically meeting the region's energy needs. The plant serves customers in four different states: Arizona, California, New Mexico, and Texas.

As a base load resource, Four Corners generates energy night and day, and it has done very well by APS customers in that regard. Over the past decade, the plant has had an average annual capacity factor of 86%, proving it a reliable, costeffective resource. Units 1-3, the smallest of the five units, have a combined

27 28

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

II.

О.

A.

output of 560 MW. Units 4 and 5 are much larger, each providing 770 MW of electricity. SCE owns 48% of Units 4 and 5 (receiving a total of 739 MW) and APS owns 15% (231 MW). The other owners of Units 4 and 5 include Public Service Company of New Mexico (13%), Salt River Project ("SRP") (10%), El Paso Electric Company (7%) and Tucson Electric Power Company ("TEP") (7%). APS operates the plant on behalf of all participants.

# Q. DOES FOUR CORNERS IMPACT THE NAVAJO NATION AND SURROUNDING COMMUNITY?

A. Yes, very much so. Four Corners is the economic lifeblood of the Navajo Nation, contributing millions of dollars in payroll and tax revenue to the Navajo Nation and surrounding community. Together, Four Corners and the Navajo mine provide jobs to roughly 1,000 people, more than 75% of whom are Native American. The combined annual payroll is over \$100 million, a key contribution to the local economy. The Navajo Nation receives approximately \$65 million in tax and royalty payments annually as a result of plant operations, making up an impressive 35% of the Nation's total general fund. Federal, state, and local economies also benefit from nearly \$40 million in tax payments that Four Corners and the Navajo mine make each year. Plant operations support local vendors as well, contributing an estimated \$20 million annually for the services and goods those vendors provide.

The plant, mine, and their employees also contribute importantly to the community in other ways. APS and its employees alone donate over \$600,000 each year to the local United Way and provide more than 10,000 hours of volunteer community services. The Navajo mine also grants a number of college scholarships to the surrounding tribal community.

It is not an overstatement to say that plant operations stabilize the entire region's economy. The Navajo Nation reports that it already suffers from a more than 50% unemployment rate – five times the national average. In the words of the Nation's president, retiring all of Four Corners' five units would be "cataclysmic" to the economic well-being of the tribe and surrounding community.<sup>1</sup>

III.

Q.

## THE PLANT'S SEVERAL CHALLENGES

# DOES FOUR CORNERS FACE ANY ENVIRONMENTAL CHALLENGES?

A. Yes. Four Corners faces several complex environmental issues that threaten the plant's viability. On one front, the EPA has promulgated, or is expected to soon promulgate, a series of regulations that require coal generators to install a retinue of environmental controls in the near future. These include, among others, Clean Air Act Regional Haze rules (requiring certain plants, including Four Corners, to install the "Best Available Retrofit Technologies" ("BART") to reduce haze in national parks and wilderness areas), Coal Combustion By-Products regulation, and strict emissions limitations for mercury and other air pollutants. Environmental groups have alleged a series of "New Source Review" (a Clean Air Act program) violations at coal units around the country, including Four Corners. And, of course, there is the yet uncertain impact of federal carbon legislation, which, if enacted, will impact many energy resources but will strike at coal generation the hardest.

None of these pressures is unique to Four Corners. Power companies across the country struggle with how strategically to position their energy supply portfolios to address similar uncertainties. But these issues now directly confront Four Corners. A recent EPA proposed ruling (pending final approval) would require

<sup>&</sup>lt;sup>1</sup> See March 1, 2010 Letter from President Dr. Joe Shirley of the Navajo Nation to Dr. Anita Lee of the EPA, attached hereto as Attachment MAS-1.

the plant's participants to install post-combustion controls (a technology called Selective Catalytic Reduction ("SCR")) to address nitrogen oxide ("NOx") emissions on each of the plant's five units and additional particulate emissions controls on Units 1-3, at a total estimated cost to APS customers of \$660 million.<sup>2</sup> If the proposed rules become final, APS will have to install the equipment and incur this capital cost by approximately 2016.

## **DOES FOUR CORNERS FACE ANY OTHER CHALLENGES?**

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

19

20

21

23

24

25

Q.

A.

Yes, several. Four Corners is located on the Navajo reservation pursuant to a lease that expires in 2016. Before installing any environmental controls that will extend the life of the plant beyond then, the plant's participants must negotiate and gain Navajo Council approval of reasonable lease renewal and right-of-way extension agreements with the Navajo Nation – an effort that is politically complicated and currently ongoing. Because environmental improvements must begin soon, for all practical purposes, the approval process must be quickly completed. An extension of the existing fuel agreement between APS and BHP must also be negotiated in the near term.

In addition, SCE has advised APS that rules established by the California Public 18 Utilities Commission ("CPUC") to implement a state greenhouse gas law prevent California utilities from making "life extending" capital expenditures at baseload power plants not meeting certain greenhouse emissions standards, including Four Corners. SCE applied for an exemption from those rules with respect to its 22 ownership interest at Four Corners, but the CPUC rejected a wholesale exemption and implicitly cautioned SCE against making any "life extending" capital investments in that plant. The CPUC decision prompted SCE to inform its coowners that it will not pay its share of any environmental compliance or other 26

<sup>27</sup> <sup>2</sup> The EPA has also indicated that Navajo Generating Station, which meets 315 MW of APS's base load energy needs, could also require significant environmental controls in the near future. 28

costs for Units 4 and 5 that could be deemed "life extending" (such as SCRs), and that it will pull out from the plant entirely by 2016. If no one picks up SCE's ownership interest, the other owners of Units 4 and 5 may opt to shut those units down, rather than take on the risk of spending millions of dollars on environmental controls for which there may be no later recovery.

## IV. THE LOGICAL SOLUTION

# Q. HAS APS DECIDED ON AN APPROACH TO RESOLVE THE COMPETING TENSIONS THAT YOU HAVE DESCRIBED?

A. Yes. APS has been working diligently to find a solution that best balances the competing interests at stake. After analyzing several alternative resource options, APS determined upon a logical solution: retire Units 1, 2, and 3 (560 MW of less efficient generation) and acquire SCE's share of Units 4 and 5 (the more efficient generation units of the five).

This approach makes good sense. Among other reasons described in other testimonies supporting this Application, it significantly reduces Four Corners' carbon dioxide and other pollutant emissions by retiring three less efficient coal units and installing significant environmental upgrades on more efficient ones. It also saves hundreds of jobs and millions of dollars of revenue that are critical to the Navajo Nation and local economy.

# Q. PLEASE DESCRIBE THE PROPOSED TRANSACTION WITH SOUTHERN CALIFORNIA EDISON.

A. After months of negotiations, SCE has agreed to sell its 48% ownership interest in Units 4 and 5 – currently providing 739 MW of cost-effective base load energy – for a cash price of \$294 million on the anticipated 10/1/12 contractual closing date. This price increases or decreases by \$7.5 million per month for each month that the closing date is accelerated or delayed respectively. APS will also assume certain of SCE's current obligations, such as plant decommissioning and mine

reclamation liabilities. Other transaction terms are specifically outlined in the Purchase and Sale Agreement executed between SCE and APS on November 8, 2010, which is attached to this testimony as Attachment MAS- 2.

From SCE's perspective, these terms represent primarily the "keep whole" price for its customers, who will pay higher replacement power costs for at least the next four years due to SCE's earlier-than-contemplated withdrawal from Four Corners compared to what they would pay absent the agreement. From APS's view, this transaction has a clear economic value for APS customers, as described in the testimony of APS witness Patrick Dinkel. The price is also a reasonable one to ensure that valuable assets already serving APS customers remain in service.

While not specifically relevant to the requests in the Application, I also want to note a couple of other items that relate to the transaction. First, in addition to the Purchase and Sale Agreement, APS and SCE have signed two ancillary agreements addressing land lease and transmission issues, the latter of which I am told will be submitted to the Federal Energy Regulatory Committee for timely approval. Second, under the terms of the Four Corners Co-Tenancy Agreement, the plant's other owners have a Right of First Refusal on the opportunity to pick up their proportionate share of SCE's ownership interest, and have until March 8, 2011 to exercise this right. If any does, APS's acquired interest would decrease commensurately.

23 24

25

26

27

Q.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

## WHAT IMPACT WILL THE APPROACH THAT YOU HAVE OUTLINED HAVE ON THE PLANT'S ENVIRONMENTAL ISSUES?

A. The proposed transaction results in the emission of fewer environmental pollutants, providing a cleaner energy resource for customers. If this application is approved and the Company accelerates the retirement of Units 1-3, the plant's

7

capacity will be reduced from 2,100 MW to 1,540 MW and additional emission controls will most likely be installed on Units 4 and 5. As a result, the plant will burn approximately 2.6 million fewer tons of coal each year compared to what it would were all five units to remain online, significantly lowering the emission of pollutants into the atmosphere. The following graph depicts the anticipated reductions:



## FOUR CORNERS WILL HAVE SIGNIFICANT SITE EMISSION REDUCTIONS IN 2017

As the foregoing makes clear, the proposed transaction will significantly reduce Four Corners site emissions compared to current levels. Closing Units 1-3 alone reduces site emissions for mercury by 61%, particulate matter by 43%, sulfur dioxide by 24%, and carbon dioxide by 30%. Retiring those units also reduces site NOx emissions by 36% – a number that rises to 86% if post-combustion controls are installed on Units 4 and 5, as the EPA has proposed. The proposed transaction will also decrease water consumption at the site by 20%. Although

APS will have slightly more coal in its portfolio as a result of this transaction, the 179 MW increment will be cleaner and will have very little impact on the Company's emissions profile compared to what it is today.

In sum, the proposed transaction reduces pollutant emissions compared to what they now are. The environment benefits as a result.

#### Q. WHAT EFFECT WILL THE APPROACH THAT YOU HAVE OUTLINED HAVE ON THE NAVAJO NATION AND SURROUNDING COMMUNITY?

Α. A significant one. The proposed transaction removes any potential that Four Corners would shut down entirely with SCE's withdrawal from the plant. If the plant were to close, over 1,000 plant and mine workers - more than 75% of whom are Native American— will lose high-quality jobs in a still-struggling economy. With the loss of those jobs comes the loss of over \$100 million in payroll revenue, an important economic stimulus to the tribe and Farmington community. The Navajo Nation will lose \$65 million in annual tax, fee, and royalty payments – monies that make up 35% of its general fund. Federal, state. and local economies will lose yet another \$40 million or so in annual tax revenue. On the other hand, if the proposed transaction moves forward, jobs will be saved and no Four Corners employee will suffer a layoff as a result. APS expects that all position reductions resulting from the retirement of Units 1-3 will occur naturally, through retirement or otherwise. The Navajo Nation and surrounding community will continue to benefit from over \$100 million in yearly payroll revenue – a critical asset to the local economy. The Navajo Nation will continue to receive more than \$60 million annually in tax, fee and royalty contributions, due to the continued operation of Units 4 and 5.

10 11

12

13

14

15

16

17

1

2

3

4

5

6

7

8

9

21

22

23

24

25

In short, from a community perspective, the proposed transaction preserves the lion's share of the economic value that presently exists.

# **CONCLUSION**

# DO YOU HAVE ANY CONCLUDING REMARKS?

The Application presents an approach that is good for the environment, good for the Navajo Nation, and good for APS customers. We respectfully ask that the Commission approve the requests we need to make these benefits happen.

V.

Q.

Α.

# **Attachment MAS-1**

Attachment MAS - 1 Page 1 of 15

BENS

Vice Presiden



DR. JOE SHIRLEY, JR.

Dr. Anita Lee (Air-3) U.S. Environmental Protection Agency Region IX 75 Hawthorne Street San Francisco, CA 94105

> Re: Advanced Notice of Proposed Rulemaking: Assessment of Anticipated Visibility Improvements at Surrounding Class I Areas and Cost Effectiveness of Best Available Retrofit Technology of Four Corners Power Plant and Navajo Generating Station, 74 FR 44313; Docket No. EPA-OAR-2009-0598

Dear Dr. Lee:

The Navajo Nation takes this opportunity to comment on the above-referenced Advanced Notice of Proposed Rulemaking ("ANPR"). The ANPR is an initial step by the United States Environmental Protection Agency ("USEPA") in a rulemaking that will determine Best Available Retrofit Technology ("BART") to limit the emissions of oxides of nitrogen ("NO<sub>x</sub>") and particulate matter ("PM") from two coal-fired electric power plants located on the Navajo Reservation – the Navajo Generating Station ("NGS") and the Four Corners Power Plant ("FCPP") (collectively "Plants")<sup>1</sup>. For reasons set forth more fully below, the Nation supports a phased approach to emissions reductions for the Plants and has concluded that combustion controls - low No<sub>x</sub> burners ("LNB") and separated over fire air technology ("SOFA"), and **not** selective catalytic reduction ("SCR"), are BART for both Plants at this time.

#### INTRODUCTION

No entity has a greater interest in NGS and FCPP than the Navajo Nation. Accordingly, before addressing the specific issues raised in the ANPR, the Nation believes it is important to lay out in broad strokes the interests of the Nation implicated by this rulemaking. NGS and FCPP are located on Navajo lands pursuant to lease agreements with the Navajo Nation. The Plants provide hundreds of skilled jobs on the Navajo Reservation, where unemployment

Office of the President and Vice President Post Office Box 7440 / Window Rock, AZ / 86515 / Telephone: (928) 871-7000 / Fax: (928) 871-4025

<sup>&</sup>lt;sup>1</sup> For ease of reference these Comments will use FCPP and NGS to refer to both the Plants and their operators, Arizona Public Service and Salt River Project, respectively.

approaches fifty percent. The Nation's most valuable saleable natural resource is its coal reserves, and the Plants were located to take advantage of and provide a market for Navajo coal. The income these two Plants provide to the Nation, both directly and indirectly, contributes substantially to the Nation's economic viability and thus, ultimately, to its sustainability as an independent sovereign.

The Navajo Reservation, or Diné'tah, is the homeland of the Navajo people. It is a place of great scenic beauty and grand scenic vistas. The Nation's tribal parks and innumerable natural and archaeological treasures draw thousands of visitors each year. The Navajo people care deeply about their homeland and do not lightly accede to its degradation. In developing these comments, the Nation has balanced the environmental impacts of the Plants, the potential for improved visibility offered by the emission control alternatives, and the economic impacts that may result from the imposition of the different control technologies. The Regional Haze Rule requires that USEPA weigh all of these interests as well.

#### **CAA/REGIONAL HAZE RULE REGULATORY FRAMEWORK**

The Regional Haze Rule, and Section 169A of the Clean Air Act ("CAA") from which the Rule was derived, have as their goal the restoration of visibility in mandatory class I Federal areas ("Class I Areas"), including national parks and monuments, to pristine conditions by 2064.<sup>2</sup> This goal is to be accomplished in incremental steps, described in the CAA and the Rule as "reasonable progress."<sup>3</sup> The Rule establishes a series of review periods in which reasonable progress is measured. The current rulemaking is the first review period for NGS and FCPP for NO<sub>x</sub> and PM.

One of the ways in which USEPA proposes to accomplish reasonable progress is to reduce emissions from large stationary sources like NGS and FCPP through the requirement of BART. The Act establishes, and the Rule reiterates, a five factor test for determining BART for such sources:

[T]he State [or in this instance USEPA] shall take into consideration the costs of compliance, the energy and nonair quality environmental impacts of compliance, any existing pollution control technology in use at the source, the remaining useful life of

<sup>&</sup>lt;sup>2</sup> 42 USC § 7491(a); 40 CFR § 51.308.

<sup>&</sup>lt;sup>3</sup> *Id.* at § 7491(a)(4); 40 CFR § 51.308(d)(1).

the source, and the degree of improvement in visibility which may reasonably be anticipated to result from the use of such technology.<sup>4</sup>

USEPA has issued *Guidelines for BART Determinations Under the Regional Haze Rule<sup>5</sup>* ("BART Guidelines") to aid in construction of the Rule. The BART Guidelines, which are mandatory for coal-fired plants like NGS and FCPP, that generate more than 750 Mw of power, establish presumptive limits for emissions of NO<sub>x</sub>. The preamble to the Federal Register Notice publishing the BART Guidelines explains that these limits, which apply to all three units at NGS and units 3, 4 and 5 at FCPP, "are based on current combustion control technology."<sup>6</sup> The preamble reflects USEPA's rationale in selecting presumptive limits based on current combustion controls:

Based on our analysis of emissions from power plants, we believe that applying these highly cost-effective controls at the large power plants covered by the guidelines would result in significant improvements in visibility and help to ensure reasonable progress toward the national visibility goal.<sup>7</sup>

The ANPR focuses primarily on two control technologies that are "current combustion controls," LNB and SOFA, and on SCR, a post-combustion control technology. The ANPR fails to adequately explain why USEPA is considering deviating from the presumptive limits and presumptive BART control technologies identified in the BART Guidelines.

While the Regional Haze Rule was created to protect scenic values in federal Class I Areas, it offers no comparable protections against visibility impacts of emissions on the Nation's lands.<sup>8</sup> To protect these and other environmental interests, the Nation has established its own environmental protection agency ("NNEPA") to monitor and regulate activities affecting the environment on the Reservation, and NNEPA has been instrumental in the development of these comments. NNEPA is still in the process of developing its regulatory programs and capacities, however, and has not applied for "treatment as a state" or program approval for a visibility program.<sup>9</sup> USEPA therefore is currently the sole regulatory authority determining

- <sup>6</sup> 40 CFR Part 51: Regional Haze Regulations and Guidelines for Best Available Retrofit Technology (BART) Determinations; Final Rule ("FR Notice"), 70 FR 39104, 39134 (July 6, 2005).
- <sup>7</sup> Id.

<sup>8</sup> The Nation could seek to protect its lands by petitioning USEPA to have the Navajo Nation designated as a class | area. 42 U.S.C. § 7474(a) and (c).

<sup>9</sup> See 42 U.S.C. § 7601(d) (CAA "treatment as a state" provision), 40 C.F.R. §§ 49.3, 49.4(e).

<sup>&</sup>lt;sup>4</sup> Id. at § 7491(g)(2); § 51.308(e)(1)(ii)(A).

<sup>&</sup>lt;sup>5</sup> 40 C.F.R. Part 51, Appendix Y ("BART Guidelines"); see 42 U.S.C. § 7491(b)(1).

BART for the Plants, pursuant to the CAA generally and the Tribal Authority Rule.<sup>10</sup> In doing so, however, USEPA not only has its usual obligation to solicit comments on its proposed regulation, but also has a trust obligation, as a federal agency, to consult with the Nation as an affected Indian tribe. Under these circumstances, USEPA must do more than simply receive and respond to the Nation's comments; it must give substantial deference to the Nation's views regarding BART for stationary sources on the Nation's lands.

#### **COMMENTS SOLICITED BY THE ANPR**

The Nation is aware that the ANPR is less comprehensive than a proposed rule and does not propose the installation of any particular control technology as BART for the Plants. Instead, as the Nation understands the purpose of the ANPR, USEPA seeks comments on two of the five factors that make up the BART test so that it may be fully informed before it issues a proposed rule. The ANPR solicits comments on data relied upon by the Plants and reviewed, and in some instances modified, by USEPA on these two factors. The first factor is the cost of installing the various control options (Factor 1). The second factor is the degree of visibility improvement that can be anticipated with installation of the various emissions control options (Factor 5), and USEPA seeks comments on the models employed and the inputs used. In the ANPR, USEPA provides that comments may also be submitted on the remaining three factors of the BART analysis. Further, USEPA indicates that comments addressing the Agency's obligations under Executive Order 13175, *Consultation and Coordination with Indian Tribal Governments* ("EO 13175")<sup>11</sup> are also welcome.

Because of the importance to the Nation of economic issues that may be considered under the BART Guidelines, and that are further implicated by EO 13175, which is one tangible manifestation of the federal government's trust relationship to Indian tribes, the Nation's comments begin with a discussion of the potential economic harm to the Nation should USEPA require the installation of control technologies that exceed the requirements of the Regional Haze Rule and that are excessively costly.

#### I. AFFORDABILITY OF CONTROLS, ENERGY IMPACTS (FACTOR 2) AND EO 13175

The Nation is aware of at least three authorities for considering the economic impacts of any proposed emission control technology. There are two distinct provisions in the BART

<sup>&</sup>lt;sup>10</sup> 40 C.F.R. § 49.11.

<sup>&</sup>lt;sup>11</sup> 65 FR 67249 (Nov. 6, 2000).

Guidelines that address economic impacts. In addition, USEPA's federal trust responsibility to Indian tribes must inform any action taken by USEPA that has the potential for serious economic harm. Both Plants have raised the specter of Plant closure should USEPA require controls so costly that they cannot be reasonably amortized over the remaining life of the Plants. One need only look to the recent shut down of the Mohave Generating Station to realize that this is no idle threat. The Mohave closure had significant impacts on the Nation, even though that plant was not located on Navajo lands. Closure of Mohave was directly responsible for the shutdown of the Black Mesa Mine, which mined coal owned jointly by the Nation and the Hopi Tribe, and resulted in the loss of jobs for tribal members and royalty revenue for the Nation. The loss of one or both of the Plants as a consequence of this rulemaking would cause the Nation far greater harm.

The BART Guidelines anticipate the situation presented here, where requiring a particular control technology has the potential to affect not just the profitability, but indeed the continued viability, of NGS and FCPP. In Section IV(E)(3), the BART Guidelines address how economics are factored into the BART determination once each of the five legislatively mandated factors have been evaluated:

In selecting a "best" alternative, should I consider the affordability of controls?

- 1. Even if the control technology is cost effective, there may be cases where the installation of controls would affect the viability of continued plant operations.
- 2. There may be unusual circumstances that justify taking into consideration the conditions of the plant and the economic effects of requiring the use of a given control technology. These effects would include effects on product prices, the market share, and profitability of the source. Where there are such unusual circumstances that are judged to affect plant operations, you may take into consideration the conditions of the plant and the economic effects of requiring the use of a control technology. Where these effects are judged to have a severe impact on plant operations you may consider them in the selection process, but you may wish to provide an economic analysis that demonstrates, in sufficient detail for public review, the specific economic effects, parameters, and reasoning.

The Nation is not prepared to concede that SCR technology is "cost effective," and the cost analyses performed by the Plants support a contrary conclusion. We await further analysis by USEPA of the costs of compliance. However, should USEPA conclude that SCRs are cost effective, the provisions of the BART Guidelines quoted above clearly allow for consideration of economic concerns when an emission control technology option has the potential to have serious deleterious economic consequences.

A second provision in the BART Guidelines lays out the process for performing an energy cost assessment as part of the analysis of the energy and non-air quality environmental impacts factor (Factor 2). Section IV(D)(h)(5) provides that "the energy impacts analysis may consider whether there are relative differences between alternatives regarding the use of locally or regionally available coal, and *whether a given alternative would result in significant economic disruption or unemployment* (emphasis added)." As discussed below, loss of one or both of the Plants would likely result in the closure of the mine that supplies its coal.

Finally, as an agency of the federal government, USEPA has a responsibility to carry out the trust responsibility of the United States to Indian tribes. EO 13175 requires federal agencies to consult with Indian tribes on federal actions with tribal implications. "Policies that have tribal implications," are defined to include the promulgation of regulations "that have substantial direct effects on one or more Indian tribes."<sup>12</sup> USEPA is clearly required by EO 13175 to consult with the Nation and other affected tribes on the instant rulemaking to determine BART for NGS and FCPP. Section 5(a) of the EO, describes the consultation requirement: "[e]ach agency shall have an accountable process to ensure meaningful and timely input by tribal officials in the development of regulatory policies that have tribal implications." Section 5(d) continues: "each agency should explore and, where appropriate, use consensual mechanisms for developing regulations, including negotiated rulemaking." USEPA's obligation to solicit comments on any proposed action is distinct from but related to its trust obligation, as a federal agency, to consult with affected Indian tribes.<sup>13</sup> It is the Nation's position that USEPA, as part of this consultative process, must do more than simply listen and respond to the Nation's comments. To the extent that the Nation recommends a particular action, in this instance, the selection of a particular control technology as BART for power plants located on the Nation's lands, USEPA must give substantial weight to that recommendation as part of its decision-making process.

USEPA must consult with the Navajo Nation and must consider the potential economic impacts of its actions when making its BART determination. In reliance on the authorities cited above, the Nation presents as part of its comments on the ANPR a discussion of its economic interests at stake in this rulemaking.

<sup>12</sup> EO 13175, § 1(a).

<sup>13</sup> *Id*. § 9(a).

#### ECONOMIC INTERESTS OF THE NAVAJO NATION

The 2005-2006 Comprehensive Economic Development Strategy of the Navajo Nation<sup>14</sup> ("Strategy Report"), summarizes Navajo Nation economic data including budget figures, primary sources of revenue, major employers, and employment figures. FCPP and NGS are listed among the largest employers within the Nation. During the period covered by the Strategy Report, FCPP employed 586 people, 72% of whom were members of the Nation, with an annual payroll of \$41 million.<sup>15</sup> Employment at NGS included 512 permanent employees, 69% of whom were Native American. The Plant employed approximately 300 additional seasonal employees, with 93% being Native American. NGS had a total annual payroll of \$47 million.<sup>16</sup>

The Plants are linked inextricably with the coal mines that supply them and the economic benefits indirectly attributable to the Plants include mine employment, payroll and royalty revenue for the Nation. FCPP burns approximately 10 million tons of coal annually from the BHP Navajo Mine.<sup>17</sup> Revenues to the Nation in the form of royalties and taxes paid by the Navajo Mine are approximately \$45.9 million.<sup>18</sup> The Navajo Mine is also a major employer on the Navajo Nation, with 357 employees, 87% of whom are Navajo tribal members. Salary and benefits paid by the Navajo Mine exceeded \$41 million in 2004.<sup>19</sup> The Peabody Kayenta Mine delivers approximately 8.3 million tons of coal to NGS.<sup>20</sup> The Kayenta Mine employs approximately 400 workers, a large percentage of whom are Native American, with a payroll of \$44.4 million.<sup>21</sup>

In an economy where unemployment has hovered just below 50% for years,<sup>22</sup> the hundreds of skilled jobs provided by the Plants and the Mines are critical to the Nation's fiscal well-being. Given that the number of people employed within the Nation has remained

<sup>14</sup> Available at http://www.navajobusiness.com/pdf/CEDS/CEDS%202005%20-%2006%20Final.pdf
 <sup>15</sup> Strategy Report at 28.

- <sup>15</sup> Id.
- <sup>17</sup> *Id.* at 28.

<sup>18</sup> BHB Billiton New Mexico Coal Operations: Calendar Year 2008, available at www.whitehouse.gov/omb.asset.aspx?AssetId=2072.

<sup>19</sup> Strategy Report at 29.

<sup>20</sup> Id. at 31.

<sup>21</sup> Id.

<sup>22</sup>/d, at 14.

essentially constant at about 30,000,<sup>23</sup> together the permanent Plant and Mine employees make up about 6% percent of the total Navajo Nation workforce. In 2008, cumulative revenues from the Plants and the Mines that supply them totaled approximately \$93.9 million, accounting for about 55% of the Nation's General Funds Budget of \$172 million.<sup>24</sup>

There is one additional concern unique to NGS. NGS is owned in part by the United States acting through the Bureau of Reclamation ("BOR"). Energy generated by NGS and attributed to BOR's ownership share is used in multiple ways to subsidize the Central Arizona Project ("CAP"), which delivers Colorado River water for domestic, municipal, industrial and agricultural uses throughout central and southern Arizona. Pursuant to the Arizona Water Settlements Act of 2004, P.L. 108-451, revenues generated by the sale of power exceeding that needed to deliver CAP water may be used to fund the costs of Indian water rights settlements in Arizona. The Nation is currently engaged in negotiations to settle its water rights claims in the Lower Colorado River Basin, and will look to these funds should it reach a settlement of these water rights claims in the state. Further, any settlement of the Nation's water rights claims in Arizona would likely also involve delivery of CAP water, and the Nation has an interest in keeping energy rates for delivery of CAP water at an economical level.<sup>25</sup> Requiring emissions controls more extensive, and more costly, than those required to comply with the Regional Haze Rule, has the potential to seriously undermine the financial stability of NGS, and threaten the BOR revenue stream upon which both existing and future Indian water rights settlement in Arizona depend.

The Strategy Report provides the following commentary on the impacts of the closure of the Mohave Generating Station on the Navajo Nation:

Because of EPA regulations, the Mohave Generating Station near Laughlin, Nevada, closed its operations. As the power plant was the sole buyer of coal from Black Mesa Mine, it had to close its operation on January 1, 2006. Closure of this mine will definitely have very adverse economic impact [*sic*] not only on the 160 or so people laid off from the mine, but also on the Navajo Nation coffers.

The Nation has already suffered the ripple effects of one USEPA rulemaking that, through the imposition of financially untenable emissions controls, resulted in the closure of

<sup>&</sup>lt;sup>23</sup> Id.

<sup>&</sup>lt;sup>24</sup> General Funds are those generated from sources internal to the Nation, as opposed to external sources, principally federal and state grants.

<sup>&</sup>lt;sup>25</sup> The Nation's water rights settlement in New Mexico, ratified by Congress in P.L. 111-11, includes provisions for the delivery of CAP water through the Navajo Gallup Water Supply Project to the Nation's capitol in Window Rock.

the Mohave Generating Station, and as a consequence, the closure of the Black Mesa Mine. If either NGS or FCPP were to close as the result of the imposition of cost-prohibitive emission controls, the mine supplying coal to the Plant would also close. Revenue and job losses of that magnitude would be cataclysmic. The ripples would grow exponentially and their impact on the Nation would be more akin to a tsunami. USEPA must factor the economic impacts of any proposed action into its BART analysis.

#### II. BART FACTORS ADDRESSED BY THE ANPR

In response to the ANPR's request for comments, the Nation takes this opportunity to express its serious concerns about both the methods used by USEPA to challenge the cost estimates of the Plants and inputs proposed to be used by USEPA to determine the degree of visibility improvement associated with the emission control technologies under consideration.

#### FACTUAL BACKGROUND/REGULATORY CONTEXT

The Navajo Nation acknowledges that FCPP and NGS emit significant amounts of pollutants that the CAA seeks to control. Plants like NGS and FCPP are the focus of BART proceedings precisely because they are large and aging. EPA recites that FCPP is the largest emitter of NO<sub>x</sub> and NGS the fourth largest in the United States. However, these rankings are based on gross emissions measured in tons per year. A more meaningful ranking is achieved when emissions are considered in the context of the size of the Plants and the energy produced, measured in pounds per MMBtu. Using the same database relied upon by USEPA,<sup>26</sup> but taking into account the size of the Plants, NGS, for example, ranks number 167 out of 1,199 coal and natural gas plants, and drops below the top 10% of polluters. When compared to only coal-fired power plants, it ranks number 156 out of 414, or in the 38<sup>th</sup> percentile.<sup>27</sup>

Further, coal-fired power plants are relatively small contributors to regional haze in the surrounding Class I Areas. The Grand Canyon Visibility Transport Commission ("GCVTC") studied the numerous sources of regional haze, including both man-made (stationary sources, mobile sources, area sources, dust from dirt roads and prescribed forest burns) and naturally occurring sources (Rayleigh scattering, smoke from wildfires, wind-blown dust and volcano eruptions). The Western Regional Air Partnership ("WRAP"), successor to the GCVTC, concluded that visibility impairment caused by PM and attributable to **all stationary sources** is probably

<sup>&</sup>lt;sup>26</sup> EPA "Clean Air Markets – Data and Maps", available at http://camddataandmaps.epa.gov/gdm/

<sup>&</sup>lt;sup>27</sup> These calculations were performed by NGS at the Nation's request.

less than 2%, and that stationary source NO<sub>x</sub> emissions probably cause between 2 and 5 percent of the visibility impairment on the Colorado Plateau.<sup>28</sup>

Finally, it is important to note that western states, in implementing the Regional Haze Rule, have largely rejected SCR technology due to its cost. For example, the Colorado State Legislature banned the consideration of SCR technology as BART for the control of  $NO_x$  emissions, even when presumptive emission limits established in the BART Guidelines would not be met.<sup>29</sup>

The Nation recites these facts in support of its position that the BART component of the CAA and Regional Haze Rule was meant to provide for a measured response to emissions from aging power plants. Requiring the most expensive controls is inconsistent with the law and regulations governing the BART process, especially when, as discussed below, the incremental improvement attributable to such controls has not been analyzed. Further, requiring two power plants over which USEPA has exclusive jurisdiction to bear a greater regulatory burden than similarly situated plants regulated by the states is contrary to the purposes of the Act, the Regional Haze Rule, and to the economic interests of the Navajo Nation.

#### COST OF COMPLIANCE (Factor 1)

The cost analysis for the emission control options being considered as BART for the Plants is the most critical of the five factors used to determine BART. Only with accurate cost projections can USEPA determine the cost effectiveness of any control technology. The disparity between the projected costs presented by the Plants and by USEPA, with the assistance of the National Park Service ("NPS"), demonstrates the mischief that can be accomplished if the cost numbers relied upon are not substantiated to the greatest possible extent. The Nation is not fully satisfied with the cost analysis performed by the Plants or USEPA. The Nation's concerns regarding the cost estimates submitted by the Plants stem

<sup>&</sup>lt;sup>28</sup> See Stationary Source No<sub>x</sub> and PM Emissions in the WRAP Region: An Initial Assessment of Emissions, Controls, and Air Quality Impacts, Final Report of the WRAP Market Trading Forum (October 1, 2003), available at: www.wrapair.org/forums/mtf/nox-pm.html.

<sup>&</sup>lt;sup>29</sup>5 CCR 1001-5, Reg. 3, Part E § IV(B) ("Electric Generating Units and Fossil Fuel Boilers do not need to consider post combustion controls for NOx purposes in the BART analysis and **the Division may not require post combustion controls for NOx purposes for Electric Generating Units** and Fossil Fuel Boilers (emphasis added.); See also, WRAP BART Status (December 12, 2009), available at:

http://www.wrapair.org/forums/ssjf/documents/bart/2009-12\_BART\_Status\_Document.pdf. The state of Oregon is taking a phased approach to PGE's Boardman Plant, a 600 Mw facility. Initially Boardman will be required to install LNB and SOFA technology. In a second phase, Boardman will be required to install SCR technology in 2017. *Id*.

primarily from the lack of transparency in the cost projections. While the Nation understands that the Plants' cost projections are premised in part on proprietary information, failure to fully document projected costs makes it difficult for the Nation to independently assess the cost analyses prepared by the Plants. The concerns that Nation has with USEPA's cost analysis are set forth below. The Nation urges USEPA to retain an independent industry expert to analyze the costs associated with the retrofit control technologies being considered as BART for NGS and FCPP.

USEPA and NPS rely heavily on the *EPA Air Pollution Control Cost Manual* (6<sup>th</sup> Ed. 2001)<sup>30</sup>("Manual") in their cost analyses, consequently, the Nation's comments on the cost of compliance analysis begin with an examination of the Manual. The latest version of the Manual was released in 2001. As the Memorandum accompanying the release of the Manual acknowledges, the latest revision contains "updated pricing information to at least 1990."<sup>31</sup> Section 4.2 of the Manual, setting forth a cost-estimating methodology for SCR systems is presented in "1998 dollars."<sup>32</sup> In meetings between the Nation and representatives of NGS and FCPP, both Plant operators expressed dismay at the reliance on such a clearly outdated source. The Nation shares this concern. Further, the Nation is of the opinion that the experts used by the Plants to develop their cost estimates likely have access to information about the design of the Plants and costs associated with retrofitting the Plants that is unavailable to USEPA/NPS, and that more accurately portrays the real world costs to be incurred by the Plants to install emission controls.

A second concern is that USEPA/NPS are not applying the Manual in a manner consistent with the BART Guidelines, which provide:

The basis for equipment cost estimates also should be documented either with data supplied by an equipment vendor (*i.e.* budget estimates or bids) or by a referenced source (such as the [Manual]). In order to maintain and improve consistency, cost estimates should be based on the [Manual], where possible. The [Manual] addresses most control technologies in sufficient detail for a BART analysis. The cost analysis should also take into account any site-specific design or other conditions identified above that affect the cost of a particular BART technology option.<sup>33</sup>

<sup>&</sup>lt;sup>30</sup> Formerly the OAQPS [EPA Office of Air Quality Planning and Standards] Control Cost Manual.

<sup>&</sup>lt;sup>31</sup> See Memorandum from Daniel Mussatti (Nov. 1, 2001) available at: http://epa.gov/ttn/catc/products.html.

<sup>&</sup>lt;sup>32</sup> Manual, Section 4.2, NO<sub>x</sub> Post-Combustion at 2-40. Further, the Manual explains that the "costs and estimating methodology in this Manual are directed toward the 'study' estimate with a nominal accuracy of  $\pm$  30% percent." *Id.* at 2-3.

<sup>&</sup>lt;sup>33</sup> BART Guidelines at Section IV(D)(4)(a)(5).

Clearly, the BART Guidelines anticipate both that sources other than the Manual may be used to establish costs, and that factors not fully addressed in the Manual, such as site-specific features, may be included in a cost of compliance analysis. In fact, USEPA confirmed this interpretation of the Manual in the preamble to the Federal Register notice published with the Guidelines:

We believe that the Control Cost Manual provides a good reference tool for cost calculations, but if there are elements or sources that are not addressed by the Control Cost Manual or there are additional cost methods that could be used, we believe that these could serve as useful supplemental information.<sup>34</sup>

USEPA disallowed, and NPS excluded, costs that appear to fall within the purview of this provision of the Guidelines.

The Nation has the following specific concerns about the revisions to the Plants' cost analyses:

- USEPA should accept the annual to capital cost ratios employed by each Plant. The ANPR does not offer adequate justification for USEPA's decision to use a cost ratio derived from other entities in neighboring states. After meeting with representatives of the Plants, the Nation is satisfied that the different ratios adopted by NGS and FCPP reflect differences in the Plants' business structures and should be retained unless USEPA provides additional support for its decision.
- Similarly, USEPA does not offer adequate justification for the proposal to install a half SCR on NGS Unit 2. Without further technical review, the Nation is not convinced that a half SCR is a realistic option for NGS.
- 3) Finally, the Regional Haze Rule requires not only a determination of cost effectiveness, but also a consideration of incremental cost-effectiveness between the control technologies considered.<sup>35</sup> Given the great differences in costs projected by the Plants between combustion controls and SCRs, an incremental cost analysis should be performed. The preamble in the Federal Register notice publishing the revised Regional Haze Rule and BART Guidelines make this clear:

<sup>&</sup>lt;sup>34</sup> FR Notice at 39127.

<sup>&</sup>lt;sup>35</sup> BART Guidelines, Section IV(D)(4)(b) and (e).

[T]he guidelines continue to include both average and incremental costs. We continue to believe that both average and incremental costs provide information useful for making control determinations. However, we believe that these techniques should not be misused. For example, a source may be faced with a choice between two available control devices, control A and control B, where control B achieves slightly greater emission reductions. The average cost (total annual cost/total annual emission reductions) for each may be deemed to be reasonable. However the incremental cost (total annual cost<sub>A-B</sub>/total annual emission reductions to be achieved by control B may be very great. In such an instance, it may be inappropriate to choose control B, based on its high incremental costs, even though its average cost may be considered reasonable.<sup>36</sup>

The failure to include an incremental cost analysis in the ANPR may have been a result of the limited scope of the ANPR, and may not reflect a decision that an incremental analysis is not required. The Nation includes a discussion of incremental costs because of the importance an incremental cost effectiveness analysis is likely to have in the final BART determination.

#### VISIBILITY IMPROVEMENT (FACTOR 5)

The BART Guidelines provide that a source/state would be well served to seek approval from USEPA prior to engaging in modeling to determine visibility improvement.<sup>37</sup> The Nation has been informed that the operators of both NGS and FCPP not only sought, but secured USEPA approval, for their modeling protocols. Nevertheless, in the ANPR, USEPA/NPS have recalculated projected visibility improvement after changing important inputs to the models. The Nation has concerns both about the process used by USEPA in the rulemaking to date and about the following changes to inputs USEPA proposes for use in the CALPUFF visibility impact model:

1. Ammonia Background Levels. Ammonia is a precursor to particulate ammonium sulfate and ammonium nitrate, particles that degrade visibility and are referred to as secondary PM. There is very little available data on ammonia background levels at

<sup>&</sup>lt;sup>36</sup>FR Notice at 39127.

<sup>&</sup>lt;sup>37</sup> "In developing your modeling protocol, you may want to consult with EPA and your regional planning organization (RPO). Upfront consultation will ensure that key technical issues are addressed before you conduct your modeling." BART Guidelines, Section III(A)(3).

> Four Corners Class I Areas, the exception being Mesa Verde. In conducting visibility improvement analyses, the Plants relied on ammonia background levels previously accepted by USEPA for the Four Corners region. USEPA proposes to substitute inputs in the CALPUFF model with inputs derived by back-calculating background ammonia from concentrations measured by the Interagency Monitoring of Protected Visual Environments ("IMPROVE") monitoring network covering Class I Areas. USEPA does not adequately justify the use of this back-calculation method for deriving background ammonia levels. NGS proposed to install, and now has installed, monitors capable of recording background ammonia levels. While USEPA discouraged NGS from monitoring for background ammonia, the Nation supports the use of recorded data where possible and urges USEPA to make use of such data as it becomes available.

- HCl and HF Emissions. The Plants both assumed that H<sub>2</sub>SO<sub>4</sub> was the only contributor to condensable organic PM. NPS proposed that HCl and HF be added to this category of pollutants. USEPA took a different route, and included HCl and HF as modeling inputs to determine PM fine. The ANPR does not adequately justify the addition of these compounds as modeling inputs.
- 3. SO<sub>4</sub> condensable inorganic PM (H<sub>2</sub>SO<sub>4</sub>). EPA and both the Plants relied on the H<sub>2</sub>SO<sub>4</sub> calculation methodology provided by the Electric Power Research Institute (EPRI, 2008). However, there is a large discrepancy in the penetration factor USEPA and the Plants each assumed. Both Plants argue that the penetration factor used by USEPA is recommended for utilities burning low-sulfur eastern bituminous coal, and the penetration factor the Plants relied on is appropriate for the coal burned and conditions existing at the Plants. In March of 2009, USEPA requested NGS to perform H<sub>2</sub>SO<sub>4</sub> emissions testing. Again, the Nation encourages USEPA to use recorded data where available. The inputs should be reviewed and the modeling results revised as necessary.

#### CONCLUSION

The Nation is of the opinion that a phased approach to emissions controls at the Plants, beginning with combustion controls, is fully consistent with both the CAA and the Regional Haze Rule. Comments submitted to USEPA to date express considerable disagreement about the interpretation of the data relied upon by both the Plants and USEPA. Many questions about the projected degree of visibility improvement turn on the models used and the inputs chosen

to reach these projections. Installing LNB and SOFA as a first step in a phased approach to BART for the Plants would:

- 1) Be consistent with the presumptive BART limits established in the BART Guidelines;
- 2) Result in a more clearly fact driven process, by allowing all interested parties to review *recorded*, not simply modeled, results at the next review phase.
- Aid in the resolution of disputes over the costs to install the emission control options under consideration by providing the opportunity to review the actual costs incurred by the Plants to install LNB and SOFA, which NGS is already in the process of installing;
- 4) Provide a reasonable timeframe in which to resolve outstanding lease and right of way issues that affect the productive life of the Plants (Factor 4).

The Nation faces conflicting concerns as it considers the question of what level of emissions controls should be determined to be BART for the Plants. The Nation reached the position expressed in these comments only after carefully weighing these interests. While the Regional Haze Rule also entails a balancing of interests, the ANPR fails to consider the significant economic interests it implicates. We recognize that the ANPR is but a first step in this rulemaking process and that USEPA will offer its view of what constitutes BART for the Plants when the Proposed Rule is published. However, USEPA must consider the array of economic impacts any proposed regulatory action may have, not only on the Navajo Nation but on other tribes as well, and the weighing of those interests must be clearly reflected in the Proposed Rule.

The Nation appreciates the efforts that USEPA has taken thus far to comply with its trust obligation to consult with the Nation on matters that have the potential to affect the Nation's resources, including informing the Nation in advance of the publication of the ANPR. We look forward to a continued dialogue as USEPA moves forward with this rulemaking process.

#### THE NAVAJO NATION

/s/ signed on original

Joe Shirley, Jr. President

# Attachment MAS -2

**Execution Copy** 

# PURCHASE AND SALE AGREEMENT

#### BY AND BETWEEN

## SOUTHERN CALIFORNIA EDISON COMPANY, a California corporation

and

ARIZONA PUBLIC SERVICE COMPANY, an Arizona corporation

Dated as of

November 8, 2010

# **TABLE OF CONTENTS**

| ARTICLE 1        | DEFINITIONS  | 1  |
|------------------|--|----|
| 1.1              | Defined Terms  | 1  |
| 1.2              | Index of Other Defined Terms                             |    |
| 1.3              | Interpretation   | 12 |
| <b>ARTICLE 2</b> | PURCHASE AND SALE OF ASSETS                              |    |
| 2.1              | Transfer of Assets                                       | 13 |
| 2.2              | Excluded Assets  | 16 |
| 2.3              | Assumption of Liabilities                                | 18 |
| 2.4              | Excluded Liabilities                                     | 19 |
| 2.5              | Control of Litigation                                    | 20 |
| 2.6              | Carbon Emission Allowances                               | 21 |
| 2.7              | California Capacity Rights                               | 22 |
| ARTICLE 3        | CLOSING  | 23 |
| 3.1              | Closing  | 23 |
| 3.2              | Purchase Price   | 23 |
| 3.3              | Pre-Closing and Post-Closing Adjustments                 | 24 |
| 3.4              | Payment  | 25 |
| 3.5              | Allocation of Purchase Price                             | 25 |
| 3.6              | Prorations   | 26 |
| 3.7              | No Assignment if Breach                                  | 27 |
| 3.8              | Deliveries by Seller                                     | 27 |
| 3.9              | Deliveries by Purchaser                                  | 28 |
| 3.10             | Facilities Contracts                                     | 29 |
| ARTICLE 4        | REPRESENTATIONS, WARRANTIES AND DISCLAIMERS OF<br>SELLER | 29 |
| 4.1              | Organization and Existence                               | 29 |
| 4.2              | Execution, Delivery and Enforceability                   | 29 |
| 4.3              | No Violation   | 29 |
| 4.4              | Compliance with Laws                                     | 30 |
| 4.5              | Permits, Licenses, Etc                                   | 30 |

i

|      | 4.6   | Litigation                                    | . 30         |
|------|-------|---|--------------|
|      | 4.7   | Title   | . 31         |
|      | 4.8   | Facilities Contracts                          | . 31         |
|      | 4.9   | Intellectual Property                         | . 31         |
|      | 4.10  | Taxes   | . <b>3</b> 1 |
|      | 4.11  | Undisclosed Liabilities                       | . 31         |
|      | 4.12  | Brokers                                       | . 31         |
| ARTI | CLE 5 | REPRESENTATIONS AND WARRANTIES OF PURCHASER   | . 32         |
|      | 5.1   | Organization and Existence                    |              |
|      | 5.2   | Execution, Delivery and Enforceability        | . 32         |
|      | 5.3   | No Violation                                  | . 32         |
|      | 5.4   | Compliance with Laws                          | . 33         |
|      | 5.5   | Litigation                                    | . 33         |
|      | 5.6   | Brokers                                       | . 33         |
|      | 5.7   | Financing                                     | . 33         |
|      | 5.8   | Qualified for Permits                         | . 33         |
|      | 5.9   | "AS IS" SALE                                  | . 33         |
| ARTI | CLE 6 | COVENANTS OF EACH PARTY                       |              |
|      | 6.1   | Efforts to Close                              | . 34         |
|      | 6.2   | Updating                                      | . 36         |
|      | 6.3   | Conduct Pending Closing                       | . 36         |
|      | 6.4   | Consents and Approvals                        | . 37         |
|      | 6.5   | Tax Matters                                   | . 38         |
|      | 6.6   | Risk of Loss                                  | . 41         |
|      | 6.7   | Cooperation Relating to Insurance             | . 42         |
|      | 6.8   | Confidentiality                               | . 42         |
|      | 6.9   | Reasonable Cooperation                        | . 43         |
|      | 6.10  | Title to Real Property and Leased Property    | . 43         |
|      | 6.11  | Right of First Refusal                        | . 43         |
|      | 6.12  | Exclusivity                                   | . 44         |
|      | 6.13  | Post Closing – Further Assurances             | . 44         |
|      | 6.14  | Post Closing – Information and Records        | . 44         |
|      | 6.15  | Post Closing – Landfill and Remediation Costs |              |

| ARTICLE 7 | INDEMNIFICATION  | . 45 |
|-----------|--|------|
| 7.1       | Indemnification by Seller  | . 45 |
| 7.2       | Indemnification by Purchaser                                       | . 46 |
| 7.3       | Notice of Claim  |      |
| 7.4       | Defense of Third Party Claims                                      | . 47 |
| 7.5       | Cooperation  | . 47 |
| 7.6       | Mitigation and Limitation on Claims                                | . 47 |
| 7.7       | Exclusivity  | . 48 |
| ARTICLE 8 | CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER<br>AT THE CLOSING | . 48 |
| 8.1       | Compliance with Provisions   | . 48 |
| 8.2       | HSR Act  | . 48 |
| 8.3       | Injunction   | . 48 |
| 8.4       | Required Regulatory Approvals                                      | . 48 |
| 8.5       | Representations and Warranties                                     | . 49 |
| 8.6       | Officer's Certificate  | . 49 |
| 8.7       | Title Policy Insurance   | . 49 |
| 8.8       | Material Adverse Effect  | . 49 |
| 8.9       | Liens  | . 49 |
| 8.10      | Seller's Required Consents   | . 49 |
| 8.11      | No Termination   | . 49 |
| 8.12      | Right of First Refusal and Notice                                  | . 49 |
| 8.13      | Termination Agreement  | . 50 |
| 8.14      | Facilities Lease Amendments  | . 50 |
| 8.15      | Fuel Agreement   | . 50 |
| ARTICLE 9 | CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER AT THE CLOSING       | . 50 |
| 9.1       | Compliance with Provisions   | . 50 |
| 9.2       | HSR Act  | . 50 |
| 9.3       | Injunction   | . 50 |
| 9.4       | Approvals  | . 50 |
| 9.5       | Representations and Warranties                                     | . 51 |
| 9.6       | Officer's Certificate  | . 51 |

iii

| 9.7                    | No Termination   | . 51 |
|------------------------|--|------|
| 9.8                    | Right of First Refusal   | . 51 |
| 9.9                    | Purchaser's Required Consents  | . 51 |
| 9.10                   | Material Adverse Effect  | . 51 |
| 9.11                   | Termination Agreement  | . 51 |
| ARTICLE 10             | TERMINATION  | . 51 |
| 10.1                   | Rights To Terminate  | . 51 |
| 10.2                   | Effect of Termination  | . 52 |
| 10.3                   | Specific Performance; Limitation of Damages  | . 53 |
| ARTICLE 11             | MISCELLANEOUS AGREEMENTS AND ACKNOWLEDGMENTS   | . 53 |
| 11.1                   | Purchaser as Operating Agent   | . 53 |
| 11.2                   | Expenses   | . 53 |
| 11.3                   | Entire Document  | . 54 |
| 11.4                   | Schedules  | . 54 |
| 11.5                   | Counterparts   | . 54 |
| 11.6                   | Severability   | . 54 |
| 11.7                   | Assignability  | . 54 |
| 11.8                   | Captions   | . 54 |
| 11.9                   | Governing Law  | . 54 |
| 11.10                  | Dispute Resolution   | . 54 |
| 11.11                  | Notices  | . 57 |
| 11.12                  | Time is of the Essence   | . 58 |
| 11.13                  | No Third Party Beneficiaries   | 58   |
| 11.14                  | No Joint Venture   | . 59 |
| 11.15                  | Construction of Agreement  | 59   |
| 11.16                  | Effect of Closing Over Known Unsatisfied Conditions or Breached<br>Representations, Warranties or Covenants          | . 59 |
| 11.17                  | Conflicts  | . 59 |
| 11.18                  | Waiver of Compliance   | . 59 |
| 11.19                  | Survival   | 59   |
| Schedules              |  |      |
| 1.1.50(a)<br>1.1.50(b) | "Seller's Officers, Employees, and Knowledgeable Persons"<br>"Purchaser's Officers, Employees and Authorized Agents" |      |

1.1.50(c) "Operating Agent's Officers, Employees and Authorized Agents"

iv
| 1.1.62      | "PNW Plans"                                     |
|-------------|---|
| 1.1.67      | "Purchaser's Required Consents"                 |
| 1.1.68      | "Purchaser's Required Regulatory Approvals"     |
| 1.1.77      | "Seller's Required Consents"                    |
| 1.1.78      | "Seller's Required Regulatory Approvals"        |
| 2.1(b)      | "Leased Real Property"                          |
| 2.1(c)      | "Rights-of-Way/Easements and Water Rights"      |
| 2.1(h)      | "Seller Facilities Contracts"                   |
| 2.1(p)      | "Miscellaneous Assets"                          |
| 2.2(a)      | "Excluded Assets"                               |
| 3.6(a)(iii) | "Operating and Maintenance Expense Pro-Rations" |
| 6.5(g)      | "Pollution Control Bonds"                       |
|             |   |

Exhibit A Landfill

## PURCHASE AND SALE AGREEMENT

This PURCHASE AND SALE AGREEMENT is made as of November 8, 2010, by and between SOUTHERN CALIFORNIA EDISON COMPANY, a California corporation ("Seller"), and ARIZONA PUBLIC SERVICE COMPANY, an Arizona corporation ("Purchaser").

## BACKGROUND

A. Seller desires to sell to Purchaser certain assets, which constitute all of Seller's participation interests in the fossil fuel generating facility known as the Four Corners Power Plant and certain other facilities and assets associated therewith or ancillary thereto, and Purchaser desires to purchase these assets from Seller, all on the terms and conditions hereinafter set forth;

**B.** Seller and Purchaser are entering into this Agreement to evidence their respective duties, obligations and responsibilities;

NOW, THEREFORE, in consideration of the respective representations, warranties, covenants and agreements contained in this Agreement, Seller and Purchaser, intending to be legally bound, hereby agree as follows:

### ARTICLE 1 DEFINITIONS

1.1 <u>Defined Terms</u>. The following terms when used in this Agreement (or in the Schedules and Exhibits to this Agreement) with initial letters capitalized have the meanings set forth below:

**1.1.1** <u>ACC</u>. "ACC" means the Arizona Corporation Commission or its regulatory successor, as applicable.

1.1.2 <u>Affiliate</u>. "Affiliate" of a Person means any other Person that (a) directly or indirectly controls the specified Person; (b) is controlled by or is under direct or indirect common control with the specified Person; or (c) is an officer, director, employee, representative or agent or subsidiary of the Person. For the purposes of this definition, "control," when used with respect to any specified Person, means the power to direct the management or policies of the specified Person, directly or indirectly, whether through the ownership of voting securities, partnership or limited liability company interests, by contract or otherwise.

**1.1.3** <u>Agreement</u>. "Agreement" means this Purchase and Sale Agreement, together with the Schedules and Exhibits hereto.

**1.1.4** <u>Ancillary Agreements</u>. "Ancillary Agreements" means the Deed, the Bill of Sale, the Assignment and Assumption Agreement and any other agreement to be executed and delivered by the Parties under this Agreement.

1.1.5 <u>Article</u>. "Article" means a numbered article of this Agreement. An Article includes all the numbered sections of this Agreement that begin with the same number as that Article.

**1.1.6** <u>Assets</u>. "Assets" has the meaning set forth in <u>Section 2.1</u>.

1.1.7 <u>Assignment and Assumption Agreement</u>. "Assignment and Assumption Agreement" means the assignment and assumption agreement between Seller and Purchaser, to be delivered at the Closing, in such form as shall be reasonably acceptable to Seller and Purchaser, pursuant to which Seller shall assign to Purchaser all of Seller's right, title and interest in and to the Facilities Contracts, certain intangible assets and certain other Assets, and Purchaser shall accept such assignments and assume the Assumed Liabilities.

**1.1.8** <u>Assumed Liabilities</u>. "Assumed Liabilities" has the meaning set forth in <u>Section 2.3</u>.

**1.1.9** <u>Bill of Sale</u>. "Bill of Sale" means the bill of sale from Seller to Purchaser, to be delivered at the Closing, in such form as shall be reasonably acceptable to Seller and Purchaser.

1.1.10 <u>Business Day</u>. "Business Day" means a day other than Saturday, Sunday or a day on which banks are legally closed for business in the State of Arizona.

**1.1.11** <u>California ISO</u>. "California ISO" means the Independent System Operator described in Article 3 of Chapter 2.3 of Part 1 of Division 1 of the California Public Utilities Code.

1.1.12 <u>Capital Expenditure</u>. "Capital Expenditure" means any additions to or replacements of property, plant and equipment in accordance with any of the Facilities Contracts.

1.1.13 <u>Carbon Emission Allowance</u>. "Carbon Emission Allowance" means an Emission Allowance or authorization to emit one specified unit of carbon dioxide or, if applicable, another pollutant addressed under Environmental Laws to mitigate global warming or climate change.

1.1.14 <u>Closing</u>. "Closing" has the meaning set forth in <u>Section 3.1</u>.

**1.1.15** <u>Closing Date</u>. "Closing Date" has the meaning set forth in <u>Section 3.1</u>.

1.1.16 <u>Code</u>. "Code" means the Internal Revenue Code of 1986, as amended.

1.1.17 <u>Commercially Reasonable Efforts</u>. "Commercially Reasonable Efforts" means efforts by a reasonable Person in the position of a Party which are designed to enable a Party to satisfy a condition to, or otherwise assist in the consummation of, the transactions contemplated by, or to perform its obligations under, this Agreement and which do

not require the performing Party to expend any funds or assume liabilities other than expenditures and liabilities which are customary and reasonable in nature and amount for transactions like those contemplated by this Agreement.

**1.1.18** <u>Confidential Information</u>. "Confidential Information" has the meaning ascribed to such term in the Confidentiality Agreement.

**1.1.19** <u>Confidentiality Agreement</u>. "Confidentiality Agreement" means that certain Multiparty Confidentiality Agreement by and among Seller, Purchaser and the other Facilities Owners dated August 4, 2009.

**1.1.20** <u>CPUC</u>. "CPUC" means the California Public Utilities Commission, or its regulatory successor, as applicable.

**1.1.21** <u>Decommissioning Report</u>. "Decommissioning Report" means the Final Report Facility-Wide Indicative Demolition Cost Estimate for the Four Corners Power plant issued in December of 2009 by The Shaw Group Power Generation Services.

**1.1.22** <u>Deed</u>. "Deed" means the special warranty deed as customarily used in the state where the Facilities are located pursuant to which Seller will convey all of its right, title and interest in the real property Assets sold to Purchaser under this Agreement, subject to Permitted Encumbrances.

1.1.23 <u>Edison-Arizona Transmission Agreement</u>. "Edison-Arizona Transmission Agreement" means that certain Transmission Agreement between Southern California Edison Company and Arizona Public Service Company executed July 20, 1966, as the same may be amended to the Closing Date.

**1.1.24** <u>Effective Date</u>. "Effective Date" means the date on which this Agreement has been executed and delivered by the Parties.

1.1.25 Emission Allowance. "Emission Allowance" means an authorization to emit one specified unit of pollutant or Hazardous Substance from the Assets, which units are established by the Governmental Authority with jurisdiction over the Assets under (a) an air pollution control and emission reduction program designed to mitigate global warming or climate change or interstate or intrastate transport of air pollutants, (b) a program designed to mitigate environmental impairment of surface waters, watersheds, or groundwater or (c) any pollution reduction program with a similar purpose. Emission Allowances include allowances, as described above, including credits, regardless of whether the Governmental Authority establishing such allowances designates such allowances by a name other than "allowances." Except as specifically addressed in Sections 2.2(1) and 2.6 with respect to Carbon Emission Allowances and Section 2.2(k) with respect to SO<sub>2</sub> Emission Allowances, the amount of the Emission Allowances shall be all Emission Allowances granted to the Facilities or to Seller or Purchaser as a result of its or their ownership interests in the Facilities and in existence and not consumed as of the Effective Date or subsequently authorized in respect of the Assets, reduced by the Emission Allowances consumed in the operation of the Facilities between the Effective Date and the Closing Date in the ordinary course of business.

12193761.3

**1.1.26** <u>Encumbrances</u>. "Encumbrances" means any and all mortgages, pledges, claims, liens, security interests, conditional and installment sales agreements, easements, activity and use restrictions and limitations, exceptions, rights-of-way, deed restrictions, defects of title, encumbrances, and charges of any kind.

**1.1.27** Environment. "Environment" means all soil, real property, air, water (including surface waters, streams, ponds, drainage basins, washes and wetlands), groundwater, water body sediments, drinking water supply, stream sediments or land (including land surface or subsurface strata), fish, plants, wildlife and other biota or other environmental medium or natural resource.

1.1.28 <u>Environmental Condition</u>. "Environmental Condition" means the presence, Release or threatened Release to the Environment of Hazardous Substances, including any migration of Hazardous Substances through the Environment, at, to or from the Facilities or the Facilities Switchyard or the Navajo Mine regardless of when such presence, Release or threatened Release occurred or is discovered. As used in this Agreement, "threatened Release" shall have the meaning ascribed thereto by the Comprehensive Environmental Response, Compensation, and Liability Act (42 U.S.C. § 9607(a)).

Environmental Laws. "Environmental Laws" means all Federal, 1.1.29 state, local and tribal civil and criminal laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders relating to the Environment or human health and welfare, as the same may be amended or adopted, including, without limitation, those relating to Releases or threatened Releases to the Environment or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, Release, threatened Release, transport, disposal or handling of Hazardous Substances, including but not limited to, the Comprehensive Environmental Response, Compensation, and Liability Act (42 U.S.C. § 9601 et seq.), the Hazardous Materials Transportation Act (49 U.S.C. § 1801 et Seq.), the Resource Conservation and Recovery Act (42 U.S.C. § 6901 et seq.), the Federal Water Pollution Control Act (33 U.S.C. § 1251 et seq.), the Clean Air Act (42 U.S.C. § 7401 et seq.), the Toxic Substances Control Act (15 U.S.C. § 2601 et seq.), the Oil Pollution Act (33 U.S.C. § 2701 et seq.), the Emergency Planning and Community Right-to-Know Act (42 U.S.C. § 11001 et seq.), the Oil Pollution Act (33 U.S.C. Sec. 2701 et seq.), the Safe Drinking Water Act (42 U.S.C. Secs. 300f through 300j), the Occupational Safety and Health Act (29 U.S.C. Sec. 651 et seq.), or any similar laws of any Governmental Authority having jurisdiction over the site at which the Assets are located or otherwise applicable to the Assets.

**1.1.30** Excluded Assets. "Excluded Assets" has the meaning set forth in Section 2.2.

**1.1.31 Excluded Liabilities**. "Excluded Liabilities" has the meaning set forth in <u>Section 2.4</u>.

**1.1.32 Exhibits**. **"Exhibits**" means the exhibits to this Agreement.

**1.1.33** <u>Facilities</u>. "Facilities" means the "Four Corners Project," as that term is defined in the Facilities Co-Tenancy Agreement, as well those facilities defined by the

following terms in the Facilities Co-Tenancy Agreement, to the extent they relate to the Four Corners Project, and to the extent such facilities exist, as of the Closing Date: "Existing New Facilities," "Existing Related Facilities," "Future New Facilities," and "Future Related Facilities."

1.1.34 <u>Facilities Co-Tenancy Agreement</u>. "Facilities Co-Tenancy Agreement" means that certain Four Corners Project Co-Tenancy Agreement executed as of July 19, 1966, by and among the Facilities Owners, as the same may be amended to the Closing Date.

**1.1.35** <u>Facilities Contracts</u>. "Facilities Contracts" has the meaning set forth in <u>Section 2.1(h)</u>.

**1.1.36** <u>Facilities Fuel Agreement</u>. "Facilities Fuel Agreement" means the Four Corners Coal Supply Agreement, effective January 1, 2010, between BHP Navajo Coal Company and the Facilities Owners, as the same may be amended to the Closing Date.

1.1.37 <u>Facilities Insurance Policies</u>. "Facilities Insurance Policies" means all insurance policies carried by or for the benefit of the Facilities Owners with respect to the ownership, operation or maintenance of the Facilities or the Facilities Switchyard, including all liability, property damage, self insurance arrangements, retrospective assessments and business interruption policies in respect thereof.

1.1.38 <u>Facilities Lease</u>. "Facilities Lease" means the Indenture of Lease dated December 1, 1960 between the Navajo Tribe of Indians and Purchaser, as amended, supplemented and revised by the Supplemental and Additional Indenture of Lease executed as of July 6, 1966 between the Navajo Tribe of Indians and the Facilities Owners, and as further amended by the Amendment and Supplement No. 1 to the Supplemental and Additional Indenture of Lease dated April 25, 1985 between the Navajo Tribe of Nations and the Facilities Owners, as the same may be amended to the Closing Date.

1.1.39 <u>Facilities Operating Agreement</u>. "Facilities Operating Agreement" means that certain Four Corners Project Operating Agreement entered into as of May 15, 1969, by and among the Facilities Owners, as the same may be amended to the Closing Date.

1.1.40 <u>Facilities Owner</u>. "Facilities Owner" means each Person who, as of the relevant time, is a "Participant" under the Facilities Co-Tenancy Agreement, which, as of the date of this Agreement, means Purchaser, El Paso Electric Company, Public Service Company of New Mexico, Salt River Project Agricultural Improvement and Power District, Seller and Tucson Electric Power Company, in each case in such Person's capacity as a "Participant".

**1.1.41** <u>Facilities Switchyard</u>. "Facilities Switchyard" means the 500 kv and 345 kv switchyards located at and adjacent to the Facilities.

**1.1.42** <u>FERC</u>. "FERC" means the Federal Energy Regulatory Commission as established by the Department of Energy Organization Act of 1977, 42 U.S.C. § 7171, as amended, or its regulatory successor, as applicable.

**1.1.43** <u>FIRPTA Affidavit</u>. "FIRPTA Affidavit" means the Foreign Investment in Real Property Tax Act Certificate and Affidavit of Seller, to be delivered at the Closing.

**1.1.44** <u>Governmental Authority</u>. "Governmental Authority" means any federal, state, local or other government; any governmental, regulatory or administrative agency, commission, body or other authority exercising or entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power; any court or governmental tribunal; and any Tribal Authority; but does not include Purchaser, Seller, any Affiliate thereof, or any of their respective successors in interest or any owner or operator of the Assets (if otherwise a Governmental Authority).

1.1.45 <u>Hazardous Substances</u>. "Hazardous Substances" means (a) any petroleum, asbestos, urea formaldehyde foam insulation and/or transformer or other equipment that contains polychlorinated biphenyls; (b) any chemical, material or substance defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "hazardous constituents," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "toxic pollutants," "contaminants," "pollutants" or "hazardous air pollutants," or words of similar meaning and regulatory effect, under any Environmental Law; and/or (c) any other chemical, material or substance that is listed or regulated under any Environmental Law because it poses a hazard to human health or welfare or the Environment.

**1.1.46** <u>HSR Act</u>. "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended from time to time.

**1.1.47** <u>Income Tax</u>. "Income Tax" means any Tax imposed by any Governmental Authority (a) based upon, measured by or calculated with respect to gross or net income, profits or receipts (including municipal gross receipt Taxes, capital gains Taxes and minimum Taxes) or (b) based upon, measured by or calculated with respect to multiple bases (including corporate franchise Taxes) if one or more of such bases is described in clause (a), in each case together with any interest, penalties or additions attributable to such Tax.

**1.1.48** <u>Independent Accounting Firm</u>. "Independent Accounting Firm" means such nationally recognized, independent accounting firm as is mutually appointed by Seller and Purchaser for purposes of this Agreement.

**1.1.49** <u>Initial Purchase Price</u>. "Initial Purchase Price" means Two Hundred Ninety-Four Million Dollars (\$294,000,000).

**1.1.50** Knowledge. The term "Knowledge" or similar phrases in this Agreement means: (a) in the case of Seller, the extent of the actual and current knowledge of Seller's officers, employees, and knowledgeable persons listed in <u>Schedule 1.1.50(a)</u> at the Effective Date (or, with respect to the certificate delivered pursuant to <u>Section 8.6</u>, the date of delivery of the certificate) without any implication of verification or investigation concerning such knowledge; (b) in the case of Purchaser, the extent of the actual and current knowledge of Purchaser's officers, employees and authorized agents listed in <u>Schedule 1.1.50(b)</u> at the Effective Date (or, with respect to the certificate delivered pursuant to <u>Section 9.6</u>, the date of

delivery of the certificate) without any implication of verification or investigation concerning such knowledge; and (c) in the case of Operating Agent, the extent of the actual and current knowledge of Operating Agent's officers, employees and authorized agents listed in <u>Schedule 1.1.50(c)</u> at the date of this Agreement or at the Closing Date, as well as the Persons who, as of the date of this Agreement or as of the Closing, serve as the plant manager of the Facilities and the Person or Persons to whom the plant manager reports, without any implication of verification or investigation concerning such knowledge.

**1.1.51 <u>Landfill</u>**. "Landfill" means that certain landfill as identified in the sections labeled "LANDFILL" on the map attached as <u>Exhibit A</u> hereto.

**1.1.52** <u>Laws</u>. "Laws" means all Federal, state, local and tribal civil and criminal laws, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders.

1.1.53 Material Adverse Effect. "Material Adverse Effect" means (x) any event, circumstance or condition materially impairing a Party's authority, right, or ability to consummate the transactions contemplated by this Agreement or the Ancillary Agreements, or (y) any change (or changes taken together) in, or effect on, the Assets that is materially adverse to the operations or physical condition of the Facilities and the Facilities Switchyard, taken as a whole, which exist as of the Closing, including an unscheduled shutdown that is materially adverse to the operations or physical condition of the Assets following the Closing, but excluding (a) any change (or changes taken together) generally affecting the international, national, regional or local electric industry as a whole and not affecting the Assets in any manner or degree materially different than other facilities like the Facilities, (b) any change (or changes) resulting from the international, national, regional or local markets for fuel used at the Facilities, (c) any change (or changes taken together) in the North American, national, regional or local transmission system, (d) any change (or changes taken together) to the extent constituting or involving an Excluded Asset or Excluded Liability, or (e) any change which is cured (including by the payment of money) before the earlier of the Closing or the termination of the Agreement under Section 10.1.

**1.1.54** <u>Moenkopi Switchyard</u>. "Moenkopi Switchyard" means the 500-kV transmission switching station located at the Moenkopi Substation as defined in the Edison-Arizona Transmission Agreement.

**1.1.55** <u>Navajo Mine</u>. "Navajo Mine" means the coal mine located on the Navajo Nation property that is operated by BHP Navajo Coal Company ("BHP") and that supplies coal to the Facilities under the Facilities Fuel Agreement.

**1.1.56** <u>Operating Agent</u>. "Operating Agent" means Arizona Public Service Company, as operating agent under the Facilities Co-Tenancy Agreement and the Facilities Operating Agreement, or its successor in interest.

**1.1.57** <u>Operating Agent's Actuary</u>. "Operating Agent's Actuary" means the Person acting as the actuary for the Operating Agent with respect to the Facilities, or its successors or assigns, which at the time of this Agreement is Towers Watson & Co.

1.1.58 <u>Party</u>. "Party" means either Seller or Purchaser, as the context requires; "Parties" means, collectively, Seller and Purchaser.

1.1.59 <u>Pension and OPEB Liabilities</u>. "Pension and OPEB Liabilities" means the pension plan accumulated benefit obligation (ABO) and the other post-retirement benefit obligation (APBO) for the Operating Agent and its Affiliates determined in accordance with Statement of Financial Accounting Standards No. 87 (FAS 87), Statement of Financial Accounting Standards No. 106 (FAS 106) and Accounting Standards Codification 715, as amended.

1.1.60 Permitted Encumbrances. "Permitted Encumbrances" means (a) liens for Property Taxes and other governmental charges and assessments which are not yet due and payable, (b) all exceptions set forth in the Preliminary Title Report to the extent deemed approved by Purchaser under Section 6.10, (c) during the period prior to the Closing, the lien of Seller's Mortgage, (d) liens, encumbrances or title imperfections with respect to the Assets created by or resulting from the acts or omissions of Purchaser or Operating Agent, (e) liens, charges, claims, pledges, security interests, equities and encumbrances arising under the Facilities Contracts, or which will be and are discharged or released either prior to, or simultaneously with, the Closing, (f) the Assumed Liabilities, and (g) liens, charges, claims, pledges, security interests, equities and encumbrances that do not apply only and exclusively, to the interest of Seller but that also constitute liens, charges, claims, pledges, security interests, equities or encumbrances upon the interests of the other Facilities Owners in common and/or the Operating Agent, as agent for any of the Facilities Owners and that individually, or in the aggregate, do not constitute a Material Adverse Effect with respect to the Facilities or the Facilities Switchyard other than Material Adverse Effects of which the Operating Agent has Knowledge.

**1.1.61** <u>Person</u>. "Person" means an individual, partnership, joint venture, corporation, limited liability company, trust, association or unincorporated organization, or any Governmental Authority.

**1.1.62 PNW Plan Assets.** "**PNW Plan Assets**" means the fair market value of the Operating Agent's and its Affiliates' investments in its retirement and other post-retirement plans listed on <u>Schedule 1.1.62</u>.

**1.1.63** <u>Preliminary Title Report</u>. "Preliminary Title Report" has the meaning set forth in <u>Section 6.10</u>.

**1.1.64** <u>Property Tax</u>. "Property Tax" means any Tax resulting from and relating to the assessment of real or personal property or a possessory interest in real or personal property by any Governmental Authority.

1.1.65 <u>Purchase Price</u>. "Purchase Price" has the meaning set forth in <u>Section 3.2</u>.

**1.1.66** <u>**Purchaser**</u>. "**Purchaser**" has the meaning set forth in the introductory paragraph of this Agreement.

1.1.67 <u>Purchaser's Required Consents</u>. "Purchaser's Required Consents" means all consents specified in <u>Schedule 1.1.67</u>, which include the consent of any Person (other than a Governmental Authority) necessary for Purchaser's consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, except where the failure to obtain such Person's consent would not have a Material Adverse Effect.

1.1.68 <u>Purchaser's Required Regulatory Approvals</u>. "Purchaser's Required Regulatory Approvals" means all approvals specified in <u>Schedule 1.1.68</u>, which include the approval of the purchase and sale contemplated hereby by (i) the ACC; (ii) the FERC under the Federal Power Act, which approval shall be without conditions or constraints that would limit Purchaser's ability to take delivery and deliver power from the Facilities for purposes of serving Purchaser's retail load or selling at wholesale on terms and conditions reasonably satisfactory to Purchaser; and (iii) any other Governmental Authority with general regulatory authority over Purchaser's consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, except where the failure to obtain such Governmental Authority's approval would not have a Material Adverse Effect.

**1.1.69** <u>Reclamation Report</u>. "Reclamation Report" means the Final Reclamation Closure Plan and Cost Estimate at the Navajo Mine for APS issued in August of 2010 by Marston & Marston, Inc.

**1.1.70** <u>Release</u>. "Release" means any release, spill, leak, discharge, disposal of, pumping, pouring, emitting, emptying, injecting, leaching, dumping, depositing, dispersing, escaping or migration of a Hazardous Substance into, onto or through the Environment or within any building, structure, facility or fixture, including the abandonment or discarding of Hazardous Substances in barrels, drums, or other containers.

"Remediation" means any action of any kind to 1.1.71 Remediation. address an Environmental Condition or Release or threatened Release or the presence of Hazardous Substances on or in the Environment relating to the Facilities, the Facilities Switchyard, the Navajo Mine or any other location at which Hazardous Substances or nonhazardous substances or materials generated or originating at the Facilities were transported. stored or disposed of, including the following: (i) monitoring, investigation, treatment, cleanup, containment, remediation, removal, mitigation, response or restoration work; (ii) obtaining any permits, consents, approvals or authorizations of any Governmental Authority necessary to conduct any such work; (iii) preparing and implementing any plans or studies for such work; (iv) obtaining a written notice, from a Governmental Authority with jurisdiction under applicable Environmental Laws that no material additional work is required by such Governmental Authority; (v) any response to or preparation for, any inquiry, order, hearing or other proceeding by or before any Governmental Authority with respect to any such Environmental Condition, Release or threatened Release or presence of Hazardous Substances, and (vi) any other activities reasonably determined by the Operating Agent of the Facilities or the Facilities Switchyard, as applicable, to be necessary or appropriate or required under Environmental Laws to address an Environmental Condition, the presence, Release or threatened Release of Hazardous Substances on or in the Environment at the Facilities, the Facilities Switchyard, the Navajo Mine or any

other location at which Hazardous Substances or non-hazardous substances or materials generated or originating at the Facilities were transported, stored or disposed of.

**1.1.72 Schedules**. "Schedules" means the schedules to this Agreement.

**1.1.73** <u>Section</u>. "Section" means a numbered section of this Agreement included within the Article that begins with the same number as that section.

1.1.74 § 323 Grants. "§ 323 Grants" means one or more grants of rights-ofway and easements under the Act of February 5, 1948 (62 Stat. 17, 18, 25 U.S.C. § 323-328), the Act of March 3, 1879 (20 Stat. 394, 5 U.S.C. § 485), as amended, and the Acts of July 9, 1832, and July 27, 1868 (4 Stat. 564, 15 Stat. 228. 25 U.S.C. § 2) and such regulations promulgated thereunder, as are applicable, including 25 C.F.R. § 1.2 and 25 C.F.R. Part 169 granted to the Facilities Owners pursuant to the Facilities Lease, as the same may be amended in connection with the Facilities Lease amendments referenced in Section 8.14.

**1.1.75** <u>Seller</u>. "Seller" has the meaning set forth in the introductory paragraph of this Agreement.

**1.1.76** <u>Seller's Mortgage</u>. "Seller's Mortgage" means Seller's First Mortgage Bond Trust Indenture, dated as of October, 1923, as amended.

1.1.77 <u>Seller's Required Consents</u>. "Seller's Required Consents" means all consents specified in <u>Schedule 1.1.77</u>, which include the consent of the trustee under the Seller's Mortgage if required under the Seller's Mortgage, and any Person (other than a Governmental Authority) necessary for Seller's consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, except where the failure to obtain such Person's consent would not have a Material Adverse Effect.

1.1.78 <u>Seller's Required Regulatory Approvals</u>. "Seller's Required Regulatory Approvals" means all approvals specified in <u>Schedule 1.1.78</u>, which include the approval of the purchase and sale contemplated hereby by (i) the CPUC, (ii) the FERC under the Federal Power Act, in form and substance reasonably satisfactory to Seller, (iii) the California ISO, and (iv) any other Governmental Authority with general regulatory authority over Seller or the business and assets represented by the Assets and whose approval is required for Seller's consummation of the transaction contemplated by this Agreement and the Ancillary Agreements, except where the failure to obtain such Governmental Authority's approval would not have a Material Adverse Effect.

1.1.79 Seller's Share of Underfunded/Overfunded Pension and OPEB Liabilities. "Seller's Share of Underfunded/Overfunded Pension and OPEB Liabilities" means the adjusted product of (i) Pension and OPEB Liabilities minus PNW Plan Assets, multiplied by (ii) the proportional share of Pension and OPEB Liabilities related to the Facilities as determined by the Operating Agent's Actuary, multiplied by (iii) an allocation percentage of 34.76%. This product shall reflect an adjustment whereby (a) amounts billed by the Operating Agent to Seller since 1982 related to Pension and OPEB Liabilities shall be assumed to have been invested in PNW Plan Assets since such billed amounts' respective years of payment, and (b) the Operating Agent and its Affiliates shall be deemed to have made contributions in respect of Pension and OPEB Liabilities at at least the same rate as the Facilities Owners, and such contributions shall be assumed to have been invested in PNW Plan Assets.

1.1.80 <u>Tax</u>. "Tax" means any federal, Tribal Authority, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, (including taxes under Code Section 59A), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property (including assessments, fees or other charges based on the use or ownership of real property), personal property, transactional, use, transfer, registration, value added, alternative or add-on minimum, estimated tax, or other tax of any kind whatsoever, including any interest, penalty or addition thereto, whether disputed or not, including, without limitation, any item for which liability arises as a transferee or successor-in-interest.

1.1.81 <u>Tax Return</u>. "Tax Return" means any return, report, information return, declaration, claim for refund, or other document, together with all amendments and supplements thereto (including all related or supporting information), required to be supplied to any Governmental Authority responsible for the administration of Laws governing Taxes.

**1.1.82** <u>Termination Agreement</u>. "Termination Agreement" means the agreement entered into on or about the Effective Date between Seller and Purchaser with respect to the termination of the Edison-Arizona Transmission Agreement.

**1.1.83** <u>Third Party Claim</u>. "Third Party Claim" means a claim by a Person that is not a member of the Seller Group or the Purchaser Group, including any claim for the costs of conducting Remediation or seeking an order or demanding that a Person undertake Remediation.

**1.1.84** <u>Transferable Permits</u>. "Transferable Permits" means all those permits relating to the Facilities or the Facilities Switchyard (and all applications pertaining thereto) which are transferable under applicable law from Seller to Purchaser with or without a filing with, notice to, or consent or approval of any Governmental Authority.

**1.1.85** <u>**Transfer Tax**</u>. "**Transfer Tax**" means any sales Tax, transaction privilege Tax, transaction Tax, conveyance fee, use Tax, stamp Tax, stock transfer Tax or other similar Tax, including any related penalties, interest and additions thereto.

**1.1.86** <u>**Tribal Authority**</u>. **"Tribal Authority**" means any sovereign nation recognized by the United States government, Indian tribe, or any governmental subdivision, agency, department, or instrumentality thereof with the authority to administer and collect Taxes, administer and enforce tribal laws and administer and enforce tribal agency processes.

#### 1.2 Index of Other Defined Terms.

| Defined Term       | Section  |
|--------------------|----------|
| Allocation         | 3.5      |
| Applicable Tax Law | 3.5      |
| Arbitrator         | 11.10(e) |
| BHP                | 1.1.54   |

12193761.3

| Classical Alticutor and                 | 2 2(0)             |
|---|--------------------|
| Closing Adjustment                      | 3.3(a)<br>2.3(h)   |
| Cost Update                             | 2.3(n)<br>2.3(c)   |
| Decommissioning                         | 3.2(b)             |
| Emergency Capital Expenditures          | 3.2(0)<br>3.3(a)   |
| Estimated Adjustment                    | 3.3(a)             |
| Estimated Closing Statement             | 2.3(c)             |
| Excess Decommissioning Costs            | 2.3(b)<br>2.3(h)   |
| Excess Reclamation Costs                | 2.2(h)             |
| Excluded Claims<br>Facilities Documents | 2.2(ii)<br>2.1(j)  |
| Facilities Permits                      | 2.1(j)<br>2.1(i)   |
|   | 3.5                |
| Final Allocation                        | 3.5                |
| Final Pre-Closing Allocation            | 2.1(e)             |
| Fuel Inventory                          | 7.6                |
| Indemnifiable Claim                     | 7.8                |
| Indemnitee                              | 7.3                |
| Indemnitor                              |                    |
| Inventory                               | 2.1(f)<br>11.10(d) |
| JAMS                                    |                    |
| Leased Property                         | 2.1(b)             |
| Mediator                                | 11.10(d)<br>7.3    |
| Notice of Claim                         |                    |
| Owned Real Property                     | 2.1(a)<br>6.12     |
| Participating Owner                     |                    |
| Pollution Control Bonds                 | 6.5(g)             |
| Post-Closing Adjustment                 | 3.3(b)             |
| Post-Closing Statement                  | 3.3(b)             |
| Preliminary Title Report                | 6.10<br>2.2(h)     |
| Proposed Post-Closing Adjustment        | 3.3(b)             |
| Purchaser Claims                        | 7.1(a)             |
| Purchaser Group                         | 7.1(a)             |
| Receiving Party                         | 6.4(e)             |
| Reclamation                             | 2.3(h)             |
| Retained Environmental Liabilities      | 2.4(i)             |
| Seller Claims                           | 7.2(a)             |
| Seller Group                            | 7.2(a)             |
| Seller Permits                          | 4.5                |
| Seller's Facilities Share               | 2.6(a)(i)          |
| SO <sub>2</sub> Emission Allowances     | 2.2(k)             |
| Title Insurer                           | 8.7                |
| Title Policies                          | 8.7                |

**1.3** <u>Interpretation</u>. In this Agreement, unless a clear contrary intention appears:

(a) the singular number includes the plural number and vice versa;

(b) reference to any Person includes such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement, and reference to a Person in a particular capacity excludes such Person in any other capacity;

(c) reference to any gender includes each other gender;

(d) reference to any agreement (including this Agreement), document or instrument means such agreement, document or instrument as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof;

(e) reference to any Article, Section, Schedule or Exhibit means such Article, Section, Schedule or Exhibit to this Agreement, and references in any Article, Section, Schedule, Exhibit or definition to any clause means such clause of such Article, Section, Schedule, Exhibit or definition;

(f) "hereunder," "hereof," "hereto" and words of similar import are references to this Agreement as a whole and not to any particular Section or other provision hereof or thereof;

(g) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding such term;

(h) relative to the determination of any period of time, "from" means "from and including," "to" means "to but excluding" and "through" means "through and including;"

(i) reference to any law (including statutes and ordinances) means such law as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, including rules and regulations promulgated thereunder; and

(j) any agreement, instrument, insurance policy, statute, regulation, rule or order defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument, insurance, policy, statute, regulation, rule or order as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes, regulations, rules or orders) by succession of comparable successor statutes, regulations, rules or orders and references to all attachments thereto and instruments incorporated therein.

## ARTICLE 2 PURCHASE AND SALE OF ASSETS

2.1 <u>Transfer of Assets</u>. Upon the terms and subject to the satisfaction of the conditions contained in this Agreement, at the Closing, Seller will sell, convey, assign, transfer and deliver to Purchaser and Purchaser will purchase and acquire from Seller, all of Seller's interest in the Facilities and the Facilities Switchyard, including Seller's undivided interest therein as a tenant in common, which Seller owns or to which Seller has rights by reason of any of the Facilities Contracts, free and clear of all Encumbrances other than Permitted Encumbrances, including, without limitation, Seller's interest in the following, but excluding all Excluded Assets (collectively, the "Assets"):

(a) <u>Real Property Rights</u>. The parcels of real property (or interests therein), if any, owned by Seller, or by the Operating Agent on behalf of Seller, as one of the Facilities Owners, relating to the Facilities or the Facilities Switchyard, together with all buildings, facilities and other improvements thereon and all appurtenances thereto, including all construction work in process (the "Owned Real Property");

(b) <u>Leased Real Property</u>. The real property leasehold estates and the related lease or sublease agreements, if any, related to the Facilities or the Facilities Switchyard, together with all buildings, fixtures and real property improvements thereon and thereto, including all construction work in process (the "Leased Property"), including, without limitation, the items set forth on <u>Schedule 2.1(b)</u>;

(c) <u>Rights-of-Way/Easements and Water Rights</u>. All rights-of-way, easements, grants and privileges (including all water rights) appurtenant to the Owned Real Property or the Leased Property, including, without limitation, the items set forth on <u>Schedule 2.1(c)</u>;

(d) <u>Equipment</u>. All machinery, mobile or otherwise, equipment (including computer hardware and software and communications equipment), vehicles, tools, fixtures, furniture and furnishings, and other tangible personal property that (i) are not Inventory, (ii) are licensed, owned or leased by Seller, or the Operating Agent, on behalf of the Facilities Owners or on behalf of Seller, as one of the Facilities Owners, as of the Closing, and (iii) are related to, used, or useful, in the operation of the Facilities or the Facilities Switchyard, or are typically located at the Facilities, the Facilities Switchyard, the Navajo Mine or other locations or facilities which are owned, operated, maintained or under the control of the Operating Agent;

(e) <u>Fuel Inventory</u>. All coal under contract or in inventory relating to the operation of the Facilities located at or in transit to the Facilities (the "Fuel Inventory");

(f) <u>Inventory</u>. The following items intended to be consumed at the Facilities or the Facilities Switchyard in the ordinary course of business: inventories of spare parts; maintenance, shop and office supplies; and other similar items of tangible personal property in existence as of the Closing, wherever located, excluding Fuel Inventory (the "Inventory");

(g) <u>Emission Allowances</u>. All Emission Allowances, except for allowances which are to be retained by Seller pursuant to <u>Section 2.2(k)</u> or <u>Section 2.2(l)</u>;

(h) **Facilities Contracts**. Subject to the receipt of necessary consents and approvals, the contracts, agreements, arrangements, licenses and leases of any nature, (i) to which Seller, in its capacity as a Facilities Owner, is a party, including, without limitation, the items set forth on <u>Schedule 2.1(h)</u>, or (ii) to which the Operating Agent, on behalf of the Facilities Owners or on behalf of Seller, as one of the Facilities Owners, is a party, and by or to which Seller, the Facilities, or the Facilities Switchyard are bound or subject, in each case relating to the ownership, lease, maintenance or operation of the Facilities or the Facilities Switchyard (the "Facilities Contracts"); provided that Seller shall retain all rights under the Facilities Contracts with respect to any Excluded Assets or Excluded Liabilities;

(i) <u>Permits, Licenses, Etc</u>. Subject to the receipt of necessary consents and approvals, the Transferable Permits and any other permits, licenses, approvals, registrations, franchises, certificates, other authorizations and consents of Governmental Authorities relating to the ownership, lease, maintenance or operation of the Facilities or the Facilities Switchyard that, in each case, as of the Closing are in favor of the Facilities Owners, or the Operating Agent, as agent for the Facilities Owners, except for and to the extent that such licenses, permits, approvals, registrations, franchises, certificates, other authorizations and consents relate to Excluded Assets (the "Facilities Permits");

(j) **Documents.** The books, records, materials, documents, information, drawings, reports, operating data, operating safety and maintenance manuals, inspection reports, engineering design plans, blueprints, specifications, and procedures and similar items (i) located at and relating to the Facilities or the Facilities Switchyard or (ii) otherwise relating to the Facilities Switchyard and owned by the Facilities Owners in common or by the Operating Agent as agent for the Facilities Owners (the "Facilities Documents"); provided that Seller may retain, at its own expense, and may use subject to any confidentiality obligations that may apply to the Facilities Owners, copies of any Facilities Documents related to any Excluded Assets or Excluded Liabilities;

(k) <u>Third Party Warranties</u>. All unexpired, transferable warranties and guarantees from third parties with respect to the Facilities or the Facilities Switchyard or arising out of the Facilities Contracts or any contracts entered into thereunder, except to the extent they relate to Excluded Assets or Excluded Liabilities;

(1) <u>Intellectual Property</u>. All intangible assets of an intellectual property nature, including all patents and patent rights, trademarks and trademark rights, inventions, trade names and copyrights relating to the Facilities or the Facilities Switchyard, including the name of the Facilities and the Facilities Switchyard and all pending applications therefor, together with any trade secrets relating to the Facilities or the Facilities Switchyard, in each case that are owned in common by the Facilities Owners or by the Operating Agent as agent for the Facilities Owners;

(m) <u>Claims, Rights and Causes of Action</u>. All rights in, to and under (i) any claims, rights or causes of action against any third parties (including indemnification, contribution and insurance claims) relating to the Assets or the Assumed Liabilities, whether occurring prior to, on or after the Closing, if any, including any claims for refunds, prepayments, offsets, recoupment, insurance proceeds, condemnation awards, judgments and the like; whether received as payment or credit against future liabilities, and (ii) any actual or potential claim or cause of action as a Facilities Owner against the Operating Agent, whether known or unknown, contingent or accrued, arising prior to and in existence at the Closing, except in each case for Excluded Claims;

(n) <u>Prepayment</u>. Advance payments, prepayments, prepaid expenses, deposits and the like (i) made by Seller or the Operating Agent on Seller's behalf in the ordinary course of business prior to the Closing specifically with respect to the Facilities or the Facilities Switchyard, (ii) which exist as of the Closing and (iii) with respect to which Purchaser will receive the benefit after the Closing;

(o) <u>Insurance Proceeds</u>. The right to any proceeds from insurance policies to the extent covering the Assets or the Assumed Liabilities, except for Excluded Claims; and

(p) <u>Miscellaneous</u>. Any miscellaneous assets necessary, useful or used in or ancillary to operating the Facilities or the Facilities Switchyard and primarily utilized in connection therewith but not otherwise enumerated above, including, without limitation, the assets specified on <u>Schedule 2.1(p)</u>, except for Excluded Assets, which in the ordinary course of business are typically located at the Facilities, the Facilities Switchyard, the Navajo Mine or other locations or facilities which are owned, operated, maintained or under the control of the Operating Agent or one of its Affiliates.

2.2 Excluded Assets. Nothing in this Agreement will constitute or be construed as conferring on Purchaser, and Purchaser is not acquiring, any right, title or interest of Seller in or to the following (the "Excluded Assets"), except to the extent Seller owns an interest in any such physical assets as a tenant in common with the other Facilities Owners, in which event such interests in such assets are Assets:

(a) the assets listed or described on <u>Schedule 2.2(a)</u>, which are associated with the Assets but are specifically excluded from the sale;

(b) certificates of deposit, shares of stock, securities, bonds, debentures, evidences of indebtedness, and interests in joint ventures, partnerships, limited liability companies and other entities;

(c) all cash, cash equivalents, bank deposits, accounts and notes receivable (trade or otherwise), except for such assets on deposit with, or under the control of, the Operating Agent;

Affiliates;

(d) any and all data and information pertaining to customers of Seller or its

(c) rights in, to and under all agreements and arrangements of any nature, which are not assigned to Purchaser under the terms of this Agreement, including any agreements for the sale by Seller of energy, capacity or ancillary services from the Facilities prior to the Closing, and any trade accounts receivable and all collateral, security arrangements, notes, bonds, and other evidences of indebtedness of and rights to receive payments arising out of or related to such sales, including any rights with respect to any third party collection procedures or any other actions or proceedings which have been commenced in connection therewith;

(f) rights arising under this Agreement or any instrument or document executed and delivered pursuant to the terms hereof;

(g) any and all books and records not described in <u>Section 2.1(j)</u>;

(h) any rights in, to and under (i) any claims, rights or causes of action against any third parties (including indemnification, contribution and insurance claims) relating to the Excluded Assets or the Excluded Liabilities, whether occurring prior to, on or after the Closing, if any, including any claims for refunds, prepayments, offsets, recoupment, insurance proceeds, condemnation awards, judgments and the like; whether received as payment or credit against future liabilities and (ii) any actual or potential claim or cause of action as a Facilities Owner against the Operating Agent, whether known or unknown, contingent or accrued, arising prior to and in existence at the Closing relating to the Excluded Assets or the Excluded Liabilities ("Excluded Claims");

(i) all privileged or proprietary books, records, materials, documents, information, drawings, reports, operating data, operating safety and maintenance manuals, inspection reports, engineering design plans, blueprints, specifications, and procedures and similar items not owned by the Facilities Owners in common or by the Operating Agent as agent for the Facilities Owners and any and all rights to use the same, including, without limitation, intangible assets of an intellectual property nature such as trademarks, service marks and trade names (whether or not registered), computer software that is proprietary to Seller, or the use of which under the pertinent license therefor is limited to operation by Seller or its Affiliates or on equipment owned by Seller or its Affiliates;

(j) the right to receive mail and other communications relating to any of the Excluded Assets or Excluded Liabilities, all of which mail and other communications shall be promptly forwarded by Purchaser to Seller;

(k) Emission Allowances for sulfur dioxide  $(SO_2)$  ("SO<sub>2</sub> Emission Allowances") related to Seller's share of the Facilities that are of past vintage as of the Closing Date and either: (i) already distributed to Seller as of the Closing Date; or (ii) in excess of the amount needed to cover the Facilities' SO<sub>2</sub> emissions corresponding to Seller's ownership interest in the previous calendar year, but not yet distributed to Seller as of the Closing Date; and Seller's share of the proceeds from any United States Environmental Protection Agency auction of SO<sub>2</sub> Emission Allowances related to the Facilities occurring before the Closing Date, even if such proceeds have not yet been distributed as of the Closing Date;

(1) any Carbon Emission Allowances or rights thereto retained by Seller under <u>Section 2.6;</u>

(m) properties of Seller that are not used in the ownership or operation of the Assets, or that relate to the Excluded Liabilities; and

Section 2.1.

(n)

any rights specifically excluded from the definition of the Assets under

At any time or from time to time, up to ninety (90) days following the Closing, any and all of the Excluded Assets may be removed from the Facilities and the Facilities Switchyard by Seller (at no expense to Purchaser, but without charge by Purchaser for temporary storage), provided that Seller shall do so in a manner that does not unduly or unnecessarily disrupt normal business activities at the Facilities and the Facilities Switchyard, and provided further that Excluded Assets may be retained at the Facilities and the Facilities Switchyard to the extent permitted by easements, licenses, agreements or similar arrangements in favor of Seller.

2.3 <u>Assumption of Liabilities</u>. From and after the Closing, Purchaser will assume the following obligations and liabilities of Seller to the extent such obligations and liabilities relate to the Assets (the "Assumed Liabilities"):

(a) All liabilities or obligations (including, without limitation, any fines, penalties or costs imposed by a Governmental Authority) arising under Environmental Laws (whether such laws are enacted before or after the Closing Date), and all liabilities and obligations relating to Environmental Conditions or Hazardous Substances, in each case to the extent attributable to actions or failures to act occurring, or conditions first arising, after the Closing Date in connection with Purchaser's ownership of the Assets or the operation thereof or with respect to the Navajo Mine, including any threatened Releases that do not exist prior to the Closing Date;

(b) Except for the payment obligations pro-rated to Seller under <u>Section 3.6</u>, or as specifically contemplated under <u>Section 2.4</u>, all liabilities and obligations under all agreements, contracts, undertakings, and licenses assigned to Purchaser under this Agreement, including the Facilities Contracts, and the Transferable Permits in accordance with the terms thereof, except in each case to the extent such liabilities and obligations were incurred by Seller prior to the Closing Date;

(c) All liabilities and obligations of Seller with respect to decommissioning the Facilities and the Facilities Switchyard, including without limitation the dismantling and removal of the Facilities and the Facilities Switchyard and the restoration of their sites, as described in the Decommissioning Report (collectively, "Decommissioning"). Notwithstanding the foregoing, the Decommissioning liabilities and obligations assumed by Purchaser do not include Excess Decommissioning Costs which would otherwise constitute Retained Environmental Liabilities and do not include liabilities and costs identified in Section 2.4(j). "Excess Decommissioning Costs" mean, for any work included in the Decommissioning Report, the decommissioning costs related thereto arising from (i) changes in Environmental Laws after the Effective Date, or (ii) Remediation activities for Environmental Conditions not reflected in the cost estimate in the Decommissioning Report including, without limitation, any decommissioning activities related to Morgan Lake or for soil and subsurface Environmental Conditions, which, in either case, imposes additional costs on Purchaser in excess of the cost estimate, if any, for that work in the Decommissioning Report on an inflation adjusted basis;

(d) All costs of modifications to the Facilities or their operations or of supplemental environmental projects legally required to operate the Facilities after the Closing or agreed to by the Facilities Owners (other than the amount of any fines or penalties which would otherwise constitute Retained Environmental Liabilities that were avoided by the agreement to implement the supplemental environmental projects, which amount shall be (i) as specified in the related settlement, (ii) as agreed to by the Parties if such amount is not specified and (iii) as determined by the provisions of <u>Section 11.10</u> if the amount is not specified and the Parties are unable to agree), whether or not the liabilities or obligations related to such costs are alleged, claimed, enforced, settled or paid for after the Closing Date, including without limitation the costs of any selective catalytic reduction technology or modifications to the Facilities related to the storage or handling of coal ash or other coal combustion residuals required to operate the Facilities after the Closing except, with respect to the storage or handling of coal ash or other

coal combustion residuals, to the extent that such costs would have been required even if the Plant were to be shut down on or about July 6, 2016.;

(e) All other liabilities expressly allocated to Purchaser in the Agreement;

(f) Subject to <u>Section 3.2(e)</u> and <u>Section 3.3(b)</u>, all of Seller's share of any liabilities or obligations of the Operating Agent or its Affiliates with respect to pensions or other post-employment benefits attributable to Operating Agent's operation of the Facilities;

(g) All Seller's obligations, if any, under the Facilities Contracts, (i) with respect to any Capital Expenditures that Seller cannot fund under California law, and (ii) to fund selective catalytic reduction technology if legally required to be installed at the Facilities; and

(h) All liabilities and obligations of Seller with respect to post-Closing reclamation and all final reclamation of the Navajo Mine, and the site comprising the same or on which the Navajo Mine exists or has existed as detailed in the Reclamation Report (collectively, "**Reclamation**"). Notwithstanding the foregoing, the Reclamation liabilities and obligations assumed by Purchaser do not include Excess Reclamation Costs which would otherwise constitute Retained Environmental Liabilities. "**Excess Reclamation Costs**" means, for any work included in the Reclamation Report, the reclamation costs related thereto arising from (i) changes in Environmental Laws after the Effective Date, or (ii) Remediation activities for Environmental Conditions not reflected in the cost estimate in the Reclamation Report which, in either case, imposes additional costs on Purchaser in excess of the cost estimate, if any, for that work in the Reclamation Report (as supplemented by the Cost Update) on an inflation adjusted basis. "**Cost Update**" means the Marston (Final Report) FCPP Reclamation Cost Table 2010 previously delivered by the Operating Agent to the Facilities Owners on September 21, 2010.

For the avoidance of doubt, Purchaser is not assuming any liabilities or obligations of any of the Facilities Owners other than Seller pursuant to this Agreement.

2.4 <u>Excluded Liabilities</u>. Purchaser shall not assume or be obligated to pay, perform or otherwise discharge any liabilities or obligations of Seller other than the Assumed Liabilities. All obligations and liabilities of Seller other than the Assumed Liabilities are referred to herein as the "Excluded Liabilities", all of which Excluded Liabilities shall remain the sole responsibility of Seller. The Excluded Liabilities include, without limitation, the following:

(a) Any liabilities or obligations of Seller in respect of any Excluded Assets or other assets which are not Assets and the ownership, operation and conduct of any business in connection therewith or therefrom;

(b) Any liabilities or obligations of Seller in respect of costs under <u>Section 3.6</u> and Taxes attributable to the ownership, operation or use of Assets before the Closing Date (except for Taxes for which Purchaser is liable pursuant to <u>Section 3.6</u>) and any Taxes for which Seller is liable under <u>Section 6.5</u>;

(c) Except as otherwise specifically set forth in <u>Section 2.3</u> herein, liabilities or obligations arising prior to the Closing Date under any of the agreements or contracts assumed by Purchaser, including the Facilities Contracts;

(d) Liabilities or obligations under any of the Facilities Contracts which would be included in the Assets but for the provisions of <u>Section 3.7</u>, unless Purchaser is provided with the benefits thereunder as contemplated by <u>Section 3.7</u>;

(e) Except as otherwise set forth in <u>Section 2.4(i)</u>, any fines, penalties or costs, other than costs specified in <u>Section 2.3(d)</u>, imposed by a Governmental Authority with respect to the Assets resulting from (i) an investigation, proceeding, request for information or inspection before or by a Governmental Authority pending or, to Seller's Knowledge, threatened prior to Closing, but only relating to actions or omissions or conditions existing prior to the Closing Date or (ii) violations of applicable law or illegal acts of Seller;

(f) Any liability of Seller arising out of a breach by Seller of any of its obligations under this Agreement, the Confidentiality Agreement or the Ancillary Agreements;

(g) Any obligation of Seller to indemnify any Person who is a member of the Purchaser Group pursuant to <u>ARTICLE 7</u>;

(h) Any costs or expenses for which Seller is liable under this Agreement;

(i) Seller's share of all liabilities or obligations (including, without limitation, any fines, penalties or costs imposed by a Governmental Authority) arising under Environmental Laws (whether such laws are enacted before or after the Closing Date), and all liabilities or obligations relating to Environmental Conditions or Hazardous Substances, to the extent attributable to actions or failures to act occurring, or conditions first arising, prior to the Closing Date in connection with Seller's ownership of the Assets or the operation thereof or with respect to the Navajo Mine, whether or not such liabilities and obligations are alleged, claimed, enforced, settled, or paid for after the Closing Date (the "**Retained Environmental Liabilities**"), but excluding all liabilities assumed by Purchaser under <u>Section 2.3(c)</u>, <u>Section 2.3(d)</u> and <u>Section 2.3(h)</u>, and related to any threatened Releases that do not exist prior to the Closing Date;

(j) Seller's share of the costs of Remediation or removal of the Landfill if the Facilities Owners are required to Remediate or remove such Landfill under Laws, the Facilities Lease or the § 323 Grants.

## 2.5 Control of Litigation.

(a) The Parties acknowledge and agree that, from and after the Closing Date, as between Seller and Purchaser, Seller shall be entitled exclusively to control, defend and settle any suit, action, proceeding or investigation arising out of or related to any Excluded Assets, Excluded Liabilities or Tax and related audit, appeals process or litigation for taxable periods occurring prior to the Closing Date, in each case, not involving claims against the Operating Agent or the other Facilities Owners, and Purchaser agrees to cooperate reasonably in connection therewith, it being understood that Purchaser shall not be required to incur any cost in connection with any such settlement but may be required to provide a release to a third party claimant in respect of the specific matters involved in such suit, action, proceeding or investigation; provided, however, that Seller shall reimburse Purchaser for all reasonable costs and expenses incurred in providing such cooperation to Seller and shall not unreasonably interfere with operations at the Facilities or the Facilities Switchyard.

(b) The Parties acknowledge and agree that, from and after the Closing Date, as between Seller and Purchaser, Purchaser shall be entitled exclusively to control, defend and settle any suit, action, proceeding or investigation arising out of or related to any Assets or Assumed Liabilities, in each case, not involving Excluded Assets or Excluded Liabilities, and Seller agrees to cooperate reasonably in connection therewith, it being understood that Seller shall not be required to incur any cost in connection with any such settlement but may be required to provide a release to a third party claimant in respect of the specific matters involved in such suit, action, proceeding, or investigation; provided, however, that Purchaser shall reimburse Seller for all reasonable costs and expenses incurred in providing such cooperation to Purchaser and shall not unreasonably interfere with Seller's operations.

(c) For suits, actions, proceedings, or investigations arising out of or related to both Excluded Assets and/or Excluded Liabilities, and Assets and/or Assumed Liabilities:

(i) For suits, actions, proceedings or investigations which are reasonably expected to result in costs and liabilities to Seller of less than two hundred fifty thousand dollars (250,000) and in which Seller is not a named party, such matters shall be controlled and defended by the Operating Agent under the Facilities Operating Agreement, with Seller exercising the rights it retains under <u>Section 2.1(h)</u> through Purchaser and the costs thereof allocated between Seller and Purchaser in accordance with the allocation of Assumed Liabilities and Excluded Liabilities under this Agreement, provided that to the extent not already required by Seller's retention of rights it retains under <u>Section 2.1(h)</u>:

(1) Purchaser shall keep Seller informed of material developments related to such suits, actions, proceedings or investigations in a timely manner;

(2) Seller's approval shall be required for any compromise or settlement of any liability or obligation of Seller, such approval not to be unreasonably withheld; and

(3) Seller's approval shall be required for any admission of liability or guilt in any civil or criminal matter, with such approval at Seller's sole discretion.

(ii) For all other suits, actions, proceedings or investigations, the Parties agree to coordinate with each other with respect to the defense thereof. Without limiting the foregoing, for suits, actions, proceedings or investigations in which Seller and Purchaser are named parties, Seller and Purchaser shall discuss the feasibility of having one counsel represent Seller and Purchaser.

### 2.6 Carbon Emission Allowances.

(a) To the extent that legislation and/or regulations creating and allocating Carbon Emission Allowances are adopted after the Effective Date, the Parties agree that Seller shall receive the Carbon Emission Allowances to which it would have been entitled if (i) the Facilities were operated through July 6, 2016 and (ii) Seller retained its current interest in the Facilities through that date. Accordingly: (i) <u>Carbon Emission Allowances allocated by a Governmental</u> <u>Authority before July 6, 2016</u>: Seller will retain any Carbon Emission Allowances allocated to Seller, or attributable to a 48% share of the Units 4 and 5 at the Plant ("Seller's Facilities Share"), and Purchaser will surrender to Seller any such Carbon Emission Allowances attributable to Seller's Facilities Share, except that Seller will surrender to Purchaser or Purchaser will retain the Carbon Emission Allowances needed to cover the operation of Seller's Facilities Share from the Closing through July 6, 2016. After the Closing, Seller shall have no obligation to purchase any Carbon Emission Allowances if the amount of Carbon Emission Allowances allocated to Seller or attributable to Seller's Facilities Share is less than the amount needed to cover such operation;

(ii) <u>Carbon Emission Allowances allocated by a Governmental</u> <u>Authority after July 6, 2016 based on any pre-July 6, 2016 measurement period</u>: Seller will retain any Carbon Emission Allowances allocated to Seller, or attributable to Seller's Facilities Share, and Purchaser will surrender to Seller any such Carbon Emission Allowances attributable to Seller's Facilities Share; and

(iii) <u>Carbon Emission Allowances allocated by a Governmental</u> <u>Authority after July 6, 2016 based on a post-July 6, 2016 measurement period</u>: Seller will surrender to Purchaser or Purchaser will retain any such Carbon Emission Allowances.

(b) If, prior to July 6, 2016, Purchaser has the option of selecting a measurement period for Carbon Emission Allowances for the Facilities, then the Carbon Emission Allowances to be retained by or surrendered to Seller shall be calculated as if Purchaser had selected the measurement period which results in the highest amount of Carbon Emission Allowances retained by or surrendered to Seller.

2.7 California Capacity Rights. If the Closing Date occurs prior to October 1, 2012, Seller will have the month-by-month option to retain the capacity rights for the Seller's Facilities Share for purposes of satisfying the requirements of California's Resource Adequacy program for each month from the Closing Date until October 1, 2012. Seller may exercise such option by providing Purchaser with twenty days advanced prior written notice of the exercise of such option for each such month; provided that with respect to the month in which the Closing Date occurs, if Seller has not been able to provide twenty days advanced prior written notice, Seller shall be deemed not to have exercised such option. In no event shall Purchaser have any obligation to secure replacement capacity or any other remedy in the event the Facilities are not operating at full capacity. For any month after the Closing Date for which Seller retains the capacity rights for Seller's Facilities Share, Purchaser shall, and shall have the exclusive authority to, submit or cause to be submitted all bids, including but not limited to supply bids for energy, self-schedules and self-provision of ancillary services for Seller's Facilities Share in the California ISO day-ahead market, hour-ahead scheduling process and real-time market, as required for Seller to satisfy the requirements of California's Resource Adequacy program. All revenues produced from such bids and self-schedules from the day-ahead market, hour-ahead scheduling process, real-time market and other revenues (including but not limited to revenues from ancillary services, the integrated forward market and residual unit commitment) related to Seller's Facilities Share flowing from mechanisms other than the capacity market shall accrue to Purchaser. If the energy is called for, Purchaser will deliver such energy to the Moenkopi

Switchyard. Any wheeling costs for energy delivery incurred by Seller for delivery of energy from the Moenkopi Switchyard to the middle of the Colorado River will be netted from Purchaser's market energy revenues.

# ARTICLE 3 CLOSING

3.1 <u>Closing</u>. The closing of the sale of the Assets to, and the assumption of the Assumed Liabilities by, Purchaser (the "Closing") will take place at the offices of Arizona Public Service Company, 400 North Fifth Street, Phoenix, Arizona 85004, at 10:00 a.m. local time on the first day of the first full month following the date on which the conditions set forth in <u>ARTICLE 8</u> and <u>ARTICLE 9</u> have been either satisfied or waived by the Party for whose benefit such conditions precedent exist, or if such day is not a Business Day, on the next succeeding Business Day, or on such other date and at such other place as the Parties may mutually agree. The time and date of Closing is hereinafter called the "Closing Date." Notwithstanding anything in this Agreement to the contrary, the Closing shall be deemed to have taken place at 12:01 a.m., Fruitland, New Mexico prevailing time, on the Closing Date, or if the Closing Date is not the first day of the month because such day is not a Business Day, on the first day of the month.

**3.2** <u>Purchase Price</u>. At or, as applicable, after, the Closing, the Initial Purchase Price shall be adjusted, without duplication, to account for the following items and Closing Adjustments, and Post-Closing Adjustments, as set forth in <u>Section 3.3</u>, the sum of which is hereinafter referred to as the "**Purchase Price**":

(a) <u>**Prorations**</u>. The Initial Purchase Price shall be adjusted to account for the items prorated as of the Closing Date pursuant to <u>Section 3.6</u>.

Capital Expenditures. Subject to Section 6.6, the Initial Purchase Price (b) will be increased by an amount equal to (i) the aggregate Capital Expenditures funded by Seller during 2010 and 2011 in excess of its share of the aggregate 2010 and 2011 capital budgets for the Facilities approved by the Facilities Owners, plus (ii) Capital Expenditures funded by Seller during 2012 and thereafter, until the Closing Date, in each case minus the amount of any depreciation Seller incurred as a result of such Capital Expenditures through the Closing Date; provided, however, that any costs or expenditures for which Seller is responsible under Section 2.4(i), or which are made on an emergency basis to address actual or anticipated equipment failures that would adversely affect the operating capacity of the Facilities prior to Closing and are economically viable to Seller ("Emergency Capital Expenditures"), will not increase the Initial Purchase Price; provided further that, with respect to Emergency Capital Expenditures made in 2012 and thereafter, Seller has received appropriate regulatory approval of cost recovery for such Emergency Capital Expenditures. For purposes of the foregoing, an Emergency Capital Expenditure will be considered economically viable if the net benefits to Seller associated with making the repair up to the Closing Date exceed the net costs to Seller associated with making the repair. This provision is intended only to address potential adjustments to the Initial Purchase Price, and is not intended to modify the parties' rights and obligations under the Facilities Contracts.

(c) <u>Timing of Closing</u>. If the Closing Date is prior to October 1, 2012, the Initial Purchase Price shall be increased by Seven Million Five Hundred Thousand Dollars (\$7,500,000) for each month between the Closing Date and October 1, 2012 and if the Closing Date is after October 1, 2012, the Initial Purchase Price shall be decreased by Seven Million Five Hundred Thousand Dollars (\$7,500,000) for each month between October 1, 2012 and the Closing Date.

(d) <u>California Capacity Rights</u>. If Seller exercises its option under <u>Section 2.7</u>, the Initial Purchase Price shall be decreased by Three Million Dollars (\$3,000,000) per month for each month for which Seller exercises its option to retain such capacity rights.

(e) <u>Pension and OPEB Liabilities</u>. If Seller's Share of Underfunded/Overfunded Pension and OPEB Liabilities is greater than \$0, the Initial Purchase Price shall be decreased by such amount, otherwise the Initial Purchase Price shall be increased by the absolute value of such amount, in each case, determined as of the Closing Date pursuant to <u>Section 3.3(b)</u>.

#### 3.3 <u>Pre-Closing and Post-Closing Adjustments</u>.

(a) At least thirty (30) calendar days prior to the Closing Date, Purchaser, with the assistance and participation of, and in consultation with, Seller, shall prepare and deliver to Seller an estimated closing statement (the "Estimated Closing Statement") that shall set forth Purchaser's best estimate of all estimated adjustments to the Initial Purchase Price required by Section 3.2 (the "Estimated Adjustment"). Within ten (10) calendar days after the delivery of the Estimated Closing Statement by Purchaser to Seller, Seller may object in good faith to the Estimated Adjustment in writing. If Seller objects to the Estimated Adjustment within such ten (10) day period, the Parties shall attempt to resolve their differences by negotiation. If the Parties are unable to do so prior to the Closing Date (or if Seller does not object to the Estimated Adjustment), the Initial Purchase Price shall be adjusted (the "Closing Adjustment") at the Closing by the amount of the Estimated Adjustment not in dispute. The disputed portion shall be resolved in accordance with the provisions of Section 3.3(b).

(b) Within sixty (60) days after the Closing Date, Purchaser, with the assistance and participation of, and in consultation with, Seller shall prepare and deliver to Seller a final closing statement (the "**Post-Closing Statement**") that shall set forth all adjustments to the Initial Purchase Price proposed by Purchaser to be required by <u>Section 3.2(a)</u> through <u>3.2(e)</u> not previously effected by the Closing Adjustment (the "**Proposed Post-Closing Adjustment**"); provided that if any adjustments to be made pursuant to <u>Section 3.2(d)</u> cannot be made within sixty (60) days after the Closing Date, the Parties agree that additional Post-Closing Statements can be subsequently prepared to address such adjustments. To the extent applicable, the Post-Closing Statement shall be prepared using the same accounting principles, policies and methods as the Operating Agent has historically used in connection with the calculation of the items reflected on such Post-Closing Statement. Without limiting the generality of the foregoing, for matters covered by <u>Section 3.2(e)</u>, the discount rate and other assumptions used to determine Pension and OPEB Liabilities as reflected in the Post-Closing Statement shall be selected using the same methodology historically used for selecting the discount rate and assumptions for

Pinnacle West Capital Corporation's consolidated fiscal year-end calculations reported in its audited financial statements. Within thirty (30) days after the delivery of the Post-Closing Statement by Purchaser to Seller, Seller may object in good faith to the Proposed Post-Closing Adjustment in writing, stating in reasonable detail its objections thereto. Purchaser and Seller agree to cooperate to exchange information used to prepare the Post-Closing Statement and information relating thereto. If Seller objects to the Proposed Post-Closing Adjustment, the Parties shall attempt to resolve such dispute by negotiation. If the Parties are unable to resolve such dispute within thirty (30) days after any objection by Seller, the Parties shall appoint the Independent Accounting Firm, which shall, at Seller's and Purchaser's joint expense, review the Proposed Post-Closing Adjustment and determine the appropriate adjustment to the Purchase Price, if any, within thirty (30) days after such appointment. The Parties agree to cooperate with the Independent Accounting Firm and provide it with such information as it reasonably requests to enable it to make such determination. For purposes of this Section 3.3(b) and wherever the Independent Accounting Firm is retained to resolve a dispute between the Parties, the Independent Accounting Firm may determine the issues in dispute following such procedures. consistent with the language of this Agreement, as it deems appropriate to the circumstances and with reference to the amounts in issue. No particular procedures are intended to be imposed upon the Independent Accounting Firm, it being the desire of the Parties that any such disagreement shall be resolved as expeditiously and inexpensively as reasonably practicable. The Independent Accounting Firm shall have no liability to the Parties in connection with such services except for acts of bad faith, willful misconduct or gross negligence, and the Parties shall provide such indemnities to the Independent Accounting Firm as it may reasonably request. The finding of such Independent Accounting Firm shall be binding on the Parties hereto. Upon determination of the appropriate adjustment (the "Post-Closing Adjustment") by agreement of the Parties or by binding determination of the Independent Accounting Firm, the Party owing the difference shall deliver such amount to the other Party no later than two (2) Business Days after such determination, in immediately available funds or in any other manner as reasonably requested by the payee.

**3.4** <u>**Payment**</u>. Any cash payments required by this Agreement shall be paid in U.S. dollars in immediately available funds. The recipient of such funds will designate the account or accounts to which the funds will be wire transferred.

**3.5** <u>Allocation of Purchase Price</u>. The Parties will file all Tax Returns consistently with the allocation of the Purchase Price determined in accordance with this <u>Section 3.5</u>. The allocation of the Purchase Price (including any portion of the Assumed Liabilities if applicable) will be negotiated by the Parties in accordance with Applicable Tax Law (as defined below). Purchaser shall propose and deliver to Seller a preliminary allocation among the Assets of the Purchase Price and such other consideration to be paid to Seller pursuant to this Agreement (an "Allocation") sufficiently far in advance of the Closing to allow the Final Pre-Closing Allocation referred to below to be determined prior to the Closing. The Allocation shall be consistent with Code Section 1060 ("Applicable Tax Law")</u> and the regulations thereunder and in a manner which facilitates Property Tax reporting and shall separately allocate Assets in the Facilities Switchyard. Seller shall within thirty (30) days thereafter propose any changes to the Allocation. Within thirty (30) days following delivery of such proposed changes, together with a reasonably detailed explanation of the reasons therefor. If Purchaser and Seller are unable to

resolve any disputed objections within ten (10) days thereafter, such objections shall be referred to the Independent Accounting Firm, which shall determine the Allocation (including any valuations). The Independent Accounting Firm shall be instructed to deliver to Purchaser and Seller a written determination of the proper allocation of such disputed items within twenty (20) Business Days from the date of engagement. Such determination shall be final, conclusive and binding upon the Parties for all purposes, and the Allocation shall be so adjusted (the allocation, including the adjustment, if any, to be referred to as the "Final Pre-Closing Allocation"). Within thirty (30) days of the determination of the Post-Closing Adjustment, the Parties shall agree to the adjustments to the Final Pre-Closing Allocation ("Final Allocation"). The fees and disbursements of the Independent Accounting Firm attributable to any Allocation shall be shared equally by Purchaser and Seller. Purchaser and Seller agree to timely file Internal Revenue Service Form 8594, and all Tax Returns, in accordance with such Allocation or Final Allocation. as the case may be, and to report the transactions contemplated by this Agreement for Federal Income tax and all other tax purposes in a manner consistent with the Allocation or Final Allocation, as the case may be. Purchaser and Seller agree to promptly provide the other Parties with any additional information and reasonable assistance required to complete Form 8594, or compute Taxes arising in connection with (or otherwise affected by) the transactions contemplated hereunder.

#### 3.6 <u>Prorations.</u>

(a) Purchaser and Seller agree that, except as otherwise specifically provided in this Agreement, all of the budgeted, ordinary, and recurring items normally charged to the Facilities Owners, including those listed below (but not including any Income Taxes and Transfer Taxes), relating to the business and operation of the Assets, shall be prorated and charged as of the Closing Date, without any duplication of payment under the Facilities Contracts, with Seller liable to the extent such items relate to any time period prior to the Closing Date, and Purchaser liable to the extent such items relate to periods commencing with the Closing Date (measured in the same units used to compute the item in question, otherwise measured by calendar days):

(i) Property Taxes having a lien date in the same calendar year as the Closing Date, provided, however, with respect to any Property Taxes imposed by a Tribal Authority, such Property Taxes shall be prorated based upon that portion of the calendar year starting with the date of expiration of any applicable tax waiver and ending with the last day of the calendar year of the Closing Date;

(ii) Retrospective adjustments and policyholder distributions for the applicable period during which the Closing occurs with respect to Facilities Insurance Policies included in the Assets occurring within twelve (12) months of Closing or ninety (90) days after the year-end following the Closing, whichever occurs first; and

(iii) Operating and maintenance expenses incurred in any period prior to the Closing Date (not including Capital Expenditures) in the nature of the expenses shown on <u>Schedule 3.6(a)(iii)</u> but only to the extent that the amount of such expenses are determined within twelve (12) months of Closing or ninety (90) days after the year-end following the Closing, whichever occurs first.

(b) In connection with the prorations referred to in <u>Section 3.6(a)</u>, in the event that actual figures are not available at the Closing Date, the proration shall be based upon the respective amounts accrued through the Closing Date or paid for the most recent year or other, appropriate period for which such amounts paid are available. All prorated amounts shall be recalculated and paid to the appropriate Party within sixty (60) days after the date that the previously unavailable actual figures become available. Seller and Purchaser shall furnish each other with such documents and other records as may be reasonably requested in order to confirm all proration calculations made pursuant to this <u>Section 3.6</u>.

3.7 No Assignment if Breach. To the extent that Seller's rights under any of the Facilities Contracts to be transferred to Purchaser hereunder may not be assigned without the consent of another Person which consent has not been obtained, this Agreement shall not constitute an agreement to assign the same if an attempted assignment would constitute a breach thereof or be unlawful, and Purchaser and Seller shall cooperate and each use Commercially Reasonable Efforts to obtain any such required consent(s) as promptly as possible. Seller and Purchaser agree that if any consent to an assignment of any of the Facilities Contracts to be transferred hereunder shall not be obtained or if any attempted assignment would be ineffective or would impair Purchaser's rights and obligations under the applicable Facilities Contracts so that Purchaser would not in effect acquire all such rights and obligations. Seller, to the maximum extent permitted by law and such Facilities Contracts, shall after the Closing appoint Purchaser to be Seller's representative and agent with respect to such Facilities Contracts, and Seller shall, to the maximum extent permitted by law and such Facilities Contracts, enter into such reasonable arrangements with Purchaser as are necessary to transfer to Purchaser the benefits and obligations of such Facilities Contracts. Seller and Purchaser shall cooperate and shall each use Commercially Reasonable Efforts after the Closing to obtain an assignment of such Facilities Contracts to Purchaser.

**3.8** <u>Deliveries by Seller</u>. Subject to the terms and conditions hereof, at the Closing Seller shall deliver, or cause to be delivered, the following to Purchaser:

(a) The Deed, duly executed by Seller and in recordable form, subject only to Permitted Encumbrances and any owner's affidavits or similar documents reasonably required by Title Insurer;

(b) The Bill of Sale, duly executed by Seller;

(c) The Assignment and Assumption Agreement, duly executed by Seller;

(d) Evidence, in form and substance reasonably satisfactory to Purchaser and its respective counsel, of Seller's receipt of (i) Seller's Required Regulatory Approvals, (ii) Seller's Required Consents, and (iii) documentation evidencing the release of all Encumbrances, except for Permitted Encumbrances, including the release of Seller's Mortgage;

(e) A Certificate of Good Standing with respect to Seller, as of a recent date, issued by the Secretary of State of the State of California and of the state where the Facilities are located;

(f) To the extent available, originals of all of the Facilities Contracts to which Seller has Knowledge that it is a party (other than Facilities Contracts referenced in the proviso to <u>Section 2.1(h)</u>), the Transferable Permits issued to Seller and of which it has Knowledge and, if not available, true and correct copies thereof;

(g) A certificate addressed to Purchaser dated the Closing Date executed by a duly authorized officer of Seller to the effect set forth in <u>Section 8.6;</u>

(h) A FIRPTA Affidavit to Purchaser, duly executed by Seller;

(i) Copies, certified by the Secretary or Assistant Secretary of Seller, of corporate resolutions authorizing the execution and delivery of this Agreement, each Ancillary, Agreement to which Seller is a party and the authorization or ratification of all of the other agreements and instruments, in each case, to be executed and delivered by Seller in connection herewith;

(j) A certificate of the Secretary or Assistant Secretary of Seller identifying the name and title and bearing the signatures of the officers of Seller authorized to execute and deliver this Agreement, each Ancillary Agreement to which Seller, is a party and the other agreements and instruments contemplated hereby; and

(k) All such other agreements, documents, instruments and writings required to be delivered by Seller at or prior to the Closing Date pursuant to this Agreement necessary to sell, assign, convey, transfer and deliver all of Seller's rights, title and interests in and to the Assets, to Purchaser, in accordance with this Agreement and, where necessary or desirable, in recordable form.

**3.9** <u>Deliveries by Purchaser</u>. Subject to the terms and conditions hereof, at the Closing, Purchaser shall deliver, or cause to be delivered, the following to Seller:

(a) The Purchase Price, by wire transfer of immediately available funds to the account of Seller designated by Seller in writing on or before the Closing Date;

(b) The Assignment and Assumption Agreement, duly executed by Purchaser;

(c) Evidence, in form and substance reasonably satisfactory to Seller and its respective counsel, of Purchaser's receipt of (i) Purchaser's Required Regulatory Approvals, and (ii) Purchaser's Required Consents;

(d) A Certificate of Good Standing with respect to Purchaser, as of a recent date, issued by the ACC and the state in which the Facilities are located;

(e) A certificate dated the Closing Date executed by a duly authorized officer of Purchaser to the effect set forth in <u>Section 9.6;</u>

(f) Copies, certified by the Secretary or Associate Secretary of Purchaser, of resolutions authorizing the execution and delivery of this Agreement, each Ancillary Agreement

to which Purchaser is a party and the authorization or ratification of all of the agreements and instruments, in each case, to be executed and delivered by Purchaser in connection herewith;

(g) A certificate of the Secretary or Associate Secretary of Purchaser identifying the name and title and bearing the signatures of the officers of Purchaser authorized to execute and deliver this Agreement, each Ancillary Agreement to which Purchaser is a party and the other agreements contemplated hereby; and

(h) All such other agreements, documents, instruments and writings required to be delivered by Purchaser at or prior to the Closing Date pursuant to this Agreement.

**3.10** <u>Facilities Contracts</u>. The Parties agree that between the date hereof and the Closing Date, the ownership, lease, maintenance and operation of the Facilities and the Facilities Switchyard will be governed by the Facilities Contracts.

## ARTICLE 4 REPRESENTATIONS, WARRANTIES AND DISCLAIMERS OF SELLER

Except as set forth in Seller's Schedule of Exceptions corresponding to the Section of this Agreement to which such disclosure applies, Seller represents, warrants and, where specified, disclaims to Purchaser as follows:

4.1 Organization and Existence. Seller is a corporation, duly organized, validly existing and in good standing under the laws of the State of California and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as is now being conducted. Seller is duly qualified to do business and is in good standing in the state where the Facilities are located. Seller has heretofore delivered to Purchaser complete and correct copies of its Articles of Incorporation and Bylaws as currently in effect.

4.2 **Execution.** Delivery and Enforceability. Seller has full corporate power to enter into, and carry out its obligations under, this Agreement and the Ancillary Agreements which are executed by Seller and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance of this Agreement and the Ancillary Agreements which are executed by Seller, and the consummation of the transactions contemplated hereby and thereby, have been duly and validly authorized by all necessary corporate action required on the part of Seller and no other corporate proceedings on the part of Seller are necessary to authorize this Agreement and the Ancillary Agreements to which it is a party or to consummate the transactions contemplated hereby and thereby. Assuming Purchaser's due authorization, execution and delivery of this Agreement and the Ancillary Agreements when executed by Purchaser, this Agreement does and the Ancillary Agreements when executed by Seller will constitute the valid and legally binding obligations of Seller, enforceable against Seller in accordance with its and their terms, except as such enforceability may be limited by bankruptcy. insolvency, reorganization, moratorium or other similar laws of general application relating to or affecting the enforcement of creditors' rights and by general equitable principles.

**4.3** <u>No Violation</u>. Subject to Seller obtaining Seller's Required Regulatory Approvals and Seller's Required Consents, and except for compliance with the requirements of the HSR Act, neither the execution and delivery of this Agreement or any of the Ancillary

Agreements executed by Seller, nor the compliance with any provision hereof or thereof, nor the consummation of the transactions contemplated hereby or thereby will:

(a) violate, or conflict with, or result in a breach of any provisions of the Articles of Incorporation or Bylaws of Seller;

(b) result in a default (or give rise to any right of termination, cancellation or acceleration) under or conflict with any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license, or agreement or other instrument or obligation to which Seller is a party or by which Seller or any of the Assets may be bound, except for such defaults (or rights of termination or acceleration) as to which requisite waivers or consents have been, or prior to the Closing will have been, obtained or which would not, individually or in the aggregate, create a Material Adverse Effect;

(c) violate any law, rule, regulation, order, writ, injunction, or decree, applicable to Seller or any of its assets, except where such violations, individually or in the aggregate, would not create a Material Adverse Effect and will not affect the validity or enforceability of this Agreement or the Ancillary Agreements or the validity of the transactions contemplated hereby or thereby; or

(d) require consent or approval of, filing with, or notice to any Person which, if not obtained would prevent Seller from performing its obligations hereunder.

4.4 <u>Compliance with Laws</u>. Seller has no Knowledge that it is in material violation of any laws, orders, ordinances, rules, regulations or judgment of any Governmental Authority in existence as of execution of this Agreement with respect to the Assets, except for (a) violations or alleged violations the subject matter of which Purchaser or the Operating Agent has Knowledge, (b) violations or alleged violations by the Facilities Owners in common, or by the Operating Agent acting on their behalf, or (c) violations or alleged violations that will not have a Material Adverse Effect.

4.5 <u>Permits, Licenses, Etc.</u> Prior to the Closing Date, Seller will hold all permits, registrations, franchises, certificates, licenses and other authorizations, consents and approvals of all Governmental Authorities that Seller requires in order to own any of the Assets (collectively, "Seller Permits"), except for such failures to hold such Seller Permits as to which Purchaser or the Operating Agent has Knowledge, are also failures of all of the other Facilities Owners (or all other than the Operating Agent) or would not, individually or in the aggregate, have a Material Adverse Effect.

4.6 <u>Litigation</u>. There is no claim, action, proceeding or investigation pending, or to Seller's Knowledge, threatened against or relating to Seller or its Affiliates before any court, arbitrator or Governmental Authority, or any judgment, decree or order of any court, arbitrator or Governmental Authority, which could, individually or in the aggregate, reasonably be expected to result, or has resulted, in (a) the institution of legal proceedings to prohibit or restrain the performance of this Agreement or any of the Ancillary Agreements, or the consummation of the transactions contemplated hereby or thereby, (b) a claim against Purchaser or its Affiliates for damages as a result of Seller entering into this Agreement or any of the Ancillary Agreements, or the consummation by Seller of the transactions contemplated hereby or thereby, (c) a material impairment of Seller's ability to perform its obligations under this Agreement or any of the Ancillary Agreements, or (d) a Material Adverse Effect, except for claims, actions, proceedings or investigations pending against, or judgments, decrees or orders involving all of the other Facilities Owners or the Operating Agent as agent for the Facilities Owners, or as to which Purchaser has Knowledge.

4.7 <u>Title</u>. Subject to the right of first refusal contained in the Facilities Co-Tenancy Agreement, Seller has good and marketable title, or valid and effective leasehold rights in the case of leased property, and valid and effective licenses in the case of licensed rights, to the tangible personal property included in the Assets to be sold, conveyed, assigned, transferred and delivered to Purchaser by Seller, free and clear of all liens, charges, claims, pledges, security interests, equities and encumbrances of any nature whatsoever, except for (a) those created by Purchaser, (b) those which will be discharged or released prior to or substantially simultaneously with, the Closing, (c) Permitted Encumbrances, (d) those which do not apply only and exclusively to the interest of Seller but that also apply to interests of the other Facilities Owners in common and/or the Operating Agent, as agent for any of the Facilities Owners, and (e) possible minor matters that do not materially interfere with the intended use of the Assets.

4.8 <u>Facilities Contracts</u>. Seller has no Knowledge of any claim, action, proceeding or investigation, pending or threatened, challenging the enforceability against Seller of the Facilities Contracts, except for challenges to the enforceability of such contracts against the Facilities Owners in common, challenges of which Purchaser or the Operating Agent has Knowledge, or challenges which are not likely to result in a Material Adverse Effect.

**4.9** Intellectual Property. Seller does not own or otherwise have any right to use any patent, trade name, trademark, service mark or other intellectual property that is used in and necessary for the operation of the Facilities or the Facilities Switchyard, other than such as may be included in the Assets or is licensed to the Facilities Owners or the Operating Agent, acting on their behalf.

**4.10** <u>Taxes</u>. At least sixty (60) Business Days before the Closing, Seller will advise Purchaser in writing of any taxing jurisdictions in which Seller owns assets or conducts business that require a notification to a taxing authority of the transactions contemplated by this Agreement, if the failure to make such notification, or obtain Tax clearances in connection therewith, would either require Purchaser to withhold any portion of the Purchase Price or would subject Purchaser to any liability for any Taxes of Seller.

4.11 <u>Undisclosed Liabilities</u>. Except for liabilities and obligations specifically referred to in <u>Section 2.3</u> or <u>Section 2.4</u>, the Assets are not, to the Knowledge of Seller, subject to any liability or obligation that has arisen solely as a result of an act or omission by Seller, except for Permitted Encumbrances, acts or omissions of which Purchaser or the Operating Agent has Knowledge, or liabilities and obligations that are not reasonably likely to have a Material Adverse Effect.

4.12 <u>Brokers</u>. All negotiations relating to this Agreement and the transactions contemplated hereby have been carried on by Seller and in such a manner as not to give rise to

any valid claim against Purchaser (by reason of Seller's actions) for a brokerage commission, finder's fee or other like payment to any Person.

# ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as set forth in Purchaser's Schedule of Exceptions corresponding to the Section of this Agreement to which such disclosure applies Purchaser represents, warrants and, where specified, disclaims to Seller as follows:

**5.1** <u>Organization and Existence</u>. Purchaser is a corporation, duly organized, validly existing and in good standing under the laws of the State of Arizona and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as is now being conducted. Purchaser has heretofore delivered to Seller complete and correct copies of its Articles of Incorporation and Bylaws as currently in effect.

5.2 **Execution, Delivery and Enforceability**. Purchaser has full corporate power to enter into, and carry out its obligations under, this Agreement and the Ancillary Agreements which are executed by Purchaser and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance of this Agreement and the Ancillary Agreements which are executed by Purchaser, and the consummation of the transactions contemplated hereby and thereby, have been duly and validly authorized by all necessary corporate action required on the part of Purchaser and no other corporate proceedings on the part of Purchaser are necessary to authorize this Agreement and the Ancillary Agreements to which it is a party or to consummate the transactions contemplated hereby and thereby. Assuming Seller's due authorization, execution and delivery of this Agreement and the Ancillary Agreements when executed by Seller, this Agreement does and the Ancillary Agreements when executed by Purchaser, will constitute the valid and legally binding obligations of Purchaser, enforceable against Purchaser in accordance with its and their terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws of general application relating to or affecting the enforcement of creditors' rights and by general equitable principles.

5.3 <u>No Violation</u>. Subject to Purchaser obtaining the Purchaser's Required Regulatory Approvals and the Purchaser's Required Consents, and except for compliance with the requirements of the HSR Act, neither the execution and delivery of this Agreement or any of the Ancillary Agreements executed by Purchaser, nor the compliance with any provision hereof or thereof, nor the consummation of the transactions contemplated hereby or thereby will:

(a) violate, or conflict with, or result in a breach of any provisions of the Articles of Incorporation or Bylaws of Purchaser;

(b) result in a default (or give rise to any right of termination, cancellation or acceleration) under or conflict with any of the terms, conditions or provisions of any note, bond, mortgage, indenture, license, or agreement or other instrument or obligation to which Purchaser is a party or by which Purchaser may be bound, except for such defaults (or rights of termination or acceleration) as to which requisite waivers or consents have been, or prior to the Closing will have been, obtained or which would not, individually or in the aggregate, create a Material Adverse Effect;

(c) violate any law, rule, regulation, order, writ, injunction, or decree, applicable to Purchaser or any of its assets, except where such violations, individually or in the aggregate; would not create a Material Adverse Effect and will not affect the validity or enforceability of this Agreement or the Ancillary Agreements or the validity of the transactions contemplated hereby or thereby; or

(d) require consent or approval of, filing with, or notice to any Person which, if not obtained would prevent Purchaser from performing its obligations hereunder.

**5.4** <u>Compliance with Laws</u>. Except as otherwise disclosed in writing by Purchaser to Seller contemporaneously with the execution of this Agreement, Purchaser has no Knowledge that it is in material violation of any applicable laws, orders, ordinances, rules, regulations or judgment of any Governmental Authority in existence as of the Effective Date with respect to the Assets, except for violations or alleged violations that are reasonably expected not to have a Material Adverse Effect.

5.5 Litigation. There is no claim, action, proceeding or investigation pending, or to Purchaser's Knowledge, threatened against or relating to Purchaser or its Affiliates before any court, arbitrator or Governmental Authority, or any judgment, decree or order of any court, arbitrator or Governmental Authority, which could, individually or in the aggregate, reasonably be expected to result, or has resulted, in (a) the institution of legal proceedings to prohibit or restrain the performance of this Agreement or any of the Ancillary Agreements, or the consummation of the transactions contemplated hereby or thereby, (b) a claim against Seller or its Affiliates for damages as a result of Purchaser entering into this Agreement or any of the Ancillary Agreements, or the consummation by Purchaser of the transactions contemplated hereby or thereby, (c) a material impairment of Purchaser's ability to perform its obligations under this Agreement or any of the Ancillary Agreements, or (d) a Material Adverse Effect.

5.6 <u>Brokers</u>. All negotiations relating to this Agreement and the transactions contemplated hereby have been carried on by Purchaser and in such a manner as not to give rise to any valid claim against Seller (by reason of Purchaser's actions) for a brokerage commission, finder's fee or other like payment to any Person.

5.7 <u>Financing</u>. Purchaser has now, and at the Closing Purchaser will have, liquid capital or committed sources therefor sufficient to permit Purchaser to perform timely its obligations hereunder and under the Ancillary Agreements.

**5.8 Qualified for Permits.** To Purchaser's Knowledge, Purchaser is, or will be prior to the Closing, qualified to obtain any Facilities Permits necessary for the ownership and operation by Purchaser of the Assets as of the Closing in substantially the same manner as the Assets are currently operated.

5.9 <u>"AS IS" SALE</u>. EXCEPT AS OTHERWISE EXPRESSLY PROVIDED HEREIN, PURCHASER UNDERSTANDS AND AGREES THAT THE ASSETS ARE BEING ACQUIRED "AS IS, WHERE IS" ON THE CLOSING DATE, AND IN THEIR

CONDITION ON THE CLOSING DATE, AND THAT PURCHASER IS RELYING ON ITS OWN EXAMINATION OF THE ASSETS. WITHOUT LIMITING THE **GENERALITY OF THE FOREGOING AND EXCEPT FOR THE REPRESENTATIONS** AND WARRANTIES EXPRESSLY SET FORTH IN THIS AGREEMENT, PURCHASER UNDERSTANDS AND AGREES THAT SELLER EXPRESSLY DISCLAIMS ANY **REPRESENTATIONS OR WARRANTIES AS TO LIABILITIES, OPERATIONS OF** THE ASSETS, TITLE, CONDITION, VALUE OR OUALITY OF THE ASSETS OR THE **PROSPECTS (FINANCIAL OR OTHERWISE), RISKS AND OTHER INCIDENTS OF** THE ASSETS AND ANY REPRESENTATION OR WARRANTY OF **MERCHANTABILITY**, USAGE, SUITABILITY OR FITNESS ANY FOR PARTICULAR PURPOSE WITH RESPECT TO THE ASSETS OR ANY PART THEREOF, OR AS TO THE WORKMANSHIP THEREOF, OR THE ABSENCE OF ANY DEFECTS THEREIN, WHETHER LATENT OR PATENT. PURCHASER FURTHER AGREES THAT NO INFORMATION OR MATERIAL PROVIDED BY OR **COMMUNICATION MADE BY SELLER OR ANY REPRESENTATIVE OF SELLER** WILL CAUSE OR CREATE ANY REPRESENTATION OR WARRANTY DISCLAIMED BY THE FOREGOING EXCEPT AS DISCLOSED IN THIS AGREEMENT OR IN A SCHEDULE ATTACHED HERETO.

## ARTICLE 6 COVENANTS OF EACH PARTY

## 6.1 <u>Efforts to Close.</u>

(a) <u>Commercially Reasonable Efforts</u>. Subject to the terms and conditions herein provided, each of the Parties hereto agrees to use its Commercially Reasonable Efforts to consummate and make effective, as soon as reasonably practicable, the transactions contemplated hereby, including the satisfaction of all conditions thereto set forth herein. Such actions shall include, without limitation, exerting their Commercially Reasonable Efforts to (i) obtain the consents, authorizations and approvals of all private parties and any Governmental Authority whose consent is reasonably necessary to effectuate the transactions contemplated hereby, (ii) effect all other necessary registrations and filings, including, without limitation, filings under applicable laws, including the HSR Act and all other necessary filings with the CPUC, ACC, FERC (including applications to transfer the Facilities Switchyard), and any other Governmental Authority, and in the case of Purchaser, negotiate the extension or renewal referenced in Section 8.14 and the amendment referenced in Section 8.15. Each Party will provide the other with copies of all written communications from Governmental Authorities relating to the approval or disapproval of the transactions contemplated by the Agreement and the Ancillary Agreements.

(b) **Expenses**. Whether or not the transactions contemplated hereby are consummated, except as otherwise provided in this Agreement, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the Party incurring such expenses. Notwithstanding the foregoing:

(i) Costs associated with preliminary title reports and title policies shall be borne by Seller up to the costs that would have been incurred had the title policies been

standard coverage policies of title insurance, and the remaining costs, if any, including costs for extended coverage and any endorsements shall be borne by Purchaser (except that any survey costs shall be borne one-half by Purchaser and one-half by Seller);

(ii) Documentary transfer fees, if any, will be borne by Seller and recording costs and charges respecting real property will be borne one-half by Purchaser and one-half by Seller; and

(iii) Except as otherwise specifically set forth in <u>Section 6.4</u>, all fees, charges and costs of economists and other experts, if any, jointly retained by Purchaser and Seller in connection with submissions made to any Governmental Authority and advice in connection therewith respecting approval of the transactions will be borne one-half by Purchaser and one-half by Seller.

All such charges and expenses shall be promptly settled between the Parties at the Closing or upon termination or expiration of further proceedings under this Agreement, or with respect to such charges and expenses not determined as of such time, as soon thereafter as is reasonably practicable.

## (c) <u>Environmental Investigations</u>.

(i) Prior to Closing, Seller and/or Purchaser may, at their own cost and expense, conduct or cause to be conducted their own Phase 1 and Phase 2 environmental site assessments, and any follow up investigation, of the Facilities and the Facilities Switchyard as Seller and/or Purchaser deem necessary. The party conducting such assessments shall provide the other party with (1) a copy of any written reports resulting from such assessments; and (2) timely notice of any Environmental Condition(s) that require public disclosure or reporting to a regulatory authority or Remediation. Purchaser shall cooperate with and allow Seller to conduct such assessments and investigation. The results of such assessments and investigation shall not be binding on the Parties, and shall not be deemed to constitute an agreement by the Parties as to the existence or extent of current Environmental Conditions at the Facilities.

(ii) Following the Closing, Purchaser and Seller shall each appoint a representative to serve as an environmental liaison. Purchaser's environmental liaison shall provide Seller's environmental liaison with access to all information of Purchaser related to Environmental Conditions at the Facilities, consistent with Purchaser's normal record creation and retention policies and subject to a reasonable and appropriate non-disclosure agreement. The liaisons will meet periodically to address questions of Seller and to discuss generally Environmental Conditions at the Facilities that would affect the Retained Environmental Liabilities, including the status of ongoing Remediation programs.

(iii) Without limiting the generality of the foregoing, Purchaser's environmental liaison will provide Seller's environmental liaison with notice of and documentation relating to the initiation of any legal action and any threatened legal action of which Purchaser becomes aware, in each case that could reasonably be expected to affect Retained Environmental Liabilities. Purchaser also will provide Seller's liaison with a reasonable opportunity to review and comment on any new material Remediation initiative and
on environmental expenditures in the Facilities' annual budgets that could reasonably be expected to affect Retained Environmental Liabilities. If requested by Seller's liaison, Purchaser will present Seller's views on such environmental expenditures to the Engineering and Operating Committee (as defined in the Facilities Operating Agreement), and will allow Seller to participate in such presentation.

(iv) If Seller's liaison has significant concerns that an environmental matter will adversely affect Retained Environmental Liabilities, and the liaisons cannot satisfactorily address the concerns by themselves, the liaisons shall elevate the concerns to senior officers to be designated by each party prior to the Closing for further good faith discussions.

(v) The rights granted to Seller under this Section 6.1(c) shall not in any way alter or limit any rights retained by Seller under Section 2.1(h). In the event Purchaser transfers all or substantially all of the Assets to another Person and Purchaser is no longer the Operating Agent or a Facilities Owner, Purchaser shall cause such Person to assume Purchaser's obligations to Seller under Sections 6.1(c)(ii) through (v), or make such other arrangements as are reasonably acceptable to Seller.

6.2 Updating. Seller shall promptly notify Purchaser of any changes or additions to any of Seller's Schedules to this Agreement with respect to the Assets or Assumed Liabilities related thereto by the delivery of updates thereof, if any, as of a reasonably current date prior to the Closing, but in any event not later than three (3) Business Days prior thereto. No such updates made pursuant to this Section shall be deemed to cure any inaccuracy of any representation or warranty made in this Agreement as of the date hereof, unless Purchaser specifically agrees thereto in writing, nor shall any such notification be considered to constitute or give rise to a waiver by Purchaser of any condition set forth in this Agreement. Without limiting the generality of the foregoing, Seller shall notify Purchaser promptly of the occurrence of (i) any material casualty, physical damage, destruction or physical loss respecting, or any material adverse change in the physical condition of the Facilities or the Facilities Switchvard. subject to ordinary wear and tear and to routine maintenance, reasonably likely to result in a Material Adverse Effect, and (ii) any other material event likely to impair Seller's ability to perform, if, in the cases of clauses (i) and (ii), the occurrence is one of which Seller has Knowledge and of which the Operating Agent does not have Knowledge.

6.3 <u>Conduct Pending Closing</u>. Prior to consummation of the transactions contemplated hereby or the termination or expiration of this Agreement pursuant to its terms, and except to the extent approved by Purchaser, Seller shall:

(a) Except as required by their terms, or except to the extent agreed to by all Facilities Owners (including Seller and Purchaser), not amend, terminate, renew, or renegotiate any existing Facilities Contract or enter into any new Facilities Contract, except in the ordinary course of business and consistent with practices of the recent past, or default (or take or omit to take any action that with or without the giving of notice or passage of time, would constitute a default) under any of its obligations under any Facilities Contract;

(b) Not: (i) sell, lease, transfer or dispose of, or make any contract for the sale lease, transfer or disposition of, any assets or properties which would be included in the Assets,

other than sales in the ordinary course of business which would not, individually or in the aggregate; have a Material Adverse Effect upon the operations or value of the Facilities or the Facilities Switchyard; (ii) incur, assume, guaranty, or otherwise become liable in respect of any indebtedness for money borrowed, in each case which would result in Purchaser assuming such liability hereunder after the Closing; (iii) delay the payment and discharge of any liability which, upon Closing, would be an Assumed Liability, because of the transactions contemplated hereby; (iv) encumber or voluntarily subject to any lien any Asset, except for Permitted Encumbrances or (v) except to the extent approved by the other Facilities Owners, not settle any claims against the Facilities or against Seller relating to the Facilities or the Facilities Contracts; and

(c) Not take any action which would cause any of Seller's representations and warranties set forth in <u>ARTICLE 4</u> to be materially false as of the Closing;

<u>provided</u>, that nothing in this <u>Section 6.3</u> shall (i) preclude Seller from paying, prepaying or otherwise satisfying any liability which, if outstanding as of the Closing Date, would be an Assumed Liability or an Excluded Liability, or (ii) preclude Seller from incurring any liabilities or obligations to any third party in connection with obtaining such Party's consent to any transaction contemplated by this Agreement or the Ancillary Agreements; provided that any such liabilities or obligations incurred pursuant to clause (ii) shall be Excluded Liabilities.

(d) Seller agrees to advise Purchaser of any request that Seller intends to file with the CPUC for approval of Capital Expenditures by Seller budgeted for 2012 and of any action taken by the CPUC with respect thereto. If the CPUC denies such request or CPUC approval is not received by November 30, 2011, the Parties will meet within 15 days thereafter to discuss the consequences of the CPUC denial or failure to act, including its potential impact on the 2012 capital budget for the Plant, the respective obligations of the Parties under the Facilities Contracts and the operation of the Plant in 2012; it being understood that there is no obligation of either Party to reach any agreement with respect to any of the matters discussed.

#### 6.4 Consents and Approvals.

(a) Subject to Section 6.1(a), as promptly as practicable after the date of this Agreement, Seller and Purchaser shall each file or cause to be filed with the Federal Trade Commission and the Department of Justice all notifications required to be filed under the HSR Act and the rules and regulations promulgated thereunder with respect to the transactions contemplated hereby. The Parties shall consult with each other as to the appropriate time of filing such notifications and shall agree upon the timing of such filings, respond promptly to any requests for additional information made by either of such agencies, and cause the waiting periods under the HSR Act to terminate or expire at the earliest possible date after the date of filing. Purchaser and Seller shall be equally responsible for the cost of all filing fees under the HSR Act and each Party will bear its own costs for the preparation of any such filing.

(b) Subject to <u>Section 6.1(a)</u>, as promptly as practicable after the date of this Agreement, Purchaser shall file with FERC any other applications required under the Federal Power Act for the purchase and sale contemplated hereby, which filing(s) may be made individually by Purchaser or jointly with Seller, as reasonably determined by the Parties.

(c) Subject to Section 6.1(a), Seller shall be responsible for obtaining CPUC approval of the transactions contemplated by this Agreement and the Ancillary Agreements, and Purchaser shall be responsible for obtaining ACC approval of the transactions contemplated by this Agreement and the Ancillary Agreements, in each case including the filing of the necessary applications therefor and diligently prosecuting any resulting proceedings. Each Party shall afford the other Party the opportunity to review such filings. Unless requested by the filing Party, the other Party agrees not to intervene in the regulatory proceedings related to the approval of the transaction.

(d) Subject to <u>Section 6.1(a)</u>, Purchaser shall have the primary responsibility for securing the transfer, reissuance or procurement of the Facilities Permits, effective as of the Closing Date. Seller shall use Commercially Reasonable Efforts to cooperate with Purchaser's efforts in this regard and assist in any transfer or reissuance of Facilities Permits held by Seller or the procurement of any other Facilities Permits when so requested by Purchaser.

(e) Within fifteen (15) days after the receipt of any Purchaser's or Seller's Required Regulatory Approval, the Party receiving such approval (the "Receiving Party") shall notify the other Party in writing if the approval contains any condition that the Receiving Party determines could reasonably be expected to have a Material Adverse Effect on the Receiving Party or, in the case of Purchaser, on the Assets; provided, however, that if the Receiving Party does not provide such notice to the other Party within the fifteen (15) day period specified in this sentence, the Receiving Party shall be deemed to have accepted such Required Regulatory Approval, including any condition contained therein, and the condition to Closing set forth in Section 8.4 or Section 9.4, as applicable to such Party with respect to such Required Regulatory Approval, shall be deemed satisfied. Within fifteen (15) days after receipt of any notice specified in the previous sentence, Seller and Purchaser shall meet to consider what Commercially Reasonable Efforts the Receiving Party intends to take in order to obtain the Required Regulatory Approval or to eliminate the materially adverse conditions. After the Receiving Party has completed such agreed upon Commercially Reasonable Efforts with respect to the materially adverse condition contained in such Required Regulatory Approval, within fifteen (15) days of such completion or as soon as practicable thereafter, the Receiving Party shall notify the other Party, if the materially adverse condition has been eliminated or remains in effect, and whether the Receiving Party either will accept such materially adverse condition by a waiver of the applicable Closing condition in Section 8.4 or Section 9.4 with respect to such materially adverse condition or deem that the applicable Closing condition in Section 8.4 or Section 9.4 cannot be satisfied due to the materially adverse condition in such Required Regulatory Approval.

(f) From the date hereof through Closing, the Parties shall consult with each other at the senior management executive level prior to any party intervening in any regulatory proceeding of another Party, or commencing legal action or pursuing contractual remedies against any other Party with respect to the Facilities.

## 6.5 <u>Tax Matters.</u>

(a) All Transfer Taxes incurred in connection with this Agreement and the transactions contemplated hereby shall be borne one-half by Seller and one-half by Purchaser.

Seller will file, to the extent required by applicable law, all necessary Tax Returns and other documentation with respect to all such Transfer Taxes, and Purchaser will be entitled to review such returns in advance and, if required by applicable law, will join in the execution of any such Tax Returns or other documentation. Not less than five (5) Business Days prior to the due date of such Tax Returns, Purchaser shall pay Seller one-half of the amount shown as due on such Tax Returns, as determined in accordance, with this Agreement, and shall, to the extent required by Law, join in the execution of any such Tax Return. Prior to the Closing Date, Purchaser will provide to Seller, to the extent possible, an appropriate exemption certificate in connection with this Agreement and the transactions contemplated hereby, with respect to each applicable taxing authority.

(b) With respect to taxes to be prorated in accordance with Section 3.6 of this Agreement (except for pro-rated Property Taxes required to be paid by Seller), Purchaser shall prepare and timely file all Tax Returns required to be filed after the Closing with respect to the Assets, if any, and shall duly and timely pay all such Taxes shown to be due on such Tax Returns. For Property Tax purposes, any returns or filings with a lien or due date prior to Closing shall be prepared by Seller. Purchaser's preparation of any such Tax Returns shall be subject to Seller's approval, which approval shall not be unreasonably withheld or delayed. Purchaser shall make such Tax Returns available for Seller's review and approval (which approval shall not be unreasonably withheld or delayed) no later than fifteen (15) Business Days prior to the due date for filing such Tax Returns, it being understood that Seller's failure to approve any such Tax Returns shall not limit Purchaser's obligation to timely file such Tax Returns and duly and timely pay all Taxes shown to be due thereon. Not less than five (5) Business Days prior to the due date of any such Tax Return, Seller shall pay to Purchaser the amount shown as due on such Tax Returns as determined in accordance with Section 3.6 of this Agreement and shall, to the extent required by law, join in the execution of any such Tax Returns.

(c) With respect to pro-rated Property Taxes, specifically including but not limited to Property Tax Returns prepared and filed with any Tribal Authority, Seller's preparation of any such Tax Return shall be subject to Purchaser's approval, which approval shall not be unreasonably withheld or delayed. Seller shall make such tax Returns available for Purchaser's review and approval no later than fifteen (15) Business Days prior to the due date for filing such Tax Return, it being understood that Purchaser's failure to approve any such Tax Return shall not limit Seller's obligation to timely file such Tax Returns. In preparing and reviewing said Property Tax Returns, each Party shall cooperate and act in good faith to resolve any disagreement related to such Tax Returns as between the Parties or as between either Party and any Governmental Authority.

(d) Purchaser and Seller shall provide the other Party with such assistance as may reasonably be requested by the other Party in connection with the preparation of any Tax Return, any audit or other examination by any taxing authority, or any judicial or administrative proceedings relating to liability for Taxes, and each will retain and provide the requesting Party with any records or information which may be relevant to such return, audit or examination, proceedings or determination. Any information obtained pursuant to this <u>Section 6.5</u> or pursuant to any other Section hereof providing for the sharing of information or review of any Tax Return or other schedule relating to Taxes shall be kept confidential by the Parties hereto in accordance with Section 6.8.

(e) In the event that a dispute arises between Seller and Purchaser as to the amount of taxes, the Parties shall attempt in good faith to resolve such dispute, and any amount so agreed upon shall be paid to the appropriate Party. If such dispute is not resolved within thirty (30) days thereafter, the Parties shall submit the dispute to the Independent Accounting Firm for resolution, which resolution shall be final, conclusive and binding on the Parties. Notwithstanding anything in this Agreement to the contrary, the fees and expenses of the Independent Accounting Firm in resolving the dispute shall be borne equally by Seller and Purchaser. Any payment required to be made as a result of the resolution of the dispute by the Independent Accounting Firm shall be made within ten (10) days after such resolution, together with any interest determined by the Independent Accounting Firm to be appropriate.

(f) Seller hereby certifies that all Transfer Tax liabilities of Seller accruing before the Closing Date have been or will be fully satisfied or provided for. In the event Purchaser is assessed any Transfer Tax with respect to the Assets for any period prior to the Closing Date, Purchaser shall notify Seller promptly and shall provide Seller with a validly executed power of attorney authorizing Seller to act in Purchaser's stead with regard to the assessment. Whether Seller determines to contest any such assessment in whole or in part, Seller shall indemnify and hold harmless Purchaser, in connection with any assessment of Tax described in this Section 6.5, whether or not contested hereunder, to the extent such Tax is determined to be due and owing, together with interest and penalties as well as any expenses incurred (including legal fees that may be incurred by Purchaser) in participating in any action related to such assessment. If the laws of the State or the local taxing authority require payment of assessed Taxes as a condition to contesting or further contesting their applicability, Seller shall make such payments together with interest and penalties. Purchaser agrees to cooperate fully in initiating and pursuing any action directed by Seller for recovery of such payments and shall refund any amounts received (including interest and penalties) within three (3) days of receipt by Purchaser. Any action to contest Tax assessments hereunder or to recover Taxes paid hereunder by Seller on behalf of Purchaser shall be under the control of Seller and at Seller's sole cost and expense.

(g) Notwithstanding any other provision hereof, Purchaser covenants and agrees that, after the Closing Date, Purchaser will, to the extent practicable, and at Seller's expense, (i) provide or cause to be provided written notice to Seller sixty (60) days in advance of taking any of the actions specified on a Schedule to be provided by Seller to Purchaser, within one hundred twenty (120) days of the Effective Date, which Schedule shall be reasonably acceptable to Purchaser; listing actions, including discontinuing the operation of the Facilities, or modifications to the Assets which in Seller's reasonable opinion could result in a loss of the exclusion of interest on the Pollution Control Bonds from gross income for federal income tax purposes under Code Section 103, and (ii) take any reasonable actions which it has authority to take that are reasonably requested by Seller in writing for the purpose of maintaining such exclusion (including without limitation, inserting notification requirements, in operating manuals and posting notices within the Facilities). Notwithstanding anything in this Agreement to the contrary, (i) Purchaser will have no liability whatsoever in excess of \$250,000 to Seller or any other Person if Purchaser fails to comply with the covenants in the preceding sentence and

(ii) Purchaser shall not be required to take, or refrain from taking, any action inconsistent with Purchaser's rights or obligations under any of the Facilities Contracts. Purchaser further covenants and agrees that, in the event that Purchaser transfers any of the Assets, Purchaser, shall obtain from its transferee a covenant and agreement that is analogous to Purchaser's covenants and agreements in this Section 6.5(g) pursuant to the first sentence of this Section 6.5(g) as well as a covenant and agreement that is analogous to that of this sentence. This covenant shall survive Closing and shall continue in effect so long as such Pollution Control Bonds remain outstanding. Seller agrees to promptly notify Purchaser at such time as no Pollution Control Bonds remain outstanding. Seller will reimburse Purchaser for any expenses incurred by Purchaser, in connection with Purchaser's compliance with this Section 6.5(g) and any refundings thereof, issued or to be issued on behalf of Seller in connection with the Assets.

#### 6.6 <u>Risk of Loss.</u>

(a) Between the date hereof and the Closing Date, all risk of loss or damage to the property included in the Assets shall be borne by Seller.

(b) If, before the Closing Date, all or any portion of the Facilities or the Facilities Switchyard becomes subject to or is threatened with any condemnation or eminent domain proceeding, Seller shall notify Purchaser promptly in writing of such fact. If such taking would create a Material Adverse Effect, then Purchaser may, at its option, (i) receive from Seller an assignment of any claim, settlement or proceeds thereof and proceed with the transactions contemplated by this Agreement, or (ii) terminate this Agreement pursuant to Section 10.1.

If, before the Closing Date all or any portion of the Facilities or the (c) Facilities Switchyard are damaged or destroyed (whether by fire, theft, vandalism or other casualty) in whole or in part, and Seller's share of the fair market value of such damage or destruction or the cost of repair of the Facilities or the Facilities Switchyard that were damaged, lost or destroyed is less than fifteen percent (15%) of the Initial Purchase Price, Seller shall, at its option, either (i) reduce the Purchase Price by the lesser of (x) Seller's share of the fair market value of the Facilities or the Facilities Switchyard damaged or destroyed (such value to be determined as of the date immediately prior to such damage or destruction), or (y) Seller's share of the estimated cost to repair or restore the same (any disagreement with respect thereto being resolved in accordance with Section 11.10), (ii) upon the Closing, transfer the proceeds or the rights to the proceeds of applicable insurance to Purchaser, or (iii) bear Seller's share of the costs of repairing, or restoring such damaged or destroyed portions of the Facilities or the Facilities Switchyard and, at Seller's election, delay the Closing and any right to terminate this Agreement for a reasonable time necessary to accomplish the same. If any part of the Facilities or the Facilities Switchyard is damaged or destroyed (whether by fire, theft, vandalism or other casualty) in whole or in part prior to the Closing and the lesser of Seller's share of the fair market value of the Facilities or the Facilities Switchyard damaged or destroyed or Seller's share of the cost of repair is greater than fifteen percent (15%) of the Initial Purchase Price, then Purchaser may elect either to (x) require Seller upon the Closing to transfer the rights to Seller's share of proceeds (or the right to the proceeds) of applicable insurance to Purchaser and proceed with the transactions contemplated by this Agreement, or (y) terminate this Agreement.

6.7 **Cooperation Relating to Insurance.** Until the Closing, Seller will not take any action that will decrease the level of insurance coverage for the Facilities and the Facilities Switchyard as in effect on the date hereof, including, without limitation, property damage and liability insurance, unless agreed by the other Facilities Owners. In addition, Seller agrees to use Commercially Reasonable Efforts to assist Purchaser in making any claims against pre-Closing insurance policies of Seller that may provide coverage related to Assumed Liabilities. Purchaser agrees that it will indemnify Seller for its reasonable out-of-pocket expenses incurred in providing such assistance and cooperation. On and after the Closing, Seller authorizes the Operating Agent to take any actions necessary to remove Seller from any Facilities Insurance Policies and, except with respect to insurance rights retained by Seller pursuant to Sections 2.1(m), 2.1(o) or 2.2(h), Seller agrees to waive its rights with respect to such insurance coverage from and after the Closing. If requested by Seller, Purchaser agrees to exercise Commercially Reasonable Efforts to assist Seller, at Seller's cost, in obtaining so-called "tail" coverage in respect of claims brought after the Closing for events occurring prior to the Closing, including, if appropriate, listing Seller as an additional insured or named insured in policies of Purchaser and/or the Facilities Owners. Seller agrees that it will reimburse Purchaser for its reasonable out-of-pocket expenses incurred in providing such assistance to Seller in obtaining tail coverage.

#### 6.8 <u>Confidentiality.</u>

(a) General. Each Party (and its officers, employees, counsel, representatives and agents) will, using the same degree of care as that Party takes to preserve and safeguard its own confidential information, but not less than reasonable care, maintain in confidence and not disclose to third Persons, any Confidential Information received from the other Party (or its officers, employees, counsel, representatives and agents) in connection with the transactions contemplated by this Agreement. Each Party may disclose Confidential Information received from the other Party if and to the extent required by law, court order, subpoena or other lawful order of a Governmental Authority with jurisdiction, provided that the other Party is given written notice of such disclosure ten (10) days in advance, or as soon in advance as is reasonably practicable, or with the prior written consent of the other Party. If this Agreement is terminated pursuant to <u>ARTICLE 10</u>, each Party will return promptly, if so requested by the other Party, any Confidential Information provided to it and will use Commercially Reasonable Efforts to return any copies thereof that may have been provided to others in accordance with this Section 6.8. To the extent practicable, the Parties further agree, subject to Section 6.11, to not issue any public announcement, statement, press release or other public disclosure with respect to this Agreement or the transactions contemplated hereby; without the prior written consent of the other Party, which consent will not be unreasonably withheld. To the extent the provisions of this Section 6.8(a) conflict with the Confidentiality Agreement, as between Seller and Purchaser, this Section 6.8(a) shall control.

(b) <u>Regulatory Agencies</u>. Subject to <u>Section 6.1(a)</u>, upon the other Party's prior written approval (which, except as provided below, will not be unreasonably withheld), either Party may provide Confidential Information to the CPUC, ACC, FERC or any other Governmental Authority with jurisdiction as necessary to obtain Seller's Required Regulatory Approvals or Purchaser's Required Regulatory Approvals or approval under the HSR Act. The disclosing Party will seek confidential treatment for the Confidential Information provided to any

Governmental Authority and the disclosing Party will notify the other Party as far in advance as is practicable of its intention to release to any Governmental Authority any Confidential Information.

6.9 <u>Reasonable Cooperation</u>. Each Party agrees to use Commercially Reasonable Efforts to cooperate with the other Party to effect the consummation of the transactions contemplated by this Agreement, and to provide the other Party with such access or information related to the Facilities as may reasonably be requested in connection with such transactions. Without limiting the generality of the forgoing, the Parties shall work with each other prior to the Closing Date to determine if any Facilities Contract which is not currently listed on <u>Schedule 1.1.67</u> or <u>Schedule 1.1.77</u>, or approval of any Governmental Authority which is not currently listed on <u>Schedule 1.1.67</u> or <u>Schedule 1.1.77</u>, should be listed on such Schedule.

Title to Real Property and Leased Property. As soon as reasonably possible 6.10 after the Effective Date, Seller and Purchaser shall work cooperatively to cause Title Insurer to deliver a current preliminary title report on the real property and leased property included in the Assets, accompanied by legible copies of all documents referred to in the exception portion of such report, to Purchaser (the "Preliminary Title Report"). Purchaser shall have not more than thirty (30) days from the delivery date of the Preliminary Title Report in which to review and to give Seller and Title Insurer written notice of any title exception which is unacceptable to Purchaser, and, in the event any amendment is issued to the Preliminary Title Report, Purchaser shall have not more than thirty (30) days from the delivery of an amendment to deliver a written objection to any title exception, appearing for the first time in such amendment. If Purchaser is dissatisfied with any exception to title as shown in the Preliminary Title Report, then, Seller shall have until the Closing to eliminate any disapproved exceptions from the Preliminary Title Report, or obtain title insurance endorsements against such exceptions. If Seller cannot remove such exceptions or obtain title insurance endorsements before the Closing, then Purchaser may either cancel this Agreement, or Purchaser may waive such objections and the transaction shall close as scheduled, provided that if Purchaser disapproves any title exception that would otherwise qualify as a Permitted Encumbrance under Section 1.1.60 but for Purchaser's position that such title exception constitutes or will constitute a Material Adverse Effect, then Seller shall have the right to terminate this Agreement on fifteen (15) Business Days' notice given within thirty (30) days following Purchaser's disapproval of such title exception. Notwithstanding any other provision hereof, the following exceptions shall be deemed accepted by Purchaser and need not be removed or endorsed over: (a) Permitted Encumbrances, and (b) exceptions not objected to in writing by Purchaser during the time periods set forth above.

6.11 <u>Right of First Refusal</u>. Seller hereby agrees to promptly orally notify Purchaser, confirmed in writing, as to any notices received by Seller pursuant to Section 13 of the Facilities Co-Tenancy Agreement regarding the Facilities Owners' right of first refusal. If one or more of the Facilities Owners exercises their right of first refusal with respect to the Assets under the Facilities Co-Tenancy Agreement, Purchaser shall, subject to the terms and conditions of the Facilities Owners thereunder, automatically and without further notice to Seller be deemed to have exercised its right of first refusal with respect to the Assets to the maximum extent permitted by the Facilities Co-Tenancy Agreement. In the event one or more of the Facilities Owners exercises such right, the interest in the Assets to be transferred pursuant to this

Agreement and the Initial Purchase Price shall both be reduced to reflect the pro rata interest in the Assets to be purchased by Purchaser pursuant to Section 13.10 of the Facilities Co-Tenancy Agreement or as otherwise agreed to by Seller, Purchaser and the Facilities Owner exercising its right of first refusal, and Seller and Purchaser will enter into an amendment to this Agreement to reflect such reductions and other changes that the Parties deem appropriate.

**6.12** Exclusivity. During the term of this Agreement, and except as necessary to fulfill its obligations under the Facilities Co-Tenancy Agreement, or an order of the CPUC, Seller will (a) deal exclusively with Purchaser and other Facilities Owners that elect to participate in the purchase of the Assets under this terms of this Agreement ("Participating Owners"), and will not offer to sell, solicit offers to sell or negotiate with any third party for the sale of the Assets; and (b) promptly notify Purchaser and Participating Owners of any unsolicited offer, interest or inquiry by a third party concerning a possible purchase of the Assets and will not provide any information with respect to a possible sale of the Assets to any third party.

**Post Closing – Further Assurances.** At any time or from time to time after the 6.13 Closing; each Party, will, upon the reasonable request of the other Party, execute and deliver any further instruments or documents, and exercise Commercially Reasonable Efforts to take such further actions as may reasonably be required to fulfill and implement the terms of this Agreement or realize the benefits intended to be afforded hereby. After the Closing, and upon prior reasonable request, each Party shall exercise Commercially Reasonable Efforts to cooperate with the other, at the requesting Party's expense (but including only out-of-pocket expenses to third parties and not the costs incurred by any Party for the wages or other benefits paid to its officers, directors or employees), in furnishing non-privileged records, information, testimony and other assistance in connection with any inquiries, actions, audits, proceedings or disputes involving either of the Parties hereto (other than in connection with disputes between the Parties hereto) and based upon contracts, arrangements or acts of Seller, Purchaser, the other Facilities Owners or the Operating Agent on behalf of one or more of the Facilities Owners which were in effect or occurred on, prior to, or after Closing and which relate to the Assets, including, without limitation, arranging discussions with (and calling as a witness) officers, directors, employees, agents, and representatives of Purchaser or Seller.

#### 6.14 <u>Post Closing – Information and Records.</u>

(a) Following the Closing, Purchaser will not dispose of any books, records, documents or information reasonably relating to any Excluded Assets or Excluded Liabilities except in accordance with Purchaser's existing record retention policies. During such period, Purchaser will permit Seller to examine and make copies, at Seller's expense, of such books, records, documents and information for any reasonable purpose, including any litigation now pending or hereafter commenced against Seller, or the preparation of income or other Tax Returns. Seller will provide reasonable notice to Purchaser of its need to access such books, records, documents or other information.

(b) Seller shall not be entitled to examine or copy privileged and/or attorney work product documents or information pursuant to <u>Section 6.14(a)</u>. If privileged and/or attorney work product documents or information, including communications between Purchaser and its counsel, are disclosed to Seller in the books, records, documents or other information

made available by Purchaser, Seller agrees (1) such disclosure is inadvertent, (2) such disclosure will not constitute a waiver, in whole or in part, of any privilege or work product, (3) such information will constitute Confidential Information, and (4) Seller will promptly return to Purchaser (or will destroy or make inaccessible such Confidential Information to the extent reasonably possible and certify as such to Purchaser) all copies of such books, records, documents or other information in the possession of Seller.

6.15 <u>Post Closing – Landfill and Remediation Costs</u>. Purchaser covenants that: (a) it shall not cause the Landfill to be used or operated at any time after the Closing; and (b) in the event that Purchaser, without Seller's approval, enters into (i) any lease or lease amendment or extension or (ii) any other agreement of any kind with a Person other than a Governmental Authority, in either case altering or purporting to alter in any material respect any obligations of Seller with respect to Remediation of any Environmental Conditions or Hazardous Substances or the removal or Remediation of the Landfill, Purchaser shall hold Seller harmless from any incremental Remediation or removal costs, resulting from such lease, lease amendment or extension or other agreement.

#### <u>ARTICLE 7</u> INDEMNIFICATION

#### 7.1 Indemnification by Seller.

(a) <u>Purchaser Claims</u>. From and after the Closing, Seller will indemnify, defend and hold harmless Purchaser and its parents and Affiliates, and each of their officers, directors, employees, attorneys, agents and successors and assigns (collectively, the "Purchaser Group"), from and against any and all demands, suits, penalties, obligations, damages, claims, losses, liabilities, payments, costs and expenses (including reasonable legal, accounting and other expenses in connection therewith) and including costs and expenses incurred in connection with investigations, and settlement proceedings which arise out of, in connection with, or relate to, the following (collectively, "Purchaser Claims"):

(i) any breach or violation of any covenant or agreement of Seller set forth in this Agreement;

(ii) any breach or inaccuracy of the representations or warranties made by Seller contained in this Agreement in <u>ARTICLE 4;</u>

(iii) the Excluded Liabilities; and

(iv) any loss or damages resulting from or arising out of Seller's ownership of the Assets prior to Closing, except for any loss or damage resulting from or arising out of Assumed Liabilities.

(b) <u>Seller Limitations</u>. If the Closing occurs, the Purchaser Group will not be entitled to any punitive, incidental, indirect, special or consequential damages resulting from or arising out of any Purchaser Claims, including damages for lost revenues, income, profits or tax benefits, diminution in value of the Facilities, or any other damage or loss resulting from the

disruption to or loss of operation of the Assets, except to the extent due on any Third Party Claim.

## 7.2 Indemnification by Purchaser.

(a) <u>Seller Claims</u>. From and after the Closing, Purchaser will indemnify, defend and hold harmless Seller and its parents and Affiliates and each of their officers, directors, employees, attorneys, agents and successors and assigns (collectively, the "Seller Group"), from and against any and all demands, suits, penalties, obligations, damages, claims, losses, liabilities, payments, costs and expenses (including reasonable legal, accounting and other expenses in connection therewith) and including costs and expenses incurred in connection with, investigations and settlement proceedings which arise out of or relate to the following (collectively, "Seller Claims"):

(i) any breach or violation of any covenant or agreement of Purchaser set forth in this Agreement;

(ii) any breach or inaccuracy of any of the representations or warranties made by Purchaser contained in this Agreement in <u>ARTICLE 5;</u>

(iii) the Assumed Liabilities; and

(iv) any loss or damages resulting from or arising out of Purchaser's ownership or operation of the Assets from and after the Closing, except for any loss or damage resulting from or arising out of Excluded Liabilities.

(b) <u>Purchaser Limitations</u>. If the Closing occurs, the Seller Group will not be entitled to any punitive, incidental, indirect, special or consequential damages resulting from or arising out of any Seller Claim, including damages for lost revenues, income, profits or tax benefits, diminution in the value of the Facilities or any other damage or loss resulting from the disruption to or loss of operation of the Assets, except to the extent due on any Third Party Claim.

7.3 Notice of Claim. Subject to the terms of this Agreement and upon a Party's receipt of notice of the assertion of a claim or of the commencement of any suit, action or proceeding made or brought by any Person who is not a Party to this Agreement or an Affiliate, the Party seeking indemnification hereunder (the "Indemnitee") will promptly notify the Party against whom indemnification is sought (the "Indemnitee") in writing of any damage, claim, loss, liability or expense which the Indemnitee has determined has given or could give rise to a claim under Section 7.1 or Section 7.2. (The written notice is referred to as a "Notice of Claim.") A Notice of Claim will specify, in reasonable detail, the facts known to the Indemnitee regarding the claim. Subject to the terms of this Agreement, the failure to provide (or timely provide) a Notice of Claim will not affect the Indemnitee's rights to indemnification; provided, however, the Indemnitor is not obligated to indemnify the Indemnitee for the increased amount of any claim which would otherwise have been payable to the extent that the increase resulted from the failure to deliver timely a Notice of Claim.

7.4 Defense of Third Party Claims. The Indemnitor will defend, in good faith and at its expense, any claim or demand set forth in a Notice of Claim relating to a Third Party Claim and the Indemnitee, at its expense, may participate in the defense. The Indemnitee cannot settle or compromise any Third Party Claim so long as the Indemnitor is defending it in good faith. If the Indemnitor elects not to contest a Third Party Claim, the Indemnitee may undertake its defense, and the Indemnitor will be bound by the result obtained by the Indemnitee. The Indemnitor may at any time request the Indemnitee to agree to the abandonment of the contest of the Third Party Claim or to the payment or compromise by the Indemnitor of the asserted claim or demand. If the Indemnitee does not object in writing within fifteen (15) days of the Indemnitor's request, the Indemnitor may proceed with the action stated in the request. If within that fifteen (15) day period the Indemnitee notifies the Indemnitor in writing that it has determined that the contest should be continued, the Indemnitor will be liable under this ARTICLE 7 only for an amount up to the amount which the third party to the contested Third Party Claim had agreed to accept in payment or compromise as of the time the Indemnitor made its request. This Section 7.4 is subject to the rights of any Indemnitee's insurance carrier that is defending the Third Party Claim.

7.5 <u>Cooperation</u>. The Party defending the Third Party Claim will (a) consult with the other Party throughout the pendency of the Third Party Claim regarding the investigation, defense, settlement, trial, appeal or other resolution of the Third Party Claim; and (b) afford the other Party the opportunity to be associated in the defense of the Third Party Claim. The Parties will cooperate in the defense of the Third Party Claim. The Indemnitee will make available to the Indemnitor or its representatives all records and other materials reasonably required by them for use in contesting any Third Party Claim (subject to obtaining an agreement to maintain the confidentiality of confidential or proprietary materials in a form reasonably acceptable to Indemnitor and Indemnitee). If requested by the Indemnitor, the Indemnitee will cooperate with the Indemnitor and its counsel in contesting any Third Party Claim that the Indemnitor elects to contest or, if appropriate, in making any counterclaim against the Person asserting the claim or demand, or any cross-complaint against any Person. The Indemnitor will reimburse the Indemnitee for any expenses incurred by Indemnitee in cooperating with or acting at the request of the Indemnitor.

7.6 <u>Mitigation and Limitation on Claims</u>. As used in this Agreement, the term "Indemnifiable Claim" means any Purchaser Claims or Seller Claims. Notwithstanding anything to the contrary contained herein:

(a) <u>Reasonable Steps to Mitigate</u>. The Indemnitee will take all reasonable steps to mitigate all losses, damages and the like relating to an Indemnifiable Claim, including availing itself of any defenses, limitations, rights of contribution, claims against third Persons and other rights at law or equity, and will provide such evidence and documentation of the nature and extent of the Indemnifiable Claim as may be reasonably requested by the Indemnitor. The Indemnitee's reasonable steps include the reasonable expenditure of money to mitigate or otherwise reduce or eliminate any loss or expense for which indemnification would otherwise be due under this <u>ARTICLE 7</u>, and the Indemnitor will reimburse the Indemnitee for the Indemnitee's reasonable expenditures in undertaking the mitigation, together with, interest thereon from the date of payment to the date of repayment at the "prime rate" as published in *The Wall Street Journal*.

(b) <u>Net of Benefits</u>. Any Indemnifiable Claim is limited to the amount of actual damages sustained by the Indemnitee by reason of such breach or nonperformance.

(c) <u>Minimum Claim</u>. No Party shall have any liability or obligation to indemnify under Section 7.1(a)(ii) or Section 7.2(a)(ii), as the case may be, unless the aggregate amount for which such Party would be liable thereunder, but for this provision, exceeds One Million Dollars (\$1,000,000), and recovery shall be limited only to such amounts as exceed One Million Dollars (\$1,000,000). For purposes of the foregoing, individual claims of Fifteen Thousand Dollars (\$15,000) or less shall not be aggregated for purposes of calculating such deductible threshold amount or for calculating damages in excess of such amount. Nothing in this Section 7.6 is intended to modify or limit a Party's liability or obligation hereunder for other Indemnifiable Claims or to constitute an assumption by Purchaser of any Excluded Liability or an assumption by Seller of any Assumed Liability.

7.7 <u>Exclusivity</u>. Except for intentional fraud, following the Closing, the rights and remedies of Seller, on the one hand, and Purchaser, on the other hand, for money damages under this Article are, solely as between Seller on the one hand and Purchaser on the other hand, exclusive and in lieu of any and all other rights and remedies for money damages which each of Seller on the one hand, and Purchaser on the other hand, may have under this Agreement under applicable Law with respect to any Indemnifiable Claim, whether at common law or in equity.

#### ARTICLE 8 CONDITIONS PRECEDENT TO OBLIGATIONS OF PURCHASER AT THE CLOSING

The obligations of Purchaser under this Agreement to complete the purchase of the Assets and assume the Assumed Liabilities are subject to the satisfaction or waiver, or deemed satisfaction or waiver, on or prior to the Closing, of each of the following conditions precedent:

8.1 <u>Compliance with Provisions</u>. Seller has performed or complied in all material respects with all covenants, agreements and conditions contained in this Agreement on its part required to be performed or complied with at or prior to the Closing.

**8.2** <u>HSR Act</u>. The waiting period under the HSR Act applicable to the consummation of the sale of the Assets contemplated hereby shall have expired or been terminated.

**8.3** <u>Injunction</u>. No preliminary or permanent injunction or other order or decree by any federal or state court or Governmental Authority which prevents the consummation of the sale of the Assets contemplated herein shall have been issued and remain in effect (each Party agreeing to cooperate in all efforts to have any such injunction, order or decree lifted) and no Law shall have been enacted by any state or federal government or Governmental Authority, which prohibits the consummation of the sale of the Assets.

**8.4** <u>Required Regulatory Approvals</u>. Without limiting the generality of <u>Sections 6.1(a)</u> and <u>6.4</u>, Purchaser shall have received all of Purchaser's Required Regulatory Approvals and Seller shall have received all of Seller's Required Regulatory Approvals; without limiting the generality of the foregoing, Purchaser shall have obtained a final order no longer

subject to appeal from the ACC approving the purchase by Purchaser of its share of the Facilities including financial and economic terms and conditions and the ratemaking treatment of the transaction under this Agreement, all in form and substance reasonably satisfactory to Purchaser, and without significant conditions, modifications of the transaction or qualifications in the order that are not reasonably acceptable to Purchaser.

**8.5** <u>Representations and Warranties</u>. The representations and warranties of Seller set forth in this Agreement (without giving effect to materiality, Material Adverse Effect, or similar phrases in such representations and warranties) shall be true and correct as of the Closing Date, in each case as though made at and as of the Closing Date, except as would not individually or in the aggregate result in a Material Adverse Effect.

**8.6** <u>Officer's Certificate</u>. Purchaser shall have received a certificate from Seller, executed by an authorized officer, dated the Closing Date, to the effect that the conditions set forth in <u>Sections 8.1, 8.4</u> (insofar as it relates to Seller's Required Regulatory Approvals), <u>8.5</u> and <u>8.10</u> (insofar as it relates to Seller's Required Consents) have been satisfied by Seller.

**8.7** <u>Title Policy Insurance</u>. Title to Assets comprised of interests in real property and leased property shall have been evidenced by the willingness of a title insurer mutually agreeable to the Parties (the "Title Insurer") to issue at regular rates ALTA owner's, or lessee's, as the case may be, extended coverage policies of title insurance (1990 Form B) (the "Title Policies"), with the general survey and creditors' rights exceptions removed, in amounts equal to the portion of the Purchase Price allocated to such interests, showing title to such interests in such real property vested in Purchaser in the condition described in <u>Section 6.10</u>, subject only to Permitted Encumbrances, and transfer of such interest to Purchaser. The willingness of Title Insurer to issue the Title Policies shall be evidenced either by the issuance thereof at the Closing or by the title Insurer's delivery of written commitments or binders, dated as of the Closing (but insuring title as of the date title conveyance documents are recorded), to issue such Title Policies within a reasonable time after the Closing Date, subject to actual transfer of the real property in question.

**8.8** <u>Material Adverse Effect</u>. Subject to <u>Section 6.6</u>, since the Effective Date, no Material Adverse Effect shall have occurred and be continuing with respect to the Facilities and the Facilities Switchyard.

**8.9** <u>Liens</u>. Any and all liens and encumbrances (other than Permitted Encumbrances) on the Assets, constituting personal property shall have been released and any documents necessary to evidence such release shall have been delivered to Purchaser.

8.10 <u>Seller's Required Consents</u>. Without limiting the generality of <u>Sections 6.1(a)</u> and <u>6.4</u>, all of Seller's Required Consents shall have been obtained.

**8.11** <u>No Termination</u>. Neither Party has exercised any termination right such Party is entitled to exercise pursuant to <u>Section 10.1</u>.

**8.12** <u>**Right of First Refusal and Notice**</u>. The right of first refusal and notice periods set forth in Sections 13.3 and 13.4 of the Facilities Co-Tenancy Agreement shall have expired or all Facilities Owners other than Purchaser shall have either waived or exercised their right of first

refusal (and, in the event of an exercise of such right of first refusal, Seller and Purchaser shall have entered into the amendment to this Agreement contemplated by <u>Section 6.11</u>).

**8.13** <u>Termination Agreement</u>. Concurrently with the Closing, closing shall have occurred under the Termination Agreement.

**8.14** Facilities Lease Amendments. Amendment No. 2 to the Facilities Lease shall have become effective and Amendment No. 3 to the Facilities Lease shall have been executed by the Navajo Nation and each of the Facilities Owners other than Seller, both in substantially the form provided by Purchaser to Seller on the Effective Date.

**8.15** <u>Fuel Agreement</u>. Purchaser and BHP shall have executed an amendment or replacement to the Facilities Fuel Agreement extending the period under which coal is to be supplied thereunder until 2041, on terms reasonably acceptable to Purchaser.

## <u>ARTICLE 9</u>

## **CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER AT THE CLOSING**

The obligations of Seller under this Agreement to complete the sale of the Assets and transfer the Assets and Assumed Liabilities to Purchaser are subject to the satisfaction or waiver, or deemed satisfaction or waiver, on or prior to the Closing, of each of the following conditions precedent:

**9.1** <u>Compliance with Provisions</u>. Purchaser has performed or complied in all material respects with all covenants, agreements and conditions contained in this Agreement on its part required to be performed or complied with at or prior to the Closing.

**9.2** <u>HSR Act</u>. The waiting period under the HSR Act applicable to the consummation of the sale of the Assets contemplated hereby shall have expired or been terminated.

**9.3** <u>Injunction</u>. No preliminary or permanent injunction or other order or decree by any federal or state court or Governmental Authority which prevents the consummation of the sale of the Assets contemplated herein shall have been issued and remain in effect (each Party agreeing to use its best efforts to have any such injunction, order or decree lifted) and no Law shall have been enacted by any state or federal government or Governmental Authority in the United States which prohibits the consummation of the sale of the Assets.

**9.4** <u>Approvals</u>. Without limiting the generality of <u>Sections 6.1(a)</u> and <u>6.4</u>, Purchaser shall have received all of Purchaser's Required Regulatory Approvals, and Seller shall have received all of Seller's Required Regulatory Approvals; without limiting the generality of the foregoing, Seller will have obtained a final order no longer subject to appeal from the CPUC approving the application for, inter alia, the sale of the Assets by Seller and the ratemaking treatment of the transaction under this Agreement, including Seller's proposed cost recovery mechanism, all in form and substance reasonably satisfactory to Seller, and without significant conditions, modifications of the transaction or qualifications in the order that are not reasonably acceptable to Seller, and Seller shall have obtained written approval of the transaction and the termination of the transmission capacity under the Edison-Arizona Transmission Agreement

from the California ISO or written confirmation from the California ISO that such approval is not required.

9.5 <u>Representations and Warranties</u>. The representations and warranties of Purchaser set forth in this Agreement (without giving effect to materiality, Material Adverse Effect, or similar phrases in such representations and warranties) shall be true and correct as of the Closing Date, in each case as though made at and as of the Closing Date, except as would not individually or in the aggregate result in a Material Adverse Effect, it being understood and agreed to by the Parties that, with respect to Purchaser's representation in <u>Section 5.4</u> hereof, Purchaser's Knowledge, for purposes of this <u>Section 9.5</u>, will mean Purchaser's Knowledge as of the Closing Date, and Purchaser will be entitled to supplement its written disclosure to Seller through the Closing Date.

**9.6** Officer's Certificate. Seller shall have received a certificate from Purchaser, executed by an authorized officer, dated the Closing Date, to the effect that the conditions set forth in Sections 9.1, 9.4 (insofar as it relates to Purchaser's Required Regulatory Approvals), 9.5 and 9.9 (insofar as it related to Purchaser's Required Consents) have been satisfied by Purchaser.

**9.7** <u>No Termination</u>. Neither Party has exercised any termination right such Party is entitled to exercise pursuant to <u>Section 10.1</u>.

**9.8** <u>Right of First Refusal</u>. The right of first refusal and notice periods set forth in Sections 13.3 and 13.4 of the Facilities Co-Tenancy Agreement shall have expired or all Facilities Owners other than Purchaser shall have either waived or exercised their right of first refusal (and, in the event of an exercise of such right of first refusal, Seller and Purchaser shall have entered into the amendment to this Agreement contemplated by <u>Section 6.11</u>).

**9.9** <u>Purchaser's Required Consents</u>. Without limiting the generality of <u>Sections 6.1(a)</u> and <u>6.4</u>, all of Purchaser's Required Consents shall have been obtained, subject to <u>Section 3.7</u>, and the Closing shall not result in a material breach by Seller of a material Facilities Contract.

**9.10** <u>Material Adverse Effect</u>. Subject to <u>Section 6.6</u>, since the Effective Date, no Material Adverse Effect shall have occurred and be continuing with respect to the Facilities and the Facilities Switchyard.

**9.11** <u>Termination Agreement</u>. Concurrently with the Closing, closing shall have occurred under the Termination Agreement.

#### ARTICLE 10 TERMINATION

10.1 <u>Rights To Terminate</u>. This Agreement, or to the extent specifically permitted herein a portion thereof, may, by written notice given on or prior to the Closing Date, in the manner provided in <u>Section 11.10</u>, be terminated at any time prior to the Closing Date (or such other date as may be set forth below):

(a) by Seller if there has been a misrepresentation with respect to Purchaser's representations and warranties in this Agreement (without giving effect to materiality, Material Adverse Effect, or similar phrases in such representations and warranties) that would result in a Material Adverse Effect, or a material default or breach by Purchaser with respect to the due and timely performance of any of Purchaser's covenants and agreements contained in this Agreement, and such misrepresentation; default or breach is not cured by the earlier of the Closing Date or the date thirty (30) days after receipt by Purchaser, of written notice specifying particularly such misrepresentation, default or breach;

(b) by Purchaser if there has been a misrepresentation with respect to Seller's representations and warranties in this Agreement (without giving effect to materiality, Material Adverse Effect, or similar phrases in such representations and warranties) that would result in a Material Adverse Effect, or a material default or breach by Seller with respect to the due and timely performance of any of Seller's covenants and agreements contained in this Agreement, and such misrepresentation, default or breach is not cured by the earlier of the Closing Date, or the date thirty (30) days after receipt by Seller of written notice specifying particularly such misrepresentation, default or breach;

(c) by Purchaser if Purchaser is not at the time of termination in breach of this Agreement, upon written notice to Seller, (i) if any of Purchaser's Required Regulatory Approvals shall have been denied (and a petition for rehearing or refiling of an application initially denied without prejudice shall also have been denied), or shall have been granted but are not in form and substance reasonably satisfactory to Purchaser (including, adverse conditions relating to Purchaser or the Assets), or (ii) if the CPUC has not approved the transaction by March 31, 2012; provided that Purchaser may only exercise the termination right described in this clause (ii) prior to the time the CPUC approves the transaction;

(d) by Seller if Seller is not at the time of termination in breach of this Agreement, upon written notice to Purchaser, if any of the Seller's Required Regulatory Approvals shall have been denied (and a petition for rehearing or refiling of an application initially denied without prejudice shall also have been denied), or shall have been granted but are not in form and substance reasonably satisfactory to Seller (including adverse conditions relating to Seller or the Assets);

- (e) by Purchaser in accordance with <u>Section 6.6;</u>
- (f) by mutual agreement of Seller and Purchaser; or

(g) by Seller or Purchaser if the conditions to such Party's Closing have not occurred by December 31, 2012, unless the Party seeking to terminate is then in breach of this Agreement.

**10.2** <u>Effect of Termination</u>. If this Agreement is terminated pursuant to <u>Section 10.1</u>, all further obligations and liabilities of the Parties hereunder will terminate, except (i) as set forth in <u>ARTICLE 7</u> or as otherwise contemplated by this Agreement, (ii) for the obligations set forth in <u>Sections 4.12</u>, <u>5.6</u> and <u>6.8</u> and <u>ARTICLE 11</u>, and (iii) for the obligations of the Parties set forth in the Confidentiality Agreement. In the event this Agreement is terminated by Purchaser

pursuant to Section 10.1(b) or Seller pursuant to Section 10.1(a), the non-breaching party shall be entitled to liquidated damages in an amount equal to two percent (2%) of the Initial Purchase Price. Upon termination, the originals of any items, documents or written materials provided by one Party to the other Party will be returned by the receiving Party to the providing Party, and any Confidential Information retained by the receiving Party will be kept confidential.

10.3 Specific Performance; Limitation of Damages. Seller acknowledges that the transactions contemplated by this Agreement are unique and that Purchaser will be irreparably injured should such transactions not be consummated in a timely fashion. Consequently, Purchaser will not have an adequate remedy at law if Seller shall fail to transfer, assign and convey the Assets when required to do so hereunder. In such event, prior to any termination of this Agreement pursuant to Section 10.1, Purchaser shall have the right, in addition to any other remedy available in equity or law, to specific performance of such obligation by Seller, subject to Purchaser's performance of its obligations hereunder. Purchaser acknowledges that the transactions contemplated by this Agreement are unique and that Seller will be irreparably injured should such transactions not be consummated in a timely fashion. Consequently, Seller will not have an adequate remedy at law if Purchaser shall fail to purchase the Assets when required to do so hereunder. In such event, prior to any termination of this Agreement pursuant to Section 10.1, Seller shall have the right, in addition to any other remedy available in equity or law, to specific performance of such obligation by Purchaser, subject to Seller's performance of its obligations hereunder. Except as otherwise provided in Section 7.1(a)(iv), 7.2(a)(iv) and 10.2, neither Party will be entitled to any punitive, incidental, indirect, special or consequential damages, including damages for lost revenues, income or profits, resulting from or arising out of a breach of this Agreement, whether or not the Closing occurs.

#### ARTICLE 11

### **MISCELLANEOUS AGREEMENTS AND ACKNOWLEDGMENTS**

11.1 <u>Purchaser as Operating Agent</u>. Notwithstanding the sale of the Assets and the assignment of the Facilities Contracts by Seller to Purchaser, the actions or inactions of Purchaser, in its capacity as Operating Agent, insofar as they may affect Retained Environmental Liabilities or Excluded Liabilities, shall continue to be subject to the standard of conduct and the limitations on liability set forth in Section 22 of the Facilities Operating Agreement in effect at the time of Closing and the retention by Seller of Retained Environmental Liabilities and of rights under the Facilities Contracts with respect to Excluded Liabilities shall not impose a different standard of conduct on Purchaser, in its capacity as Operating Agent, or change in any manner the limitations of liability of Purchaser, in its capacity as Operating Agent, to Seller under the Facilities Contracts with respect to any actions of the Operating Agent that may affect Retained Environmental Liabilities or Excluded Liabilities. Nothing contained in this Section 11.1 shall excuse or limit Purchaser's performance of the specific covenants set forth in this Agreement in accordance with their terms.

11.2 <u>Expenses</u>. Except as otherwise provided herein, each Party is responsible for its own costs and expenses (including attorneys' and consultants' fees, costs and expenses) incurred in connection with this Agreement and the consummation of the transactions contemplated by this Agreement.

11.3 <u>Entire Document</u>. This Agreement (including the Exhibits and Schedules to this Agreement) the Ancillary Agreements and the Confidentiality Agreement contain the entire agreement between the Parties with respect to the transactions contemplated hereby and supersede all negotiations, representations, warranties, commitments, offers, contracts and writings (except for the Confidentiality Agreement) prior to the execution date of this Agreement, written or oral. No waiver and no modification or amendment of any provision of this Agreement is effective unless made in writing and duly signed by the Parties referring specifically to this Agreement, and then only to the specific purpose, extent and interest so provided.

11.4 <u>Schedules</u>. The Parties agree and acknowledge that the Schedules in this Agreement may be incomplete or subject to revision prior to the Closing, subject to <u>Section 6.2</u>. The Parties will cooperate and work in good faith to complete and update such Schedules in a manner consistent with the requirements of this Agreement. The Schedules delivered pursuant to the terms of this Agreement are an integral part of this Agreement to the same extent as if they were set forth verbatim herein.

11.5 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which is an original, but all of which together constitute one and the same instrument.

11.6 <u>Severability</u>. If any provision hereof is held invalid or unenforceable by any arbitrator or as a result of future legislative action, this holding or action will be strictly construed and will not affect the validity or effect of any other provision hereof. To the extent permitted by law, the Parties waive, to the maximum extent permissible, any provision of law that renders any provision hereof prohibited or unenforceable in any respect.

11.7 <u>Assignability</u>. This Agreement is binding upon and inures to the benefit of the successors and assigns of the Parties, but is not assignable by any Party without the prior written consent of the other Party.

**11.8** <u>Captions</u>. The captions of the various Articles, Sections, Exhibits and Schedules of this Agreement have been inserted only for convenience of reference and do not modify, explain, enlarge or restrict any of the provisions of this Agreement.

11.9 <u>Governing Law</u>. The validity, interpretation and effect of this Agreement are governed by and will be construed in accordance with the laws of the state in which the Facilities are located applicable to contracts made and performed in such state and without regard to conflicts of law doctrines except to the extent that certain matters are preempted by Federal law or are governed by the law of the jurisdiction of organization of the respective Parties.

#### 11.10 Dispute Resolution.

(a) <u>Intent of the Parties</u>. Subject to <u>ARTICLE 7</u> with respect to an Indemnifiable Claim, <u>Section 3.2(e)</u> with respect to Post-Closing Adjustments, <u>Section 3.5</u> with respect to the Purchase Price allocation, and Section 6.5(e) with respect to disputes regarding Taxes, and except as provided in <u>Section 11.10(b)</u>, the sole process available to either Party for resolution of any dispute or claim arising out of or relating to this Agreement or any Ancillary Agreement shall be the dispute resolution procedure set forth in this <u>Section 11.10</u>. If the Parties

12193761.3

54

cannot resolve a dispute under <u>Sections 11.10(c)</u> or <u>(d)</u>, then the dispute shall be settled through final and binding arbitration under <u>Section 11.10(e)</u>.

Provisional Relief. The Parties acknowledge and agree that irreparable (b) damage would occur if certain provisions of this Agreement are not performed in accordance with the terms of this Agreement, that money damages would not be a sufficient remedy for any breach of these provisions of this Agreement, and that the Parties shall be entitled, without the requirement of posting a bond or other security, to seek a preliminary injunction, temporary restraining order, or other provisional relief as a remedy for a breach of this Agreement in any court of competent jurisdiction, notwithstanding the obligation to submit all other disputes (including all claims for monetary damages under this Agreement) to mediation and arbitration pursuant to Sections 11.10(d) or (e). The Parties further acknowledge and agree that the results of the arbitration may be rendered ineffectual without the provisional relief. Such a request for provisional relief does not waive a Party's right to seek other remedies for the breach of the Agreement, notwithstanding any prohibition against claim-splitting or other similar doctrine. The other remedies that may be sought include specific performance and injunctive or other equitable relief, plus any other remedy specified in this Agreement for the breach of the provision, or if the Agreement does not specify a remedy for the breach, all other remedies available at law or equity to the Parties for the breach.

(c) <u>Management Negotiations</u>. The Parties will attempt in good faith to resolve any dispute or claim arising out of or relating to this Agreement or an Ancillary Agreement promptly by negotiations between a vice president (or more senior officer) of Seller or his or her designated representative and an executive of similar authority of Purchaser. Either Party may give the other Party written notice of any dispute or claim. Within twenty (20) days after delivery of said notice, the executives will confer by telephone or meet at a mutually acceptable time and place, and thereafter as often as they reasonably deem necessary to exchange information and to attempt to resolve the dispute or claim. If the matter has not been resolved within sixty (60) days of the first meeting, either Party (by notice to the other Party) may submit the controversy for non-binding mediation pursuant to Section 11.10(d).

(d) Mediation. Either Party may initiate mediation by providing Notice to the other Party in accordance with Section 11.11 of a written request for mediation, setting forth a description of the dispute and the relief requested. The Parties will cooperate with one another in selecting the mediator ("Mediator") from the panel of neutrals from Judicial Arbitration and Mediation Services, Inc. ("JAMS"), its successor, or any other mutually acceptable non-JAMS Mediator, and in scheduling the time and place of the mediation. Such selection and scheduling will be completed within forty-five (45) days after Notice of the request for mediation. Unless otherwise agreed to by the Parties, the mediation will not be scheduled for a date that is greater than one hundred twenty (120) days from the date of Notice of the request for mediation. The Parties covenant that they will participate in the mediation in good faith, and that they will share equally in its costs (other than each Party's individual attorneys' fees and costs related to the Party's participation in the mediation, which fees and costs will be borne by such Party). All offers, promises, conduct and statements, whether oral or written, made in connection with or during the mediation by either of the Parties, their agents, representatives, employees, experts and attorneys, and by the Mediator or any of the Mediator's agents, representatives and employees, will not be subject to discovery and will be confidential, privileged and inadmissible

for any purpose, including impeachment, in any arbitration or other proceeding between or involving the Parties, or either of them, provided, evidence that is otherwise admissible or discoverable will not be rendered inadmissible or non-discoverable as a result of its use in the mediation.

(e) Arbitration. Either Party may initiate final and binding arbitration with respect to the matters first submitted to mediation by providing Notice of a demand for binding arbitration before a single, neutral arbitrator (the "Arbitrator") at any time following the unsuccessful conclusion of the mediation provided for above. The Parties will cooperate with one another in selecting the Arbitrator within sixty (60) days after Notice of the demand for arbitration and will further cooperate in scheduling the arbitration to commence no later than one hundred eighty (180) days from the date of Notice of the demand. To be qualified as an Arbitrator, each candidate must be a retired judge of a trial court of any state or federal court, or retired justice of any appellate or supreme court. Unless otherwise agreed to by the Parties, the individual acting as the Mediator will be disqualified from serving as the Arbitrator in the dispute, although the Arbitrator may be another member of the JAMS panel of neutrals or such other panel of neutrals from which the Parties have agreed to select the Mediator. Upon Notice of a Party's demand for final and binding arbitration, such dispute submitted to arbitration, including the determination of the scope or applicability of this Agreement to arbitrate, will be determined by final and binding arbitration before the Arbitrator, in accordance with the laws of the State of New Mexico, without regard to principles of conflicts of laws. Except as provided for herein, the arbitration will be conducted by the Arbitrator in accordance with the rules and procedures for arbitration of complex business disputes for the organization with which the Arbitrator is associated. Notwithstanding the rules and procedures that would otherwise apply to the arbitration, and unless the Parties agree to a different arrangement, the place of the arbitration will be Phoenix, Arizona, if arbitration is initiated by Seller, and Los Angeles, California, if arbitration is initiated by Purchaser.

Also notwithstanding the rules and procedures that would otherwise apply to the arbitration, and unless the Parties agree to a different arrangement, discovery will be limited as follows:

(i) Before discovery commences, the Parties shall exchange an initial disclosure of all documents and percipient witnesses which they intend to rely upon or use at any arbitration proceeding (except for documents and witnesses to be used solely for impeachment);

(ii) The initial disclosure will occur within thirty (30) days after the initial conference with the Arbitrator or at such time as the Arbitrator may order;

(iii) Discovery may commence at any time after the Parties' initial

disclosure;

(iv) The Parties will not be permitted to propound any interrogatories or requests for admissions;

(v) Discovery by each Party will be limited to twenty-five (25) document requests (with no subparts), three (3) lay witness depositions, and three (3) expert

witness depositions (unless the Arbitrator holds otherwise following a showing by the Party seeking the additional documents or depositions that the documents or depositions are critical for a fair resolution of the dispute or that a Party has improperly withheld documents);

(vi) Each Party is allowed a maximum of three (3) expert witnesses, excluding rebuttal experts;

(vii) Within sixty (60) days after the initial disclosure, or at such other time as the Arbitrator may order, the Parties shall exchange a list of all experts upon which they intend to rely at the arbitration proceeding;

(viii) Within thirty (30) days after the initial expert disclosure, each Party may designate a maximum of two (2) rebuttal experts;

(ix) Unless the Parties agree otherwise, all direct testimony will be in form of affidavits or declarations under penalty of perjury; and

(x) Each Party shall make available for cross examination at the arbitration hearing its witnesses whose direct testimony has been so submitted.

The Arbitrator will have the authority to grant any form of equitable or legal relief a Party might recover in a court action. The Parties acknowledge and agree that irreparable damage would occur if certain provisions of this Agreement are not performed in accordance with the terms of the Agreement, that money damages would not be a sufficient remedy for any breach of these provisions of this Agreement, and that the Parties shall be entitled, without the requirement of posting a bond or other security, to specific performance and injunctive or other equitable relief as a remedy for a breach of this Agreement. Judgment on the award may be entered in any court having jurisdiction. The Arbitrator must, in any award, allocate all of the costs of the binding arbitration (other than each Party's individual attorneys' fees and costs related to the Party's participation in the arbitration, which fees and costs will be borne by such Party), including the fees of the Arbitrator and any expert witnesses, against the Party who did not prevail; provided that if neither Party prevails completely such costs shall be allocated in favor of the Party who substantially prevailed as determined by the Arbitrator. Until such award is made, however, the Parties will share equally in paying the costs of the arbitration.

11.11 <u>Notices</u>. All notices, requests, demands and other communications under this Agreement must be in writing and must be delivered in person or sent by certified mail, postage prepaid, or by overnight delivery, and properly addressed as follows:

If to Seller:

Southern California Edison Company 2244 Walnut Grove Avenue Rosemead, California 91770 Attention: Chief Financial Officer

With a copy to:

Southern California Edison Company 2244 Walnut Grove Avenue Rosemead, California 91770 Attention: General Counsel

If to Purchaser:

Arizona Public Service Company 400 North Fifth Street, Station 9085 Phoenix, Arizona 85004 Attn: Mark A. Schiavoni, Senior Vice President of Fossil Operations

With a copy to:

Arizona Public Service Company 400 North Fifth Street, Station 8695 Phoenix, Arizona 85004 Attn: Shirley Baum, Associate General Counsel

With a copy to:

Ballard Spahr LLP 1735 Market Street, 51st Floor Philadelphia, Pennsylvania 19103 Attention: Robert C. Gerlach, Esq.

Any Party may from time to time change its address for the purpose of notices to that Party by a similar notice specifying a new address, but no such change is effective until it is actually received by the Party sought to be charged with its contents.

All notices and other communications required or permitted under this Agreement which are addressed as provided in this <u>Section 11.10</u> are effective upon delivery, if delivered personally, or by overnight delivery, and, are effective five (5) days following deposit in the United States mail postage prepaid if delivered by mail.

11.12 <u>Time is of the Essence</u>. Time is of the essence of each term of this Agreement Without limiting the generality of the foregoing, all times provided for in this Agreement for the performance of any act will be strictly construed.

11.13 <u>No Third Party Beneficiaries</u>. Except as may be specifically set forth in this Agreement, nothing in this Agreement, whether express or implied, is intended to confer any rights or remedies under or by reason of this Agreement on any Persons other than the Parties and their respective permitted successors and assigns, nor is anything in this Agreement intended to relieve or discharge the obligation or liability of any third Persons to any Party, nor give any third Persons any right of subrogation or action against any Party.

11.14 <u>No Joint Venture</u>. Nothing contained in this Agreement creates or is intended to create an association, trust, partnership, or joint venture or impose a trust or partnership duty, obligation, or liability on or with regard to any Party.

**11.15** <u>Construction of Agreement</u>. Ambiguities or uncertainties in the wording of this Agreement will not be construed for or against any Party, but will be construed in the manner that most accurately reflects the Parties' intent as of the date they executed this Agreement.

11.16 Effect of Closing Over Known Unsatisfied Conditions or Breached <u>Representations, Warranties or Covenants</u>. If Seller or Purchaser elects to proceed with the Closing with Knowledge by it of any failure to be satisfied of any condition in its favor or the breach of any representation, warranty or covenant by the other Party, the condition that is unsatisfied or the representation, warranty or covenant which is breached at the Closing Date will be deemed waived by such Party, and such Party will be deemed to fully release and forever discharge the other Party on account of any and all claims, demands or charges, known or unknown, with respect to the same.

11.17 <u>Conflicts</u>. In the event of any conflicts or inconsistencies between the terms of this Agreement and the terms of any of the Ancillary Agreements, the terms of this Agreement will govern and prevail.

11.18 <u>Waiver of Compliance</u>. To the extent permitted by applicable Law, any failure of any of the Parties to comply with any obligation, covenant, agreement or condition set forth herein may be waived by the Party entitled to the benefit thereof only by a written instrument signed by such Party, but any such waiver shall not operate as a waiver, of, or estoppel with respect to, any prior or subsequent failure to comply therewith. The failure of a Party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights.

#### 11.19 <u>Survival.</u>

(a) The representations and warranties given or made by any Party in <u>ARTICLE 4</u> or <u>ARTICLE 5</u> hereof or in any certificate or other writing furnished in connection herewith shall survive the Closing indefinitely.

(b) The covenants and agreements of the Parties contained in this Agreement, including those set forth in <u>ARTICLE 7</u>, shall survive the Closing indefinitely, unless otherwise specified herein.

(c) The obligations of the Parties in <u>Section 6.8</u> will survive (i) the termination of this Agreement, (ii) the discharge of all other obligations owed by the Parties to each other, (iii) any transfer of title to the Assets and (iv) the Closing of the transactions contemplated in this Agreement.

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

# **SOUTHERN CALIFORNIA EDISON COMPANY,** a California corporation

By:

Name: Title:

## ARIZONA PUBLIC SERVICE COMPANY,

an Arizona corporation

By:

Name: Title:

Signature Page to Purchase and Sale Agreement

# Schedules to Purchase and Sale Agreement

## Schedule 1.1.50(a)

### Seller's Officers, Employees, and Knowledgeable Persons

Russ Krieger – Vice President, Power Production John Dayton – Manager of Business Planning and Development, Power Production Steven Pickett – Senior Vice President and General Counsel Daniel Cobb – Alternate E&O Committee Member

### Schedule 1.1.50(b)

#### Purchaser's Officers, Employees and Authorized Agents

Mark Schiavoni – Senior Vice President Fossil David Hansen – Vice President Fossil Operations David Falck – Executive Vice President, General Counsel and Secretary John Franchini – Fossil Plant Manager Four Corners Susan Kidd – Director Coal/Co-Owned Generation Nick Svor – General Manager Generation Engineering Frank Perkins – Plant Manager Four Corners Units 4, 5 Richard Grimes – Four Corners Environmental Section Leader

## Schedule 1.1.50(c)

Operating Agent's Officers, Employees and Authorized Agents See Schedule 1.1.50(b) which is incorporated herein by reference.

# PNW Plans

- -
- Pinnacle West Capital Corporation Retirement Plan Pinnacle West Capital Corporation Group Life and Medical Plan -

Purchaser's Required Consents

None

-

# Purchaser's Required Regulatory Approvals

- Arizona Corporation Commission -
- -
- Federal Energy Regulatory Commission Hart-Scott-Rodino Antitrust Improvements Act of 1976 -

# Seller's Required Consents

Trustee under Seller's Mortgage

-

## Seller's Required Regulatory Approvals

- California Public Utilities Commission
- Federal Energy Regulatory Commission
- California Independent System Operator
- Hart-Scott-Rodino Antitrust Improvements Act of 1976

## Schedule 2.1(b)

# Leased Real Property

- Facilities Lease
- The real property interests described in Exhibits 2 9 of the Facilities Lease
- See Schedule 2.1(c) which is incorporated herein by reference

# Schedule 2.1(c)

# Rights-of-Way/Easements and Water Rights

| Item | Existing § 323<br>Grants | Property or Facility                   | APS File # | Grant Date | Expiration<br>Date |
|------|--------------------------|--|------------|------------|--------------------|
| 1    | Plant Site               | New Lease (Units 4-5)                  |            | 07/06/66   | 07/06/16           |
| 1    |                          | New Lease (Units 4-5)                  |            | 0//00/00   | 07/00/10           |
| 2    | Ancillary<br>Facilities  | Utah Mine Haul Road (Communication     |            |            |                    |
|      |                          | Lines and Access Road)                 | IN-13      | 07/28/61   | 07/28/11           |
|      |                          | Plant – Coal Lease Area – 69 kV        | IN-15      | 12/15/61   | 12/15/11           |
|      |                          | Pumping Station to Plant Access Road   |            |            |                    |
|      |                          | Pipeline                               | IN-12      | 04/02/62   | 04/02/12           |
|      |                          | River Pumping Station to Plant – 69 kV | IN-11      | 04/02/62   | 04/02/12           |
|      |                          | Plant – EPNG Bridge / Access Rd        | IN-16      | 07/03/63   | 07/03/13           |
|      |                          | Pumping Station to Plant Access Road   |            |            |                    |
|      |                          | Pipeline Addition                      | IN-92      | 04/21/69   | 04/21/19           |
### Schedule 2.1(h)

### Seller Facilities Contracts

- 1. § 323 Grants.
- 2. Facilities Lease.
- 3. Facilities Co-Tenancy Agreement.
- 4. Facilities Operating Agreement.
- 5. Restated and Amended Four Corners Fuel Agreement Number 2, dated August 31, 2003, by and among BHP Navajo Coal Company and the Participants, as the same may be amended.
- 6. Conditional Partial Assignment of Fuel Agreement Number 2, dated September 2, 1966, by and among Utah Construction & Mining Co. and the Participants.
- 7. Memorandum for Recordation of Original Four Corners Fuel Agreement and of Four Corners Fuel Agreement No. 2 and Imposition of Equitable Servitude and Covenant Running with the Land, dated September 2, 1966, by and among Utah Construction & Mining Co. and the Participants.
- 8. Facilities Fuel Agreement.
- 9. Four Corners Project Emission Abatement System Operating Power Agreement, dated October 15, 1982 among the Participants.
- 10. Four Corners Project Unit Tripping Agreement, dated May 23, 1969 among the Participants.
- 11. Four Corners Units 4 & 5 Capital Improvements Design and Construction Agreement, dated March 23, 1981 among the Participants.
- 12. Agreement to Purchase and Sell Undivided Interest in the Reserve Auxiliary Power Source Four Corners Project, dated August 15, 1968 among the Participants.
- 13. Exchange Agreement dated March 28, 1967 among the Participants with Letter of Clarification dated March 28, 1967, a Supplemental Letter Agreement dated February 9, 1972 and Ruling of Internal Revenue Service with Letter of Transmittal.
- 14. Four Corners Designated Representative Agreement, dated March 18, 1994, by and among the Participants, John R. Denman and D. Craig Walling.
- 15. Four Corners Designated Representative Agreement Assignment and Novation, dated October 22, 2002 from D. Craig Walling to David L. Saliba.
- 16. Four Corners Designated Representative Agreement Assignment and Novation Form, dated July 31, 2009 from John R. Denman to David L. Saliba as the new designated representative and Richard Grimes as the new alternate designated representative.
- 17. Four Corners Designated Representative Agreement Assignment and Novation Form, dated January 31, 2010, from David L. Saliba to Frank E. Perkins.

12193761.3

- 18. Principals of Interconnected Operation Four Corners Project dated May 12, 1969, among the Participants as amended by Amendment No. 1 dated April 29, 1974, among the Participants.
- 19. Water Supply Agreement, dated March 2, 2007 between the Jicarilla Apache Nation, BHP Navajo Coal Company, APS on behalf of itself and with respect to Units 4 and 5 the Four Corners Participants and Public Service Company of New Mexico on behalf of itself and the San Juan Participants.
- 20. Voluntary Compliance Agreement Air Quality, dated May 18, 2005, by and among the Navajo Nation, Salt River Project Agricultural Improvement and Power District, as operating agent for the Navajo Generating Station ("NGS") and with the express written consent of each participant of NGS and APS, as operating agent for the Four Corners Power Plant and with the express written consent of each Participant.
- 21. Tax Settlement and Closing Agreement, dated August 13, 2002, by and between the Seller and the Office of the Navajo Nation Uniform Tax Administration Statute.
- 22. Shiprock-Four Corners Project 345-kV Switchyard Interconnection Agreement, dated October 2, 2002, by and among the Facilities Owners and Public Service Company of Colorado, Tri-State Generation and Transmission Association, Inc., and Western Area Power Administration.

### Schedule 2.1(p)

### Miscellaneous Assets

None

-

### Schedule 2.2(a)

**Excluded** Assets

None

-

12193761.3

### Schedule 3.6(a)(iii)

### Operating and Maintenance Expense Pro-Rations

The following costs and expenses incurred for the applicable period during which the Closing occurs shall be pro-rated between the Parties:

- 1. Seller is responsible for the operation and maintenance expenses as defined in the Facilities Operating Agreement, Section 17, Operating and Maintenance Expenses, incurred prior to the Closing Date, including but not limited to the following:
  - a. Outside services and materials and supplies, including all administrative and general loads, for operating and maintaining the plant; and
  - b. Payroll including related administrative and general, payroll taxes and benefits expenses.
- 2. Employee Incentive Plan payroll including related administrative and general, payroll taxes and benefits expenses.
- 3. Fuel expenses (Coal and Gas).
- 4. Insurance premiums.
- 5. Navajo Land Lease.
- 6. Environmental Operating Permit.
- 7. Ash Hauling Agreement costs.
- 8. All related royalties and taxes for Operating and Maintenance expenses and Fuel expenses.

12193761.3

### Schedule 6.5(g)

### Pollution Control Bonds

\$55,540,000 City of Farmington, New Mexico 5.125% Pollution Control Refunding Revenue Bonds (Southern California Edison Company Four Corners Project) 1999 Series A

City of Farmington, New Mexico Pollution Control Refunding Revenue Bonds (Southern California Edison Company Four Corners Project) \$103,460,000 2005 Series A (Non-AMT)

City of Farmington, New Mexico Pollution Control Refunding Revenue Bonds (Southern California Edison Company Four Corners Project) \$100,000,000 2005 Series B (Non-AMT)

1

Exhibit A-1



Exhibit A-1

12193761.3



Four Corners Power Plant Aerial Photo taken on 12/18/1979 EXHIBIT A-2 IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

SOUTHERN CALIFORNIA EDISON COMPANY, a California comporation

By:

Name: R.W. Krieger Title: Vice President Power Production

ARIZONA PUBLIC SERVICE COMPANY, an Arizona corporation

By:

١

Name: Title:

Signature Page to Purchase and Sale Agreement

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

### SOUTHERN CALIFORNIA EDISON COMPANY,

a California corporation

By:

Name: Title:

### **ARIZONA PUBLIC SERVICE COMPANY,**

an Arizona corporation

1 B<u>y:</u>

Name: Donald G. Robinson Title: President and Chief Operating Officer

Signature Page to Purchase and Sale Agreement

# Direct Testimony of Patrick Dinkel

| . 1 |   |
|-----|---|
| 2   |   |
| 3   |   |
| 4   |   |
| 5   |   |
| 6   |   |
| 7   |   |
| 8   | DIRECT TESTIMONY OF PATRICK DINKEL          |
| 9   | On Behalf of Arizona Public Service Company |
| 10  | On Denan of Arizona Public Service Company  |
| 11  | Docket No. E-01345A-10                      |
| 12  |   |
| 13  |   |
| 14  |   |
| 15  |   |
| 16  |   |
| 17  |   |
| 18  |   |
| 19  |   |
| 20  |   |
| 21  |   |
| 22  |   |
| 23  |   |
| 24  |   |
| 25  |   |
| 26  |   |
| 27  | November 22, 2010                           |
| 28  |   |
| _0  |   |
|     |   |
| l   | 1   |

| I.   | Table of Contents                            |
|------|--|
| I.   | INTRODUCTION                                 |
| II.  | THE PROPOSED TRANSACTION BENEFITS CUSTOMERS. |
| III. | CONCLUSION                                   |
|      | Loads and Resources TableAttachment PD       |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
| -    |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |
|      |  |

### DIRECT TESTIMONY OF PATRICK DINKEL ON BEHALF OF ARIZONA PUBLIC SERVICE COMPANY (Docket No. <u>E-01345A-10-</u>)

I. INTRODUCTION

### **Q.** PLEASE STATE YOUR NAME AND POSITION WITH APS.

A. My name is Patrick Dinkel. I am the Vice President of Power Marketing, Resource Planning and Acquisition at Arizona Public Service Company ("APS" or "Company"). In that capacity, I am responsible for power marketing and trading, the integrated resource planning function, long-term generation acquisition, and the Company's Renewable Energy Program.

## Q. WHAT IS YOUR EDUCATIONAL AND PROFESSIONAL BACKGROUND?

A. I received a Bachelor of Science degree from Marymount College and a Master of Business Administration from Northern Arizona University. I joined APS in 1986. Prior to being named Vice President of Power Marketing and Resource Planning and Acquisition, I was General Manager of Strategic Planning and Resource Acquisition, where I was also responsible for overseeing APS's long-term power procurement and renewable energy programs. Before that, I was Director of Resource Acquisitions and Renewable Energy, and have also been responsible for Corporate Planning and Business Unit Analysis and Reporting. During my career at APS, I have held various positions within APS and Pinnacle West Capital Corporation, primarily within the renewable energy, financial, and budgeting areas.

### 

Q.

## HAVE YOU PREVIOUSLY TESTIFIED BEFORE THE ARIZONA CORPORATION COMMISSION ("COMMISSION")?

 A. Yes. I testified in support of APS's requests to acquire the Sundance Assets (Docket No. E-01345A-04-0407), and, later, to include those assets in rate base (Docket No. E-01345A-05-0816). I also testified in support of APS's request for authorization to acquire the Yuma Assets (Docket No. E-01345A-06-0464), in

support of the Commission's grant of a Certificate of Environmental Compatibility for Abengoa Solar (Docket No. L-00000GG-08-0407-00139 and L-00000GG-08-0408-00140), and in the recent APS rate case (Docket No. E-01345A-08-0172).

## 6

Q.

1

2

3

4

5

7

8

9

10

11

12

24

25

26

27

28

## WHAT IS THE PURPOSE OF YOUR DIRECT TESTIMONY IN THIS PROCEEDING?

A. My testimony supports APS's application for authorization and other support needed to purchase Southern California Edison's ("SCE") existing ownership interest in Four Corners Power Plant ("Four Corners") Units 4 and 5 and retire Units 1-3 of that plant. Specifically, I will describe how that transaction benefits APS customers and makes good business sense from a resource planning perspective.

- 13 II. <u>THE PROPOSED TRANSACTION BENEFITS CUSTOMERS.</u>
- 14Q.YOU NOTED ABOVE THAT APS'S PROPOSAL TO ACQUIRE SCE'S<br/>SHARE OF FOUR CORNERS AND RETIRE UNITS 1-3 BENEFITS<br/>CUSTOMERS. PLEASE ELABORATE.

Simply put, the proposed transaction is the best value for APS customers Α. 16 compared to every reasonable resource alternative. Let me explain. The energy 17 APS receives from its current ownership interest in the Four Corners generating 18 station Units 4 and 5 represents 6% of APS's energy resources. If no one 19 acquires SCE's ownership interest in Four Corners, there is a risk that the co-20 owners of Units 4 and 5 will choose to retire those units. A shutdown of Units 4 21 and 5 results in APS losing 231 MW of a reliable and economic baseload 22 resource now serving APS customers. 23

Four Corners Units 1-3 provide APS customers with 560 MW, or 4200 GWH, of baseload energy. Although Units 1-3 currently comply with all environmental regulations, they will require significant environmentally-driven capital investment over the next five years if they are to remain in service. The first

expected tranche, \$235 million for mercury emission controls, could come as early as 2014; the second, a potential \$351 million to comply with the EPA's proposed Best Available Retrofit Technology ("BART") visibility requirements, is due as early as 2016. Units 1-3 are cost-effective for APS customers now, but that may no longer be true if a total of \$586 million must be spent in five short years to keep them online. Other costs may also be required for those units to comply with future greenhouse gas regulations. In other words, there is a risk that all of Four Corners could close by 2016.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

18

19

20

21

22

23

24

25

26

27

28

If all five units are retired, APS will lose 791 MW of low-cost base load generation that currently provides 19% of APS total generation needs. Navajo Generating Station, in which APS, SRP, and TEP each own a share, faces many of the same issues. If it closes, APS would lose yet another 315 MW of baseload capacity, posing the risk that APS could lose 1,106 MW – that is 26% of its energy – in just a few years.

## 16 Q. WHAT ALTERNATIVE RESOURCES ARE AVAILABLE TO REPLACE LOST FOUR CORNERS GENERATION? 17 17

A. Coal is a baseload resource and a fundamental component of APS's energy mix. A baseload resource is one that is designed to run 24 hours a day, seven days a week, to meet the Company's lowest around-the-clock demand. Continually called on, such a resource must be both reliable and cost-effective, or else customers will pay more for their energy. Potential replacement alternatives for any lost Four Corners generation include coal and nuclear (large, conventional "baseload" resources), geothermal and biomass/biogas (small, renewable baseload resources), and natural gas (an "intermediate" resource that is reliable although it has greater fuel cost volatility compared to others and is most costeffective when serving peak load). Solar and wind generation, while increasingly important components of APS's energy mix, are intermittent resources that a

utility cannot control and that cannot adequately substitute for one that is required night and day, 365 days each year.

### MORE SPECIFICALLY, ASSUMING THAT PLANT PARTICIPANTS OPT TO CLOSE UNITS 4 AND 5 IN 2016, HOW WOULD APS REPLACE THE RESULTING 231 MW CAPACITY LOSS?

Few of the alternative resources discussed in my prior answer are realistically available to fill the energy void left APS if Four Corners Units 4 and 5 were to shut down in 2016. Arizona does not have sufficient geothermal resources to provide such capacity, and the geothermal that is available in Southern California has many potential buyers competing for this limited resource. Any geothermal plant that might be constructed would be too small (e.g., 50 MW) to address the void left by the retirement of the coal plants. Arizona also has highly limited amounts of biogas and biomass available, and APS will continue to seek those resources irrespective of the outcome of this application. Nuclear energy takes at least ten years to develop, and requires a large upfront capital investment. Putting aside that capital outlay, a new nuclear resource would certainly not be available until several years past the 2016 need date. While energy efficiency will fill a portion of these requirements, APS is already committed to aggressively pursuing its cost effective energy efficiency programs. In any case, energy efficiency cannot be a complete solution – a point well-demonstrated in Graph 4 on page 11 of my Testimony, which compares what APS's energy mix will look like if the Company's Application is approved to what it will be if it is not.

This leaves APS with three potential options: (1) continue to operate Units 1-3 (which still leaves APS 231 MW short in 2016 if Units 4 and 5 shutdown, possibly rising to 546 MW if Navajo Generating Station retires); (2) replace any power lost from Four Corners with combined-cycle gas generation; or (3) retire

28

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

Q.

Α.

Units 1-3 and acquire SCE's interest in Units 4-5. Analysis of these options clearly shows that it is most beneficial to APS customers to retire Units 1-3 early and replace their output with the purchase of SCE's interest in Units 4 and 5.

Q.

A.

### OF THE OPTIONS YOU DESCRIBE, WHY IS THE TRANSACTION PROPOSED IN THE COMPANY'S APPLICATION THE MOST BENEFICIAL TO CUSTOMERS?

There are several reasons. First, from a cost perspective, customers will pay less under the proposed transaction than under either of the alternatives. This point is well demonstrated in the following two graphs, as well as through traditional revenue requirements analysis.





Graph 1 compares, on a dollar per kilowatt basis, the initial capital dollars that APS would pay for various generation resources. For the Four Corners-related alternatives, the noted value includes the cost of installing all required environmental controls, a \$294 million cash acquisition price, and the assumption of certain decommissioning and mine reclamation liabilities for SCE's additional

739 MW.<sup>1</sup> The graph shows that consummating the proposed transaction and installing potential environmental upgrades at Units 4 and 5 is the lowest cost alternative in terms of up-front cost.



Graph 2 compares, on a dollar per megawatt hour basis, the total cost of the generation resource, fully integrated into the electrical system, levelized over the full life cycle of the plant. For the Four Corners-related alternatives, the noted values include the cost of the environmental upgrades and an assumed internalized carbon price of \$20/ton, beginning in 2013. The current carbon price is \$0/ton; however, we believe the cost of carbon should be considered as an environmental factor in the resource decision-making process. This graph shows that the proposed transaction is the lowest cost for customers over the project life, compared to the alternatives.

<sup>&</sup>lt;sup>27</sup> <sup>1</sup> See Testimony of Mark Schiavoni at 6-7 for a description of the Purchase and Sale Agreement between 28 APS and SCE.

Finally the cost of the alternatives can be communicated in terms of the net present value of customer revenue requirements. In comparing these three alternatives, the acquisition of SCE's share of Units 4-5 results in a revenue requirement that is \$500 million less than the alternative of replacing the retired Four Corners energy with natural gas generation. The proposed transaction also results in a revenue requirement that is \$1 billion less than the alternative of investing in and continuing to run Units 1-3 over the same timeframe.

It is clear that none of the alternative resource scenarios brings the same cost benefit to APS customers as that proposed here. Consider the potential for keeping Units 1-3 in service, for example. In that case, as Graph 1 illustrates, APS customers will pay 44% more in capital costs to install the emission controls likely needed on Units 1-3 to keep those units in service than they will under the proposed transaction, an analysis that includes the cost of making the necessary environmental upgrades on Units 4 and 5. Moreover, this option simply preserves a resource that is already serving APS customers and does nothing to replace the other 231 MW of cost-effective generation that APS would forego if Units 4 and 5 retire in 2016, or protect against the potential loss of another 315 MW at Navajo Generating Station not long thereafter. APS customers would incur that much more in replacement power costs if the Company pursued this option.

Retiring Units 4 and 5 in favor of Units 1-3 also makes little sense from an operational perspective, given that Units 1-3 are smaller and less efficient, and lack the same economies of scale benefits of Units 4 and 5. By way of example, the cost of installing SCRs on Units 1-3 is approximately \$627 per kW, while the cost of installing the same equipment on Units 4-5 is roughly \$325 per kW – a significant difference.

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

Q.

### YOU NOTED THAT NATURAL GAS WAS A SECOND ALTERNATIVE TO THE PROPOSED TRANSACTION. PLEASE DISCUSS THAT OPTION.

Natural gas generation is a reliable economic resource which effectively meets Α. the marginal resource needs of a utility. It has been the "measuring stick" that APS has used in recent years when evaluating all resource alternatives conventional or renewable. However, the drawbacks of using natural gas to replace 231 MW or more of existing Four Corners capacity are significant. First, the gas option is much more expensive than the approach proposed in the Company's Application. Apart from the capital costs associated with additional combined cycle generation, a new gas resource would require APS both to build new transmission infrastructure, and to maintain the current schedule of nowplanned transmission lines. As Graph 1 on page 5 of my Testimony shows, the cost of building new combined-cycle and transmission infrastructure is double the cost of purchasing SCE's share of Units 4 and 5 and installing the required environmental controls on those units, on a dollar per kilowatt basis. Moreover, as Graph 2 depicts, APS customers will pay almost 20% more per megawatt hour over the life cycle of a new gas plant than they will if APS acquires SCE's interest in Units 4 and 5.

In addition, unlike Four Corners' fuel costs, made dependable by virtue of a negotiated long-term fuel agreement with the supplier, gas prices are highly volatile, as well-evidenced by the following graph:

28



APS has conducted sensitivity analyses that demonstrate that the economic advantage of acquiring SCE's interest in Four Corners persists over a wide range of factors. In order to break even with the life cycle cost of the proposed transaction, natural gas prices would have to be 20% lower than the current long-term forecast. Or, the price assigned to carbon would have to rise above \$50 per ton. Alternatively, replacement combined-cycle gas costs would have to be half of current cost estimates to build that resource. The following illustrates these sensitivities:



### APS CUSTOMER BENEFITS DUE TO SCE TRANSACTION

Gas is a reliable resource that has an important place in a utility's resource portfolio, but if APS's resource mix becomes too dependent on natural gas, our customers will be highly exposed to potential fuel cost increases and volatility. APS's resource choices, like those of all power generators, each have a variety of trade-offs. This is why having a diverse energy mix, which reduces reliance on any single power source, mitigates risk and makes good business sense.

Q. HOW WOULD REPLACING APS'S SHARE OF ITS EXISTING FOUR CORNERS COAL CAPACITY WITH NATURAL GAS IMPACT THE COMPANY'S RESOURCE PORTFOLIO?

A. As Graph 4 shows, if APS replaces 791 MW of its existing coal capacity with natural gas generation, the Company's resource diversity decreases and customer reliance on natural gas generation increases by 90%, with natural gas making up 40% of the Company's generation. Having 40% of the Company's generation dependent upon potentially volatile natural gas markets would put APS and its customers at a significantly higher level of risk.



## Q. ARE THERE ANY OTHER REASONS WHY THE NATURAL GAS ALTERNATIVE IS A LESS PREFERABLE ONE?

A. Yes. There is also a practical risk to replacing the Four Corners output with natural gas. Additional gas generation and the associated transmission must be sited, permitted and constructed in a very short time frame if it is to be serving APS customers by 2016. As with any construction project, there is always the risk that projects will be delayed and the resources will not be available to customers when needed. Moreover, to execute this contingency, APS's currently planned and certificated Morgan to Sun Valley transmission line (commonly known as "TS-5 to TS-9") would need to be energized by 2016 – a feat which may prove difficult given the unresolved right-of-way issues for that project. The tight time clock not only makes the Four Corners alternative more appealing, but demonstrates the practical need for having this application processed quickly.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

0.

A.

### DOES APS HAVE A NEED FOR THE CAPACITY IT WILL ACQUIRE AS A RESULT OF THIS TRANSACTION?

APS's Loads and Resource table ("L&R"), attached to my Yes, it does. Testimony as Attachment PD-1, shows that APS will require another 545 MW of resources to meet its 2017 load requirements even if this transaction moves forward. That calculation also assumes the addition of over 1400 MW of renewable resources and energy efficiency programs. If the proposed transaction fails. APS's need for new resources could increase to over 1,500 MWs in 2017. Output from Navajo Generating Station may also be lost to similar vulnerabilities, giving need for yet another 315 MW of replacement power. Were both Four Corners and Navajo Generating Station to shut down entirely, APS's existing base load resources would be limited to Cholla Power Plant (providing a total of 647 MW) and Palo Verde Nuclear Generating Station (providing 1,146 MW) – a total of 1,793 MW to serve a 2020 minimum system demand of 2,530 MW. Such a scenario would dramatically increase APS's reliance on natural gas and our customers' exposure to gas price volatility.

Given that potential, the long-term need for maintaining sufficient, reliable base load resources is clear. The proposed transaction essentially preserves a wellbalanced energy supply portfolio for APS, with a slight net increase of 179 MW – a small difference that is unavoidable under the circumstances. That additional 179 MW provides protection against volatile natural gas prices as well as the potential loss of the Navajo Generating Station capacity. APS also expects to further defer the need for new base load generation if the transaction is approved.

24 25

26

27

28

Q.

## DID APS CONSIDER PROCURING RESOURCES FROM THE COMPETITIVE WHOLESALE MARKET AS AN ALTERNATIVE TO THE PROPOSED TRANSACTION?

Yes. APS has looked at what exists in the competitive wholesale market, but none of its offerings reasonably compare to the transaction with SCE. As

discussed above, gas-fired generation – the most practical alternative to Four Corners in these circumstances – would further expose APS customers to uncertain gas prices and require that new transmission be built for any gas-fired power to reach the Company's primary load center in the Metropolitan Phoenix area. Any potential plant acquisition price is especially uncertain given the fact that the need would not be until 2017. Although APS might also procure new coal, any such resource would have significant development risk, a cost well above that of the Four Corners acquisition price, and could not be built in time to meet the Company's need.

## IS THE APPROACH OUTLINED IN APS'S APPLICATION CONSISTENT WITH ITS LONG-TERM RESOURCE PLAN?

Yes. APS's L&R table indicates that, even after acquiring SCE's share of Four Α. 12 Corners Units 4 and 5 and retiring Units 1-3, APS will still need over 500 MWs 13 of resources in the 2017 timeframe. This L&R table also includes APS's 14 commitment to exceed compliance with the Renewable Energy Standard, and 15 meet the Commission's ambitious and recently adopted Energy Efficiency 16 Standard. The Resource Plan currently on file with the Commission also stresses 17 the value of maintaining a diverse energy supply portfolio – one that balances 18 coal, gas, and nuclear generation to complement the ever-growing role of 19 renewable resources and energy efficiency in meeting its customers' energy 20 needs. Acquiring the SCE interest in Units 4 and 5, combined with the early 21 retirement of Units 1-3, is thus fully consistent with the Company's resource 22 plans. 23

24

25

26

27

28

Q.

Α.

1

2

3

4

5

6

7

8

9

10

11

Q.

### THE APPLICATION REQUESTS THAT THE COMMISSION RULE ON THIS MATTER EXPEDITIOUSLY. WHY IS THAT IMPORTANT?

If the Commission rejects the Company's requests, Four Corners Units 4 and 5, and possibly Units 1-3, risk closing no later than 2016. APS must start working to implement a contingency plan, accelerating the acquisition and construction of

new generation and transmission infrastructure and/or installing emission control devices on Units 1-3. Without a timely order from this Commission, time may run out to construct or buy new replacement generation.

III. <u>CONCLUSION</u>

## DO YOU HAVE ANY CONCLUDING REMARKS TO YOUR TESTIMONY?

A. The proposal outlined in the Company's Application simply makes good sense for APS and our customers. It has the lowest relative capital cost, greatest cost certainty, and allows APS to maintain a reliable and cost-effective source of base load generation – all while improving the plant's environmental impact and stabilizing the local economies, as APS witness Mark Schiavoni describes. It also has the lowest customer rate impact, as APS witness Jeff Guldner explains. Although, there will be significant capital cost requirements in the short term, the approach outlined in this application provides nearly a \$500 million net present value benefit to APS customers. This opportunity is fully consistent with APS's obligation to provide cost effective, reliable, and environmentally conscious service to our customers and the communities we serve. It is one worth seizing.

**Q**.

### DOES THAT CONCLUDE YOUR TESTIMONY?

A. Yes.

**Q**.

# **Attachment PD-1**

Attachment PD -1 Page 1 of 1

> WITH ACQUISITION OF SCE'S SHARE OF FOUR CORNERS 4-5 **APS LOAD AND RESOURCE PLAN**

1,454 12,486 12,486 1,339 335 1,001 3,062 254 5,991 2025 11,032 6,449 6,495 1,574 <del>4</del> 0 10,723 1,413 12,137 6,449 46 105 1,301 305 866 2,962 102 12,137 2024 1,199 6.600 5,536 10,417 1,373 11,790 11,790 2023 6,449 46 105 1,264 275 867 867 2,360 2,360 424 5,189 1,199 6,600 10,113 1,333 11,445 2022 6,449 56 105 11,445 1,228 248 587 587 587 412 412 4,835 6,610 920 11,103 9,812 1,292 11,103 2021 6,449 206 105 6,760 1,201 587 587 498 4,343 920 10,817 2020 9,616 1,201 10,817 6,449 686 105 7,240 1,126 212 489 1,300 450 3,577 86 10,486 2019 9,317 1,169 10,486 6,449 1,283 105 7,837 008 196 489 559 397 2,649 840 10, 153 6,449 1,283 105 7,837 10,153 2018 9,012 1,141 889 182 489 303 303 2,316 2,316 740 6,449 1,283 105 7,837 8,709 1,112 9,820 9,820 2017 741 2016 8,371 1,083 9,454 6,449 1,767 105 8,321 9,661 ,340 0 60 490 0 741 2015 8,035 1,053 9,088 6,449 2,262 8,794 8,794 9,957 ,163 741 2014 7,810 1,039 8,849 6,449 2,258 83 8,790 9,786 741 \* Note resources shown as their expected capacity contribution at time of system peak 2013 7,510 1,011 6,449 2,258 83 8,790 9,296 8,521 334 208 2012 7,303 6,270 2,264 83 8,616 8,936 8,298 995 52800 224 320 7,164 987 8,151 6,270 2,275 59 8,603 8,728 2011 50025005 2 Contracts on Existing Resources (at Peak) New Renewable Nameplate Capacity • Future Planned Resource Additions: Load Requirements: Peak Demand (prior to EE and DE) Total Future Resource Additions: Renewable Energy Resources \* Distributed Renewable Energy \* Short-Term Market Purchases Demand Response Contract Total Existing Resources: **Fotal Load Requirements Reserve Requirements** Existing Resources: Gas Resources / DR Existing Generation Energy Efficiency \* Total Resources: 17 6 1 2 2 2 2 2 9 **50 ~ 00 0** - 0 0 <del>4</del>

Existing resources exclude short-term market purchase or sale transactions

New Renewable Nameplate Capacity includes all future renewable resources plus any existing renewable resources in service after Jan 1, 2010 (Lines 8 and 17)

MW

# **Direct Testimony of Jeffrey B. Guldner**

| 1        |  |
|----------|--|
| 2        |  |
| 3        |  |
| 4        |  |
| 5        |  |
| 6        |  |
| 7        | DIRECT TESTIMONY OF JEFFREY B. GULDNER             |
| 8        | DIRECT TESTIMONT OF JEFFRET B. GULDNER             |
| 9        | <b>On Behalf of Arizona Public Service Company</b> |
| 10       |  |
| 11       | Docket No. E-01345A-10                             |
| 12       |  |
| 13       |  |
| 14       |  |
| 15       |  |
| 16       |  |
| 17       |  |
| 18       |  |
| 19<br>20 |  |
| 20       |  |
| 21       |  |
| 23       |  |
| 24       |  |
| 25       |  |
| 26       |  |
| 27       | November 22, 2010                                  |
| 28       |  |
|          |  |
|          |  |

| 1    |      | Table of Contents   |
|------|------|---|
| 2    | I.   | INTRODUCTION  |
| 3    | II.  | SUMMARY OF THE PROPOSED TRANSACTION                           |
| 4    | III. | RATE IMPACT OF THE PROPOSED TRANSACTION                       |
| 5    | IV.  | APS'S COMMITMENT TO RENEWABLE ENERGY AND ENERGY<br>EFFICIENCY |
| 5    | V.   | THE COMPANY'S REQUESTS  |
| 7    | VI.  | CONCLUSION  |
| 3    |      |   |
| )    |      |   |
| )    |      |   |
| .    |      |   |
| 2    |      |   |
|      |      |   |
|      |      |   |
|      |      |   |
|      |      |   |
|      |      |   |
|      |      |   |
|      |      |   |
| )    |      |   |
|      |      |   |
|      |      |   |
| - 11 |      |   |
|      |      |   |
|      |      |   |
| 5    |      |   |
|      |      |   |
| 7    |      |   |
| 3    |      |   |
|      |      | i   |
|      |      |   |

### DIRECT TESTIMONY OF JEFFREY B. GULDNER **ON BEHALF OF ARIZONA PUBLIC SERVICE COMPANY** (Docket No. E-01345A-10-

#### INTRODUCTION I.

### PLEASE STATE YOUR NAME AND POSITION WITH ARIZONA **PUBLIC SERVICE COMPANY ("APS" OR "COMPANY")?**

My name is Jeffrey B. Guldner. I am Vice President of Rates and Regulation for APS. My business address is 400 N. 5<sup>th</sup> Street, Phoenix, Arizona, 85004.

#### YOUR **EDUCATIONAL** AND IS PROFESSIONAL WHAT **BACKGROUND?**

I joined APS in 2004 as Director of Regulatory Compliance and then assumed Α. 9 responsibility for federal regulation and policy at the Company. Prior to joining 10 APS, I was a partner in the Phoenix office of Snell & Wilmer LLP, where I practiced energy and public utilities law. My practice focused primarily on 12 electric utility rate and regulatory matters, including rate cases, power plant and 13 transmission line siting, energy project finance, and utility mergers. Before 14 practicing law, I was a Surface Warfare Officer in the United States Navy. I 15 received a B.A. in political science from the University of Iowa and my J.D., 16 magna cum laude, from Arizona State University.

#### WHAT IS THE PURPOSE OF YOUR DIRECT TESTIMONY IN THIS 18 Q. **PROCEEDING?** 19

My testimony discusses the rate impact of the proposed transaction with Southern Α. California Edison Company ("SCE") outlined in the Company's Application. It also describes the support that APS needs from the Arizona Corporation Commission ("Commission") to move forward with retiring Four Corners Power Plant ("Four Corners") Units 1-3 and acquiring SCE's ownership share of Units 4 and 5.

4 5

6

7

8

11

17

20

21

22

23

24

25

26

27

28

Q.

Α.

Q:

1

2

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

II.

A.

### SUMMARY OF THE PROPOSED TRANSACTION

## Q. PLEASE SUMMARIZE THE TRANSACTION PROPOSED IN THE COMPANY'S APPLICATION.

APS proposes to acquire SCE's interest in Units 4 and 5 of Four Corners, located on the Navajo Nation in Fruitland, New Mexico. If the Commission authorizes this purchase, approval for which is required under the terms of the "self-build moratorium" imposed in Decision No. 67744 (April 7, 2005), and grants the other requests made in its Application, APS will also retire Four Corners Units 1-3. This transaction essentially trades 560 MW of less efficient generation produced by Units 1-3 for 739 MW of more efficient generation produced by SCE's share of Units 4 and 5. The 179 MW difference, which is unavoidable under the circumstances due to the "lumpy" nature of generation investments, hedges the Company's energy mix against the possibility that output from other coal units also at risk could be retired. It also helps further defer the need for baseload resources.

APS Witnesses Pat Dinkel and Mark Schiavoni speak to the environmental, community, and cost benefits of the transaction. In addition to those benefits, the Company's proposed transaction will save customers as much as four percentage points on their electric bills, compared to the likely alternatives. These several benefits are compelling. To move forward on this path, however, APS needs the Commission's support of the proposed approach in certain important ways. First, APS cannot acquire SCE's ownership interest in Units 4 and 5 without authorization to do so under the terms of the "self-build moratorium" imposed in Decision No. 67744 (April 7, 2005). APS believes that the circumstances of this transaction fully support this request and respectfully asks that the Commission grant it.

2

Moreover, while undeniably cost-effective compared to the alternatives and a good value for customers, this transaction requires significant investment by APS. To address the timing mismatch between costs and benefits that will occur between when the transaction closes and when associated costs are recovered in rates, the Company's Application also requests an accounting order that will: (1) allow the Company to defer for future recovery depreciation and amortization costs, operations and maintenance costs, property taxes, final coal mine reclamation, and carrying charges associated with APS acquiring SCE's share of Units 4 and 5; and (2) provide assurance that APS will be allowed to fully recover its investment in and carrying costs of Units 1-3, and any additional costs (most notably, decommissioning and mine reclamation) incurred in connection with the closure of those units.

III.

### RATE IMPACT OF THE PROPOSED TRANSACTION

### Q. YOU NOTED THAT THE RATE IMPACT FROM THE PROPOSED TRANSACTION COULD BE AS MUCH AS FOUR PERCENTAGE POINTS LESS THAN THAT OF THE ALTERNATIVES. PLEASE EXPLAIN.

A. As APS Witness Pat Dinkel explains, the likely alternatives to the approach outlined in the Company's Application are either to replace any lost generation from Units 4 and 5 with natural gas, or to spend an estimated \$585 million between now and 2016 to continue operating Units 1-3. As Mr. Dinkel explains, neither of these options is as cost-effective for customers as retiring Units 1-3 and picking up SCE's share of Units 4 and 5.

The proposed transaction makes sense from a rate-impact perspective, too. Every scenario relating to Four Corners will cause customer bills to rise. With the proposed transaction, however, the higher costs of the new generation will be mitigated in part by the reduced operating costs for Units 1-3. Moreover, because Units 4 and 5 are approximately 10% more efficient than Units 1-3, they produce

the same amount of energy at 10% lower fuel cost, producing a commensurate fuel cost savings. Transaction costs will also be offset by the additional fuel savings that result from the displacement of 179 MW of generation that would otherwise be produced by natural gas-fueled generating units or purchased from the wholesale market. These latter fuel savings can be significant depending on the cost of natural gas at the time, and would accrue to customers almost concurrently as they run through the Company's Power Supply Adjustor ("PSA").

Because of these factors, the anticipated rate impact of APS's proposed approach is the lowest of the alternatives. As it stands today, APS will have to spend \$660 million to install environmental controls at Four Corners if the EPA rules are passed as proposed. As a result, customer bills would already increase under the status quo. If the proposed transaction moves forward, customer bills would increase by about 4% by 2017, approximately one percentage point of which relates to APS's existing ownership obligations at Four Corners Units 4 and 5. If the plant owners chose to shutdown all five units in 2016 and APS replaced the lost energy with natural gas, APS customer bills would increase by approximately 8%. If the plant owners retired Units 4 and 5 in 2016, but APS continued to operate Units 1-3, customer bills would increase by more than 6.5% in the same period.

Each of these bill impact analyses assumes that costs are reflected in rates at the beginning of the year after that in which they are incurred. The actual bill impact, of course, will vary depending upon how and when the costs are reflected in rates. That will not be determined until a subsequent rate case, in which the Commission will decide how the cost of the transaction should flow to customers.

28

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

### IV. <u>APS'S COMMITMENT TO RENEWABLE ENERGY AND ENERGY</u> <u>EFFICIENCY</u>

### Q. WILL ACQUIRING SCE'S SHARE OF UNITS 4 AND 5 EFFECT APS'S COMMITMENTS TO RENEWABLE RESOURCES UNDER DECISION NO. 71448 OR ENERGY EFFICIENCY UNDER THE COMMISSION'S NEW STANDARD?

A. No, it will not. As I mentioned previously, the additional 179 MW of baseload energy gained as a result of this transaction is a small net increase relative to the Company's total energy mix. APS's Loads and Resources table, attached to Mr. Dinkel's Testimony as Attachment PD-1, shows that even if this transaction is approved, APS will continue to meet the 10% by 2015 commitment that the Company made in Decision No. 71448 (December 30, 2009), as well as the 22% by 2020 Energy Efficiency Standard (assuming that decoupling or another appropriate fixed-cost recovery vehicle is put in place). This point is well demonstrated in Graph 3 depicted in the Application, which shows that all of APS's future energy growth through 2017 will be met with either renewable resources or energy efficiency measures. The proposed transaction serves only to maintain the current balance of the Company's energy mix.

V.

### THE COMPANY'S REQUESTS

Q. DOES APS REQUIRE SPECIFIC APPROVAL(S) FROM THE COMMISSION IN ORDER TO CONSUMMATE THE PROPOSED TRANSACTION?

A. Yes. APS cannot move forward with the preferred approach without this Commission's action. Specifically, as required under the terms of the "self-build moratorium" imposed in Decision No. 67744, APS asks that the Commission authorize it to acquire SCE's share of Units 4 and 5. APS also seeks an accounting order that will: (1) authorize the Company to defer for future recovery certain costs relating to the transaction; and (2) provide assurance that APS may

| 1<br>2   |    | continue to recover the cost of capital, depreciation, decommissioning, mine  |
|----------|----|---|
|          | -  | reclamation, and other obligations that may arise with respect to Units 1-3.  |
| 3<br>4   | Q. | WHAT ARE THE TERMS OF THE "SELF-BUILD MORATORIUM" APPROVED IN DECISION NO. 67744?   |
| 5        | Α. | Decision No. 67744 imposed certain restrictions on the Company's ability to   |
| 6        |    | construct or acquire an ownership interest in additional generating capacity.   |
| 7        |    | Except in certain circumstances not applicable here, <sup>1</sup> that Decision broadly   |
| 8        |    | requires that the Commission expressly approve "the acquisition of a unit or  |
| 9        |    | interest in a generating unit from any merchant or utility generator" with an in-   |
| 10       |    | service date prior to January 1, 2015 before APS may acquire it. See Decision   |
| 11       |    | No. 67744 (Finding of Fact 33). The Decision also sets forth criterion for APS to   |
| 12       | -  | address when seeking such approval, intended to inform the Commission's   |
| 13       | -  | determination as to whether or not it should be granted. Generally stated, these  |
| 14       |    | include:  |
| 15       |    | 1. That there is a long term need for the resource (Paragraph 75(a));   |
| 16<br>17 |    | 2. That the Company has considered and sought the availability, if<br>any, of similar resources from the competitive wholesale power<br>market (Paragraph 75(b) and (c)); |
|          |    |   |
| 18<br>19 |    | 3. That the acquisition is consistent with APS's resource plan and competitive acquisition rules (Paragraph 75(d)); and   |
| 20       |    | 4. That the life cycle costs of the resource compared to alternatives are   |
|          |    | reasonable (Paragraph 75(e)).   |
| 21       |    | See Decision No. 67744 at Appendix A, Paragraph 75.   |
| 22<br>23 | Q. | DOES APS HAVE A LONG TERM NEED THAT JUSTIFIES THE ACQUISITION?  |
| 24       | А. | Yes. As explained at length in the Testimony of Pat Dinkel, APS clearly has a   |
| 25       |    | long-term need from the capacity it will gain from this acquisition.  |
| 26       |    |   |

 <sup>&</sup>lt;sup>1</sup> Several exceptions to the need to seek prior Commission authorization are identified in Decision No. 67744. These include renewable resources, distributed generation less than 50 MW, temporary generation, etc. See
 Decision No. 67744 at Attachment A, Paragraph 74.

| 1              | Q.         | HAS THE COMPANY CONSIDERED AND SOUGHT THE<br>AVAILABILITY, IF ANY, OF SIMILAR RESOURCES FROM THE<br>COMPETITIVE WHOLESALE POWER MARKET?  |
|----------------|------------|--|
| 3              | А.         | Yes, as Mr. Dinkel also explains in his Testimony.   |
| 4              | Q.         | IS THE PROPOSED ACQUISITION CONSISTENT WITH APS'S RESOURCE PLAN AND COMPETITIVE ACQUISTION RULES?  |
| 5              | А.         | Yes, it is – a point that Mr. Dinkel also explains well.   |
| 6<br>7         | Q.         | DOES APS BELIEVE THE LIFE CYCLE COSTS ASSOCIATED WITH<br>THE ACQUISITION FAVORABLY COMPARE TO OTHER<br>ALTERNATIVES?   |
| 8              | <b>A</b> . | Yes. Mr. Dinkel also speaks to this point.   |
| 9<br>10        | Q.         | DOES APS BELIEVE THE REQUIREMENTS OF THE "SELF-BUILD<br>MORATORIUM" HAVE BEEN SUFFICIENTLY SATISFIED FOR THE<br>COMMISSION TO GRANT APPROVAL OF THE ACQUISITION?                   |
| 11             | А.         | Yes. As discussed above, the necessary criteria defined in Decision No. 67744  |
| 12             |            | have been adequately addressed. For those reasons, APS believes the  |
| 13             |            | Commission should expressly grant a waiver from the "self-build moratorium."   |
| 14<br>15<br>16 | Q.         | DOES APS ALSO REQUEST A SPECIFIC ACCOUNTING ORDER<br>AUTHORIZING A COST DEFERRAL TO FACILITATE THE<br>PURCHASE OF UNITS 4 AND 5 AND ENABLING THE EARLY<br>RETIREMENT OF UNITS 1-3? |
| 17             | А.         | Yes. APS respectfully requests that the Commission grant an accounting order   |
| 18             |            | that will, first, authorize APS to defer and capitalize for future recovery through  |
| 19             |            | rates all non-fuel costs of owning, operating, and maintaining the acquired  |
| 20             |            | interest in Four Corners Units 4 and 5. The costs to be deferred include   |
| 21             | -          | depreciation, amortization of the acquisition adjustment, operations and   |
| 22             |            | maintenance, property taxes, final coal reclamation, capital carrying charges and  |
| 23             |            | other miscellaneous costs. Second, the Commission order needs to provide   |
| 24             |            | assurance that APS will be allowed to fully recover its investment in Units 1-3  |
| 25             |            | and all associated expenses (most notably, decommissioning and mine  |
| 26             |            | reclamation).  |
| 27             |            |  |

2

3

4

5

6

7

8

9

10

0.

### WHAT IS AN ACCOUNTING ORDER?

A. An accounting order is a ratemaking mechanism that provides APS the ability to defer costs that would otherwise be expensed using Generally Accepted Accounting Principles ("GAAP"). In this case, such an order will address the timing mismatch between costs and benefits that occurs from when the proposed transaction closes (providing an immediate cost-benefit to customers) to when APS is allowed to recover the asset in rates. It also provides important financial support to APS during this same period.

### Q. CAN YOU PLEASE EXPLAIN WHY THE REQUESTED ACCOUNTING ORDER IS IMPORTANT FOR APS TO MOVE FORWARD WITH THIS TRANSACTION?

APS customers will see substantial long-term cost savings if the proposed Α. 11 transaction is approved, as discussed in Mr. Dinkel's Testimony. Those savings, 12 however, come at a significant short term cost that would have to be absorbed 13 entirely by APS, absent a deferral. There is, for example, the \$294 million 14 purchase price and the increased operating expenses associated with the 15 additional ownership in Units 4 and 5. APS will also assume certain of SCE's 16 assets and liabilities (such as those associated with final plant decommissioning 17 and coal reclamation), which APS will record at fair value at the time of the 18 acquisition.<sup>2</sup> Collectively, these new costs amount to an estimated revenue 19 requirement of \$70 million per year. If APS is not able to defer these costs, it 20 will forever lose the opportunity to recover them. 21

> This inequitable result is amplified by the Company's PSA. Under that adjustment mechanism, customers will immediately benefit from the fuel savings that will result from the proposed transaction. But because of the 90/10 sharing component of the PSA, APS will have only ten percent of that savings available to help offset the transaction's costs. Put another way, APS customers will pay

27 ·28

22

23

24

25

<sup>&</sup>lt;sup>2</sup> The latter costs are currently estimated to total \$32 million.

less because of this transaction after it closes, but will not be required to pay for it until the acquisition is reflected in rates in a subsequent rate case. The Company will incur significant new costs to own and operate in Units 4 and 5, but will not be able to offset those cost increases in any meaningful way with the anticipated fuel savings. A deferral order will partially remedy this mismatch until the increased costs are reflected in rates.

In addition, if APS retires Units 1-3 in 2012, but is not clearly authorized by the Commission to fully recover the remaining plant value on those units and their associated costs, APS could likely be forced to take a current write-off on those assets. This would weaken the Company's equity ratio (a key concern in recent APS rate proceedings) and depress projected financial results.

To address these consequences, APS requests an accounting order that will preserve its ability to recover costs related to the proposed transaction in a subsequent rate case – accounting relief that the Commission has historically allowed. Specifically, APS seeks an order allowing it to defer and capitalize for future recovery all prudent non-fuel operating, maintenance, depreciation, and capital carrying costs associated with SCE's share of Units 4 and 5. For purposes of the cost deferral, APS will calculate capital costs using the embedded cost of debt as of December 31, 2010 and the 11% cost of equity used in APS's last general rate case, at the ratio of 46.21% debt and 53.79% equity also set in that case. The property taxes deferred will be based on actual property taxes that APS pays on the SCE share post-acquisition (less any property tax savings on Units 1-3). APS will calculate depreciation at the Commission-authorized depreciation rates for APS's present interest in Unit's 4 and 5. Operations and Maintenance ("O&M") costs will be deferred and capitalized only to the extent that APS's

share of overall Four Corners O&M increases as a result of the acquisition of SCE's interest.

A.

The deferral would not increase customer bills today, but would simply provide a mechanism for APS to ensure future recovery of the costs related to the acquisition and mitigate the impact of the transaction on the Company's financial condition.

Q. HAS THE COMMISSION PREVIOUSLY GRANTED DEFERRAL ORDERS TO ASSIST THE COMPANY IN MANAGING THE FINANCIAL COSTS ASSOCIATED WITH NEW GENERATION ASSETS?

Yes, the Commission has historically approved accounting orders in appropriate circumstances, such as these. For example, the Commission granted APS an accounting order that allowed the Company to defer the costs of owning, operating, and maintaining Sundance Generating Station (a simple cycle natural gas plant) when APS acquired that facility in 2005. *See* Decision No. 67504 (January 20, 2005). Authorizing the requested deferral, the Commission expressly recognized that the "PSA's cost/savings sharing component . . . coupled with the timing of the acquisition in relation to the pending rate case, could create a potential for inequity that justifies the creation of a deferral in this case." *Id.* at 25.

The Commission similarly granted APS two accounting deferral orders so that the Company would be able manage the financial impact of rising costs associated with owning new generation at Palo Verde Unit 2 and Palo Verde Unit 3. *See* Decision Nos. 55325 (December 5, 1986) and 55939 (April 6, 1988). In both of these instances, the Commission found the financial burden on APS absent the ability to defer the costs of owning and operating Palo Verde Units 2 and 3 and the essential inequity of the mismatch between cost recovery by APS

and benefits realized by APS customers as sufficient reasons to grant the Company's request.<sup>3</sup> That rationale applies equally here.

The ability to defer for future recovery costs associated with the newly purchased share of Units 4 and 5 and to continue recovering the depreciation, decommissioning, mine reclamation and other expense obligations related to Units 1-3 will address the timing mismatch between when the transaction closes and when the underlying costs are recovered through rates, mitigating the inequity of APS forever losing the ability to recover expenses paid while the asset is in-service, benefiting customers. The deferral ensures that the deferred costs will eventually be recovered and mitigates the impact on the Company's earnings profile. The regulatory certainty of recovery that a deferral order will provide will likely allow APS to finance the transaction more easily and at lower costs, to the long-term benefit of customers.

## Q. IS ANY SPECIFIC LANGUAGE REQUIRED TO PROVIDE THE DEFERRAL FOR ACCOUNTING PURPOSES?

A. Yes. GAAP requires that the regulatory authorization of the deferral include language that provides APS and its auditors with reasonable assurance that the Company will recover prudently incurred costs in future rates. The accounting order must also include language that provides for the future recovery of costs associated with the closure of Units 1-3 in order to avoid the potential for immediate write-off of all or a portion of those costs upon closure (or even the final announcement by APS of closure). Such language is proposed in Attachment B to the Company's Application. To effectuate any cost-deferral

 <sup>&</sup>lt;sup>3</sup> More recently, the Commission granted APS an accounting order as part of the 2009 Settlement Agreement, in which it permitted APS to defer higher employee benefit costs. See Decision No. 71448 (December 30, 2009) at Appendix A, Paragraph 9. Previous Commission decisions similarly authorized APS to defer costs associated with net metering and vegetation management. See Decision Nos. 69663 (June 27, 2007) at 89 and 67744 at Appendix A, Paragraphs 110 and 111, respectively.

authorized, APS respectfully requests that the Commission adopt this or substantially similar language in its own final order.

Q.

## DOES THE ACCOUNTING ORDER INCLUDE FUTURE CAPITAL INVESTMENTS REQUIRED FOR UNITS 4 AND 5?

A. No. The accounting order requested will not alleviate the financial impact of the capital investment that APS and other plant participants will likely make over the course of the next five years to bring Units 4 and 5 into environmental compliance. As in prior days of high generation capital requirements (such as when APS was bringing Palo Verde online), it will be important to find ways to phase-in the significant near-term incremental capital investment related to APS's coal strategy so as to avoid the financial pressure that will result were these costs shouldered all at once. APS does not ask that any such rate making mechanisms be addressed in this docket. The conversation applies equally to all coal-fired generation, not just Four Corners, and is a topic that APS intends to include in its next rate case.

16 VI. <u>CONCLUSION</u>

17 Q. DO YOU HAVE ANY CONCLUDING REMARKS?

A. Yes. The proposed acquisition of SCE's interest in Four Corners Units 4 and 5 is
 a good deal for APS customers, the Navajo Nation, and the environment. The
 requests contained in the Company's Application are critical for the transaction to
 move forward, and APS respectfully requests that they be granted.

22 Q. DOES THIS CONCLUDE YOUR DIRECT TESTIMONY?

- 23 A. Yes.